Research Report Series No. 146

16 APR 1986
DEVELOPMENT STUDIES
LIBRARY

#### SOME ASPECTS OF AGRICULTURAL PRICE AND TAXATION POLICIES IN PAKISTAN

SARFRAZ KHAN QURESHI SOHAIL JEHANGIR MALIK AHMED NAEEM SIDDIQUI

Kalbe Abbas, Ejaz Ghani and Zeb-un-Nisa

ocumentation & Library Centre University of Sussex Falmer Brighton BN1 9RE, UK

PLEASE RETURN BY:

September 1985

PAKISTAN INSTITUTE OF DEVELOPMENT ECON POST BOX 1091, ISLAMABAD (Pakistan)

# CONVENOR, RESEARCH REPORT COMMITTEE

Dr. M. Ghaffar Chaudhry

The authors would like to gratefully acknowledge the continued guidance and support of Professor Syed Nawab Haider Naqvi, Director, Pakistan Institute of Development Economics. They are also greatly indebted to Drs. Vernon W. Ruttan, Peter A. Cornelisse, and Abdul Salam for valuable written comments on earlier drafts of the study and to Dr. M. Ghaffar Chaudhry for innumerable discussions. They would like to acknowledge the valuable editorial assistance of Mr. S. H. H. Naqavi, the Literary Editor of the Pakistan Development Review (PDR). The authors alone are, however, responsible for any errors that may remain.

# CONTENTS

CONTROL IV	PAGE
LIST OF TEXT TABLES	(vi)
LIST OF ANNEXURE TABLES	(ix)
CHAPTER I INTRODUCTION	1
Organization of the Study	2
CHAPTER II	
RURAL INCOME DISTRIBUTION AND POVERTY:	4
1. INTRODUCTION	4
2. A REVIEW OF DATA	4
3. RURAL INCOME INEQUALITIES	6
4. RURAL POVERTY	14
5. CONCLUSIONS	19
CHAPTER III	103
DOMESTIC TERMS OF TRADE AND PUBLIC POLICY	22
2. BRIEF REVIEW OF PREVIOUS STUDIES	22
3. MOVEMENTS IN AGRICULTURE'S TERMS OF TRADE	27
4. SELECTED ASPECTS OF THE IMPACT ON RURAL ECONOMY	33
4.1 Price Incentives and Aggregate Farm Output	33
4.2 Trade Policy and Resource Use	35
4.3 Impact on Distribution of Income	36
5. CONCLUSIONS AND POLICY IMPLICATIONS	41
ANNEXURE	45
DEFINITION OF VARIABLES	46

		PAGE
CHAPTER	3. ASECTORAL TAX EQUITOR STATE ASSOCIATION OF THE STATE O	
INTE	RA-SECTORAL PARITY ISSUES IN PRICING OF	89
707	AGRICULTURAL OUTPUT AND INPUTS	
1821.	INTRODUCTION	89
2.	COSTS OF PRODUCTION	90
3.	TRENDS IN PRICES OF INPUTS	92
014.	TRENDS IN PRICES OF AGRICULTURAL COMMODITIES	94
5.	PARITY INDICES OF PRICES OF AGRICULTURAL CROPS AND INPUTS	96
6.	TRENDS IN GROSS INCOME OF CROPS AND INTER-CROP GROSS INCOME PARITY INDICES	98
7.	CONCLUSIONS AND POLICY IMPLICATIONS	102
CHAPTER	Account at her to you such a feet to we have the to the	
PRIC	CE RESPONSIVENESS IN PAKISTAN'S AGRICULTURE:	104
1.	INTRODUCTION	104
2.	AGGREGATE OUTPUT RESPONSE	105
3.	RESPONSIVENESS OF CROP ACREAGE	109
4.	RESPONSIVENESS OF MARKET SUPPLY	116
5.	SUMMARY AND CONCLUSION	122
CHAPTER	(a) Prodel Tax System IV	
INTE	ERSECTORAL TAX EQUITY	130
1.	INTRODUCTION	130
2.	CONCEPTUAL FRAMEWORK AND ITS LIMITATIONS	131
166	2.1 Tax Burden Measure	135
	2.2 Intersectoral Equity	136
	(ii)	

			PAGE
		RAL TAX EOUITY IN PAKISTAN:	136
	3.1	Shifting of Taxes in Pakistan	137
		3.1.1 Land Tax	137
		3.1.2 Income Tax	138
		3.1.3 Company Tax	138
58		3.1.4 Export Duties	140
	3	3.1.5 Import Duties	140
86		3.1.6 Domestic Indirect Taxes	141
	3.2	Allocation of Taxes of Sectors in Pakistan	143
		3.2.1 Burden of Non-agricultural Direct Taxes	143
		3.2.2 Burden of Shifted Indirect Taxes on Sectors	144
101	3.3	Estimation of Relative per Capita Income and per Capita Subsistence Requirement by Sectors	145
		3.3.1 Per Capita Incomes in the Sectors	146
		3.3.2 Per Capita Subsistence Requirements	147
	3.4	The Tax Burden Estimates: The First Approximation.	149
		RESPONSIVENESS OF MARKET SUPPLY	
	4. SECTO	RAL TAX EQUITY IN PAKISTAN: SECOND APPROXIMATION	156
	(a)	Model Tax System	157
130	(b)	Least Squares Method	157
0.10	(c)	Arbitrary Weights	157
	(d)	Method Adopted in the Study	159
	5. SUMMAI	RY AND CONCLUSIONS	166

		PAGE
ANN	EXURE	169
	DISTRIBUTION OF TAX BURDEN BETWEEN AGRICULTURAL AND NON-AGRICULTURAL SECTORS	169
	CENTRAL EXCISE DUTIES	170
	IMPORT DUTIES	174
	SALES TAX ON DOMESTIC AND IMPORTED GOODS	177
	PROVINCIAL EXCISE	177
PTER	VII	
NET	FISCAL BURDEN ON AGRICULTURAL PRODUCERS	178
1.	INTRODUCTION	178
2.	OPEN TAXATION OF THE AGRICULTURAL SECTOR	178
3.	CONCEALED TAXATION OF AGRICULTURE	180
4.	OPEN SUBSIDIES TO FARMERS	182
5.	CONCEALED SUBSIDIES	183
	5.1 Supply of Low Interest Loans	184
	5.2 Low Irrigation Water Rates	186
	5.3 Subsidy on Electricity	187
6.	NET FISCAL BURDEN ON FARMER-PRODUCERS	188
	LANGE THE DISTRIBUTION OF MENT PLOYER BY SOURCES	
APTER	VIII	
POLI	ITICAL ECONOMY OF LAND TAXES	189
1.	INTRODUCTION	189
2.	THE SYSTEM OF LAND TAXES IN PAKISTAN	191
3.	ANOMALIES IN THE LAND TAX SYSTEM	194
4.	EQUITY ASPECTS OF LAND TAXES	195
	4.1 Tax Collection	197
	4.2 Agricultural Population	197
	4.3 Gross and Net Value of Agricultural Production	197
5.	EFFICIENCY ASPECTS OF LAND TAXES	201

	THE PROPERTY OF THE PROPERTY O
CUA	
CHA	THE IA
	CONCLUSIONS AND POLICY IMPLICATIONS
RID	LIOGRAPHY
DID	LIV-93T
	MET FISCAL BURDEN ON AGRICULTURAL PRODUCEDS
	TATROPUCTION
178	
	CONCEALED TAXATION OF AGRICULTURE
	WEN SUBSTRIES TO FARMERS
	CONCEALED SUBSIDIES
181	6. Epulling of ceammanterment work to views 1.8
	5.2 Low Irrigation Maternal Wol 171100 and S.2
	MET FISCAL BURDEN ON FARMER-PRODUCERSXJ. SHIPP
	CONCLUSIONS AND POLICY IMPLICATIONS
	LICERAPHY STORAL TEACHER STORE STATE OF THOROUGH LASTITUTE
	AND
	and the second s
	THE SYSTEM OF LAND TAXES IN PARISTAN
	ANGMALIES IN THE LAND YAX SYSTEM
	COULTY ASPECTS OF LAND TAXLS
197	lax Collection
	neggand femulation of Agriculture) Production
1410	(v) AT BAND TA (v)

POLITICS OF LAND TAXES

6.

# LIST OF TEXT TABLES

		PAGE
PTER II		
TABLE. II.1:	RURAL INCOME CONCENTRATION RATIOS FOR VARIOUS YEARS	9
TABLE II.2:	PERCENTAGE LAND SHARES IN PAKISTAN FOR SELECTED YEARS	12
TABLE II.3 :	INCIDENCE OF POVERTY IN RURAL AREAS IN TERMS OF REAL EXPENDITURE	16
TABLE II.4 :	INCIDENCE OF POVERTY IN RURAL AREAS IN TERMS OF REAL INCOME	16
TABLE II.5 :	RURAL POVERTY ESTIMATES BASED ON PER CAPITA INCOME PER ANNUM	18
TABLE II.6 :	ESTIMATES OF RURAL POVERTY BASED ON MINIMUM NUTRITIONAL REQUIREMENTS	.18
TABLE II.7:	INCIDENCE OF RURAL POVERTY BASED ON HOUSEHOLD AND POPULATION - 1979	20
NIVIN ISO	ACCRECATION NITH ARBITRARY WELGHTS OF	
PTER III	VI LANGARA - ARMANIA ARMANIA ARMANIA MATERIA ARMANIA MATERIA.	
TABLE III.1:	TERMS OF TRADE FOR AGRICULTURE (Three Years Moving Average: 1951/52 - 1983/84)	28
TABLE III.2:	PERCENTAGE DISTRIBUTION OF WHEAT FLOUR BY SOURCES FOR TYPE OF HOUSEHOLD AND PROVINCE	<b>3</b> 8
C.016 MAN		
PTER IV	TABLE WIRE CATHATES OF TAXABLE CAPACITY PER CAPITA OF	
TABLE IV.1:	COST OF PRODUCTION AND PRICE-COST RATIOS FOR SELECTED CROPS	91
TABLE IV.2:	INDEX NUMBERS OF PRICES OF AGRICULTURAL INPUTS	93
TABLE IV.3:	PRODUCER PRICES FOR MAJOR CROPS	95
TABLE IV.4 :	PARITY INDICES OF MAJOR AGRICULTURAL COMMODITIES	97
TABLE IV.5 :	GROSS INCOME PER ACRE OF MAJOR CROPS IN PAKISTAN	99
TABLE IV.6:	INDICES OF INTER-CROP GROSS INCOME PARITY AND GROSS INCOME-INPUT PRICES PARITY	101

## GHAPTER W

- TABLE V.1 ESTIMATES OF SHORT AND LONG RUN PRICE ELASTI-CLINES AND LESTIMATED VALUES OF THE ADJUSTMENT COEFFICIENTS IN THE ADAPTIVE EXPECTATIONS MODEL, BY DIFFERENT STUDIES
- TABLE V.2 RANGES OF PRICE ELASTICITY OF MARKETED SURPLUS OF WHEAT

# CHAPTER VI

- TABLE VI.1 INCIDENCE OF FEDERAL AND PROVINCIAL TAXES ON DIFFERENT SECTORS
- TABLE VI.2 SOME INDICATORS OF TAXABLE CAPACITY OF DIFFERENT SECTORS
- TABLE VI.3 RELATIVE TAX BURDEN ON THE AGRICULTURAL SECTOR: FIRST APPROXIMATION
- TABLE VI.4 TAX PER CAPITA, EQUITY TAX PER CAPITA, RESOURCE TRANSFER FROM AGRICULTURE -FIRST APPROXIMATION
- TABLE VI.5 ESTIMATES OF RELATIVE TAXABLE CAPACITY OF THE AGRICULTURAL SECTOR WITH ARBITRARY WEIGHTS GIVEN TO:INCOME PER CAPITA AND INCOME INEQUALITY
- TABLE VI.6 RURAL AND URBAN INCOME DISTRIBUTION FOR THE YEAR 1979-80
- TABLE VI.7 TAXABLE CAPACITY OF AGRICULTURAL SECTOR PER CAPITA
- TABLE VI.8 TAXABLE CAPACITY OF NON-AGRICULTURAL SECTOR RER CAPITA
- TABLE VI.9 ESTIMATES OF TAXABLE CAPACITY PER CAPITA OF DIFFERENT SECTORS WITH MODERATE PROGRESSION DEFINED AS TAXABLE CAPACITY RAISED TO THE POWER 1.5.

GE			CIST OF ANNEXURE TABLES	PAGE
A	PTER VI	I		estra a
14	TABLE	VII.1	INCIDENCE OF DIRECT AND COMMODITY TAXES ON AGRICULTURAL PRODUCERS	179
	TABLE	VII.2	CONCEALED TAXATION OF THE AGRICULTURAL SECTOR FOR SELECTED CROPS	184
}	TABLE	VII.3	OPEN SUBSIDIES ON AGRICULTURAL INPUTS	182
	TABLE	VII.4	SUPPLY OF AGRICULTURAL CREDIT BY SOURCE AND ESTIMATES OF SUBSIDY ON AGRICULTURAL CREDIT	185
	TABLE	VII.5	SUBSIDY ON IRRIGATION WATER SUPPLIED THROUGH CANALS	186
1	TABLE	VII.6	ESTIMATED SUBSIDY ON ELECTRICITY SUPPLY TO THE AGRICULTURAL SECTOR	187
	TABLE	VII.7	TAXES AND SUBSIDIES IN AGRICULTURAL SECTOR	188-A
CHA	PTER VI	11	E-A 3.7 : DOMESTIC TERMS OF TRADE FOR AGRICULT ALTERNATIVE METHOD (Three Years Moving Average)	
	TABLE	VIII.1	REGRESSIONS OF PERCENTAGE OF INCOME PAID IN TAXES ON PER CAPITA INCOME	199
	TABLE	VIII.2	REGRESSION ESTIMATES OF VALUE OF AGRICULTURAL OUTPUT PER ACRE ON LAND TAX PER ACRE AND INDICE OF GOVERNMENT EXPENDITURE	204 S

" The second of the second

ENTER SECTIONAL TERMS OF THISBE FOR MANUFACTURING

PARTE SECTIONAL TO TRANSFER FOR ASSISTED TOTAL

SECTOR (Three Years Moving Average)

SECTOR times Ways Noving Average

INDICES OF AGRICULTURAL WHOLESALE DOMESTIC PRICES (1959-60 = 100)

1959/68 WEIGHTS FOR PROTUCTION, MARKETING, AND PURCHASES OF ANNIONITUDAL COMMODITIES

# LIST OF ANNEXURE TABLES

### CHAPTER III

DOMESTIC TERMS OF TRADE FOR MANUFACTURING TABLE-A 3.1 : SECTOR TABLE-A 3.2 : DOMESTIC TERMS OF TRADE FOR MANUFACTURING SECTOR (Three Years Moving Average) DOMESTIC TERMS OF TRADE FOR AGRICULTURE TABLE-A 3.3 : (ABSOLUTE) TABLE-A 3.4 : DOMESTIC TERMS OF TRADE FOR AGRICULTURE (Three Years Moving Average) TABLE-A 3.5 : INCOME TERMS OF TRADE FOR AGRICULTURE (Three Years Moving Average) SINGLE FACTORAL TERMS OF TRADE FOR AGRICULTURE TABLE-A 3.6 : (Three Years Moving Average) DOMESTI€ TERMS OF TRADE FOR AGRICULTURE: TABLE-A 3.7 : ALTERNATIVE METHOD (Three Years Moving Average) TABLE-A 3.8 : DOMESTIC TERMS OF TRADE FOR AGRICULTURE: ALTERNATIVE METHOD (Three Years Moving Average) TABLE-A 3.9 : DOMESTIC TERMS OF TRADE FOR AGRICULTURE: ALTERNATIVE METHOD (Three Years Moving Average) INTRA SECTORAL TERMS OF TRADE FOR MANUFACTURING TABLE-A 3.10 : SECTOR (Three Years Moving Average) INTRA SECTORAL TERMS OF TRADE FOR AGRICULTURAL TABLE-A 3.11 : SECTOR (Three Years Moving Average) TABLE-A 3.12 : PRICE INDICES FOR MANUFACTURING GOODS TABLE-A 3.13 : INDICES OF AGRICULTURAL WHOLESALE DOMESTIC PRICES (1959-60 = 100)TABLE-A 3.14 : 1959/60 WEIGHTS FOR PRODUCTION, MARKETING, AND PURCHASES OF AGRICULTURAL COMMODITIES TABLE-A 3.15 : 1959/60 WEIGHTS FOR PRODUCTION AND PURCHASES OF NON-AGRICULTURAL GOODS TABLE-A 3.16 : WEIGHTED DOMESTIC PRICE INDICES FOR NON-AGRICULTURAL COMMODITIES

		PAGE
	HTED DOMESTIC PRICE INDICES FOR SRICULTURE	81
INDIC	RNATIVELY WEIGHTED DOMESTIC PRICE CES FOR MANUFACTURING BY MAJOR ROUPINGS	82-83
INDIC	RNATIVELY WEIGHTED DOMESTIC PRICE CES FOR MANUFACTURING BY BROAD ATEGORIES	84-85
INDIC	RNATIVELY WEIGHTED DOMESTIC PRICE CES FOR AGRICULTURE BY MAJOR ROUPING	86
	CULTURAL INPUTS IN PAKISTAN - 1953-54 D 1983-84	87
AND 1	EGATE INPUT INDEX, VALUE ADDED INDEX THE TOTAL FACTOR PRODUCTIVITY INDEX -54 TO 1983-84 (Base Year 1959-60)	88
HAPTER V		
	MATED PARAMETERS FOR PRICE RESPONSIVENESS CREAGE AT THE NATIONAL LEVEL	125
	T AND LONG RUN ELASTICITY ESTIMATES BASED ATA FOR DIVISIONS	126
	T AND LONG RUN ELASTICITY ESTIMATES BASED ATA FOR SELECTED DISTRICTS.	127

### Chapter I

#### INTRODUCTION

Despite theoretical arguments against government intervention in the pricing of agricultural commodities and inputs, most developing countries actively distort agricultural prices and maintain a wide divergence between relative domestic and relative world market prices.

This distortion almost invariably involves the taxation or subsidization of one or more agricultural commodities and/or agricultural inputs. Since the commodities and inputs affected by government price intervention are often of great importance to a country's production, consumption, income and trade flows, the impact of the price intervention can be large and pervasive and may have important implications for public resource mobilization, incentives to farmers, efficiency in resource allocation, equity in the system and the rate of economic growth. A clear understanding of the nature and the magnitude of the manifold effects of agricultural price policy is of paramount importance to the pglicy-makers.

Like most other developing countries, Pakistan also has a complex system of agricultural taxes, input subsidies, government monopoly trading and price-setting arrangements. The objectives behind this wide array of policy instruments and the impact that these policy instruments may have had on the economy is an important area of enquiry. Despite the importance of this subject, not many studies have emerged. The main purpose of the

<sup>1/</sup> The most well known studies on this subject are:1) Gotsch, C. and G.Brown, Prices, Taxes and Subsidies in Pakistan Agriculture, 1960-70, World Bank Staff Working Paper No. 387, April 1980 and 2) Cheong Kee-Cheok and Emmanuel H.D'Silva, Prices, Terms of Trade, and the Role of Government in Pakistan's Agriculture, World Bank Staff Working Paper No. 643.

present study is to elucidate some of the effects of the agricultural pripolicy, provide an empirical basis for further work in this important fie of research and suggest a package of changes required in the mix of existing levies on agriculture to achieve the ends of public policy.

### Organization of the Study

An important objective of the price policy is to achieve a desir pattern of income distribution and to reduce the incidence of poverty to level deemed desirable by the policy-makers. The discussion begins with brief overview of the performance of Pakistan's economy as it has affecte the incidence of poverty and relative inequalities of income in the rural areas. Because of the complex nature of the relationship between public policies and the distributional parameters, it was not possible to measur the contribution of different policies on distribution of income. Nevertheless the crucial role of agricultural policies in bringing about a certain pattern of income distribution deemed desirable is clearly brought out.

Chapter III presents an extended discussion of changes in the domestic incentives facing farmers in Pakistan since mid 1950s. Barter, single factoral and income terms of trade for the period 1951/52 to 1983/ are computed and analysed. The impact of changing incentives on aggregat farm output, intersectoral distribution of income and resource use efficiency is also briefly traced. Chapter IV deals with the changing levels of prices, costs and income from different agricultural crops.

In a situation where a government sets specific targets of production for different crops, empirical knowledge about the relationship between costs, prices and income accruing from different crops is an essential piece of information for a policy-maker. Chapter V reviews the evidence in Pakistan regarding the responsiveness of farmers to changes in relative prices between different agricultural commodities and between agricultural and non-agricultural commodities.

Chapter VI examines the issue of intersectoral tax equity and presents the estimates and analysis of relative tax burden on the agricultural and non-agricultural sector. The broad objective is to test the hypothesis that the farm sector in Pakistan is under-taxed.

Chapter VII raises the question as to whether the price and tax policies that are consistent (or inconsistent) with resource use efficiency and equity are also consistent with the government objective of raising resources for financing development. Net fiscal burden of the plethora of taxes and subsidies on farmer-producers are estimated. The results indicate growing difficulties for resource mobilization through the prevailing system of taxes and subsidies. Chapter VIII evaluates the existing land tax system and indicates the possible improvements in the design of the tax that would simultaneously generate revenue and promote agricultural development. The policy implications are indicated in each chapter but are summarized again in the concluding chapter.

# CHAPTER II

# RURAL INCOME DISTRIBUTION AND POVERTY: A REVIEW OF EMPIRICAL EVIDENCE

#### INTRODUCTION

It is a commonly held by ief following Simon Kuznets and other scholars, that economic growth in its initial stages leads to greater inequalities in income distribution. Empirical verification for this can be found amongst other sources in the work of Ahluwalia \_\_\_\_3 \_\_\_/. This chapter presents a survey of existing evidence on rural income distribution and poverty in Pakistan. The analysis is aimed at highlighting the changes in rural income distribution and poverty over time. The following section presents an appraisal of the data used in studies on income distribution and poverty. The third section presents a review of evidence on rural income distribution while the fourth section is concerned with evidence on rural poverty. The fifth section presents the main conclusions of this survey.

# 2. A REVIEW OF DATA

Most of the studies on Income Distribution and Poverty in Pakistan are based on the periodic Household Income and Expenditure Survey conducted by the Statistics Division of the Government of Pakistan. The results of the first major survey available in printed form pertain to the year 1963-64. Survey results are also available in printed form for the years 1966-67, 1968-69, 1969-70, 1970-71, 1971-72 and 1979. Surveys were also conducted for the years 1959-60, 1964-65, 1965-66 and 1967-68. However, these pertained to either the rural sector or the

urban sector only or covered only the income and expenditure for a portion of the year. The results of the 1967-68 survey were not compiled. The published statistics from these surveys provide information on the incomes and expenditures of households within various income categories. The three major inadequacies of the Household Income and Expenditure Survey data as summarized by Kemal 1/4/ are listed below.

The small sample size makes the measurement of income inequalities imprecise, especially in view of the large sampling errors associated with thinly populated income groups. This problem particularly affects the upper income groups where the number of observations is very small and thus the incomes of the highest income categories are understated. This problem was first highlighted by Bergan / 11 / and is clear from a comparison of blown up figures for gross national product based upon these surveys and the national accounts data. The figures from the Household Income and Expenditure Surveys are consistently smaller.

The second major problem is that of non-response and this problem is severest in the highest and lowest income categories. Thus measures of income inequality based on these data would understate the true reality. The incomes of the highest income groups are further understated because corporate savings (retained earnings) are not defined as earnings in these surveys.

Azfar / 9 / tried to adjust for the understatement of incomes of the highest income groups by splicing the survey data with information on the size distribution of farms by types of

These inadequacies of the Household Income and Expenditure Survey need to be borne in mind when evaluating the evidence on rural income inequalities and poverty.

#### 3. RURAL INCOME INEQUALITIES:

There have been a number of studies done to-date to estimate rural income inequalities in Pakistan. The studies by Bergan / 11 \_7, Azfar / 9 \_7, Suleman / 85 \_/, Khandekar / 47 \_7, Nasim / 62 \_7, Allauddin / 5 \_7, Ayub / 7 \_7, Sadiq / 81 \_/ and Faiz / 22 \_7 present estimates of the gini-coefficients for the rural sector in Pakistan.

Bergan's / 11 / study is based upon the 1963-64
Household Income and Expenditure Survey Data. He rearranged
the data by deciles and computed gini-coefficients for both
the rural and urban sectors of Pakistan. He estimated the value
of the gini-coefficient for the rural areas to be 0.357. The
estimated gini coefficient for the urban areas was higher at
0.430 while the value of the rural and urban areas combined was

0.381. These values Bergan / 11 / found to be smaller than **those** for other developing countries but noted that these could be the result of errors in the data that could have biased the inequality measure downwards.

Expenditure Survey data by computing the incomes in the highest income groups from alternative sources. He found the gini-coeffici for personal income in the rural areas to be 0.334, which shows a decline in income inequality when compared with Bergan's results for 1963-64. However since the data used in the two studies are not completely comparable, it is difficult to state categorically that there was a decline in rural income inequality. There is contrary evidence in the study by Suleman / 85\_/, that shows an increase in income inequalities over the period 1963-64 to 1968-69.

Suleman / 85\_/ aggregated the income categories reported in the Household Income and Expenditure Surveys into six major groups. level of aggregation does not seem ideal for studying changes in income inequalities.

The study by Khandekar / 47 / presents gini coefficient for household income, income per capita and per earner for the ye 1963-64, 1966-67 and 1968-69. The estimates of the gini-coeffici for the rural sector show a decline in rural income inequalities the basis of all three ratios over this period. These results ca serious doubt on the results of Suleman's / 85 / study and indicate that the increase in rural income inequality reported by him could be due to the aggregation of data into a limited number categories.

The studies by Ayub [ 7 ], Sadiq [ 81 ] and Allauddin [ 5 ] extended the period of analysis of the earlier studies and reported gini-coefficients for the years 1963-64, 1966-67, 1968-69, 1969-70, 1970-71 and 1971-72. The gini-coefficients for the income of rural areas show a consistent decline during the 1960s right up to 1970-71. However the gini-coefficients for the 1971-72 in all three studies show an increase over the previous period.

In all the studies reviewed, urban income inequalities are higher than rural income inequalities. An interesting feature of these studies is that t computed values of the gini-coefficients based upon the same data differ from study to study. Kemal / 44 / asserts that this could be due to the use of different formulae by these authors.

There are studies, for example Naseem  $\mathcal{L}$  62  $\mathcal{J}$  and Allauddin  $\mathcal{L}$  5  $\mathcal{J}$  that present inequality measures for expenditure also. The trends in expenditure inequalities are however, similar to those for incomes. However, it is noticed that expenditures are more equally distributed than incomes.

Faiz L 22 J in a recent unpublished study presents rural income concentration ratios for a number of years from 1959 to 1979 based upon the same Household Income and Expenditure Survey data. These are reproduced in Table 1. A perusal of this table confirms the results of earlier studies. Rural income inequalities declined both on a per household and per capita basis upto 1970-71. However rural income inequality appears to have increase in the 1970s.

The decline in the rural income inequalities during the period of

TABLE II.1

RURAL INCOME CONCENTRATION RATIOS FOR VARIOUS YEARS

W.E.A.D.C	Income Concen	tration Ratio Based on
Y E A R S	Household Income	Per Capita Income
1959	0.348	0.228
1961	0.357	0.203
1963-64	0.348	0.223
1966-67	0.319	0.186
1968-69	0.294	0.161
1969-70	0.295	0.161
1970-71	0.291	0.146
1971-72	0.308	0.164
1979	0.318	0.184

SOURCE: [ 22] estimates based on Household Income and Survey.

remaining statement entered and an entered of the state at another profession

the 1960s coincides with the green revolution period in Pakistan, which marked a period of remarkable growth in the agriculture sector from the mid to late years of that decade. A review of literature connected to the green revolution phenomenon in Pakistan reveals a lack of concensus amongst researchers as to how the gains from this growth were distributed. A considerable body of researchers have maintained that the green revolution led, in one way or another, to a worsening of income inequalities. This view is in direct contrast to the evidence of a reduction of income inequalities in the rural sector presented by the studies of income distribution reviewed above. The studies by Gotsch  $\angle$  33 7, Falcon  $\angle$  23  $\angle$  , Pearse  $\angle$  76  $\angle$  7, Alavi  $\mathcal{L}$  6  $\mathcal{J}$ , Griffin  $\mathcal{L}$  34  $\mathcal{J}$  and Khan  $\mathcal{L}$  46  $\mathcal{J}$  however, maintain that the green revolution in one way or another, led to a worsening of income inequalities. The arguments put forward by these studies are based upon the premise that the green revolution technology is indivisible and hence only the large and financially secure farmers could employ it and partake of the gains from it. It is maintained that the increased profitability in agriculture led the large farmers to increase the size of their holdings and take up self cultivation. This not only increased the incomes of the richest classes but also reduced the incomes of the poorest tenant farmers who were evicted in the process. However none of these studies has presented substantive empirical evidence to support their arguments. any

a differential impact upon productivity across different farm sizes and it can effect income distribution via the demand for labour. Chaudhry / 16 / has presented evidence from a number of sources on each of these aspects to substantiate his hypothesis that the green revolution led to a reduction in rural sector income inequalities in Pakistan.

Chaudhry / 16/ has used data from the Agricultural Censuses to show that land concentration actually declined over the 1960s based upon a comparison of land concentration ratios from the 1960 and 1972 censuses. Faiz / 22/ has recently extended the analysis by Chaudhry to include evidence from the 1980 census. Table II.2 presents cumulative land shares and land concentration ratios based upon the three agricultural censuses. A perusal of Fable II.2 reveals that while Chaudhry's / 16/ contention of a decline in land concentration is borne out by a comparison of the ratios for 1960 and 1972, the data show an increase in land concentration when comparing 1972 with 1980. This is in line with the evidence of an increase in the 1970s presented earlier.

There is evidence, for example the Planning and Development Statistical Surveys /  $\bar{o}8$  / and Azam / 8 /, to show that fertilizer use in particular and modern input use in general was quite common amongst small farmers as a result of the green revolution. Chaudhry / 16 / has presented evidence to show that while the utilization of tractors, tubewells and fertilizers etc., was higher in large farms, the growth in the utilization of these inputs was much higher in the small farm sector. This had its effect on the growth of incomes in small farm sector. Chaudhry / 16 / finds a strong inverse relationship between farm size and productivity, and that productivity differences between small and large farms widened over time. Using Family Accounts and Farm Budgets data for cultivators in the Punjab collected by the then Punjab Board of Economic Enquiry to compute

TABLE II.2

PERCENTAGE LAND SHARES IN PAKISTAN FOR SELECTED YEARS

Percentage of Farms	Cumulat	Cumulative Land Shares for:				
1000 15224 2020 Page 1000 bar	1960	1972	1980			
owest 10 percent	0.46	0.88	0.72			
" 20 "	1.44	3.02	2.38			
30 "	3.00	6.07	5.32			
40 "	6.00	10.60	9.68			
50 "	9.72	16.09	14.97			
u 60 u	15.79	24.41	22.78			
70 "	24.91	32.75	30.86			
<b>#</b> 80 "	36.68	44.88	42.88			
" 90 "	53.76	59.14	57.52			
ALL FARMS	100	100	100			
	Land Conc	entration Ra	tios			
) Farm Area basis	0.62	0.54	0.55			
(2) Cultivated Area basis	0.58	0.47	0,50			
3) Cropland Area basis	0.53	0.42	0.45			
			and the same of the same			

cources: / 22 / estimates based on data from Pakistan Censuses of Agriculture, 1960, 1972, 1980.

farm incomes on the basis of revenues per acre, Chaudhry /16/ has shown that incomes on Smaller farms increased more rapidly during the period of the green revolution. He has also shown, on the basis of the same data source that the incomes of tenant farmers grew at the much faster rate of 19.3 per cent per annum as compared to 15.2 per cent per annum for landlords and 13.3 per cent per annum for peasant proprietors.

Chaudhry \_\_16\_/ has also presented evidence to show that the demand for labour actually increased during this period. He has presented data to show that real wages increased appreciably in the rural sector during this period. Although the reliability of this evidence cannot be directly tested because of the absence of data from other sources, given the higher intensity of labour use in the small farms and the higher rates of growth of output reported for this sector, it may be concluded that labour use would have increased over this period.

The evidence reviewed so far suggests that rural income inequalities declined during the 1960s. However, there was an increase in income inequalities during the 1970s. In a study of survey data of 750 households in 8 villages of the Punjab, Sind and NWFP, published in 1984, Ercelawn / 21 / has attempted to determine the sources of rural income inequality. He finds that the dominant source of income inequality was generally crop income. Wage employment, nonfarm enterprise and livestock were inequality-mitigating sources, and in two sample villages with substantial overseas migration, remittances accounted for the dominant share of overall income inequality. The study found that the major contributor to income inequality was unequal access to land and this accounted for the worsened relative income position of the landless. Poverty in the sample villages was found to be primarily borne by those with disadvantaged access to land.

### 4. RURAL POVERTY

The review of studies on rural income inequalities reveals a declining trend during the 1960s and an increase in income inequalities during the 1970s. However, it should be borne in mind that the studies based upon the Household Income and Expenditure Surveys suffer from the data problems associated with the large sampling errors in the highest income category arising out of the small coverage of this category in the surveys. Moreover, a decline in income inequality measured in terms of the gini-coefficients does not necessarily imply a decline in poverty as well. There have been a few studies done in Pakistan that focussed directly on the problem of poverty. The studies by Naseem [62], Allauddin [5], Mujahid [60], Wasay [88], Naseem [63], Irfan and Amjad [37] and Cheema [17] analyse the problem of poverty in Pakistan. The study by Wasay [88] was directed only at attempting to estimate a poverty line for the urban sector while the other studies evaluated the problem for both the urban and rural sectors or specifically for the rural sector and are therefore of direct interest to us.

The pioneering study by Naseem / 62 /, on the basis of arbitrarily assumed poverty lines in terms of per capita annual expenditures of Rs.250 and Rs. 300 at 1959-60 prices, estimated the incidence of rural poverty in Pakistan for the years 1963-64 to 1969-70. The study by Allauddin / 5 / extended Naseem's work to the year 1971-72 and also presented estimates based upon a poverty line computed on the basis of per capita income as well as per capita expenditure. In addition to presenting estimates of rural poor on the basis of the two poverty levels arbitrarily determined by Naseem, Allauddin also presented estimates based upon poverty lines on the basis of Rs.225 and Rs.350 per capita expenditure and income. Since these poverty lines are arbitrarily determined, setting lines below and above those considered by

Naseem enables one to get an idea of the sensitivity of the results to the definition of the poverty criteria. Naseem found that the number of rural poor according to the lower poverty level decreased significantly, however, according to the upper level of poverty the numbers increased during the period of his study. In terms of the number of households his results indicated a decrease in poverty in terms of the lower criterion and nearly constant proportions in terms of the upper criterion. Allauddin came to similar conclusions. However, she has shown that the estimates of the poor are sensitive to the determination of the poverty criterion. For example, although rural poverty seems to have declined on the basis of the Rs.250 per capita per annum consideration of the Rs. 350 criterion indicates an increase in rural poverty. The estimates from Allauddin's study in terms of real expenditure and real incomes are presented in Tables II.3 and II.4. A perusal of these tables reveals that the poverty problem is much more severe when considered in terms of real income. It can be seen that the percentage of rural population classified as poor is greater at the lower defined levels of poverty in terms of real income as compared to real expenditures in nearly all cases. Allauddin also computed calorie intakes for the urban and rural sectors. She found that while calorie intake in the rural sector was higher it declined over the period of her study from an average of about 1988 calories to 1898 calories. The calorie intake in the urban sector, however, remained constant at about 1700 calories.

The implicit assumption underlying the studies by Naseem and Allauddin that the income or expenditure of each individual in each income category is equal to the average for that category has been seriously challenged by Mujahid \( \sigma 60 \) \( \sigma \). He uses linear interpolation of the grouped data to come up with results that are in complete contrast to those obtained by Naseem and Allauddin. For example he finds that the incidence of poverty is comparatively

TABLE II.3

# INCIDENCE OF POVERTY IN RURAL AREAS IN TERMS OF REAL EXPENDITURE

Numbers in Millions Percentage of Rural Population Classified as Poor Corresponding to Poverty Lines of Rs.225 Rs.250 Rs.300 Rs Number of Poor Persons Corresponding to Poverty Lines of ears Rs. 250 Rs.300 | Rs.350 Rs. 225 1963/64 1966/67 1968/69 83. 60.50 9.97 16.53 43.1 23.20 31.83 26.0 6.15 13.13 24.49 32.0 59.7 80. 32.86 15.0 75. 61.5 4.19 10.76 26.37 31.66 10.0 25.1 1969/70 26.0 59.7 73. 11.40 26.18 32.14 1970/71 4.15 24.59 9.3 54.8 81. 36.70 1971/72 0.09 8.82 26.83 40.16 0.2 19.1 58.4

Source: Allauddin / 5 /.

TABLE II.4

INCIDENCE OF POVERTY IN RURAL AREAS IN TERMS OF REAL INCOME

ears		of Poor to Pove		Percentage of Rural Population Classified as Poor Corresponding to Poverty Lines of				
and the same of th	Rs.225	Rs.250	Rs.300	Rs . 350	Rs.225	Rs.250	Rs.300	Rs . 350
63/64	12.86	21.66	25.83	30.69	3 <b>3</b> .5	56.5	67.4	80.0
66/67	6.93	12.65	20.01	25.19	15.6	30.8	48.8	61.4
68/69	9.97	15.81	27.37	32.84	23.2	36.9	63.8	76.6
69/70	9.19	15.62	26.79	35.89	21.0	35.6	61.1	81.8
70/71	5.22	12.82	26.99	35.53	11.6	28.6	60.1	79.2
71/72	8.87	19.10	29.77	39.99	19.3	41.6	64.8	87.0

Source: Allauddin / 5 /.

lower in the earlier years than reported by the previous studies.

Also the earlier studies reported a decrease in rural poverty over time. Mujahid found that rural poverty had in fact increased.

A comparison of Mujahids results with those of Allauddin on the basis of per capita income are presented in Table II.5.

In a later study into rural poverty Naseem / 62 / has estimated the poverty line on the basis of pooling cross-section and time-series data for five years for which data were available and regressing average per head calorie intake on average per head total monthly expenditure. Two types of problems arising out of this method have been highlighted by Irfan and Amjad  $\frac{737}{1}$ . The first relates to the implicit assumption that the intercept and slope are constant across different cross-sections and the second to the fact that to the extent that the share of food in total expenditure varies inversely with the income of the household the regression equation embracing all households would tend to overestimate the poverty line. Naseem's .estimates of rural poverty based upon three lines of 95 percent of minimum, 92 percent of minimum and 90 percent of minimum requirements of 2100 calories are presented in Table II.6. An important feature of the data in Table II.6 is that it shows an increase in rural poverty during the 1970s.

In a recent study of rural poverty in Pakistan, Irfan and Amjad / 37 / have used the basic caloric requirement of 2550 calories per day per adult computed by Khan and Khan / 49 / for determination of the poverty line. They converted household size by membership into adult equivalents on the basis of nutritional scales

Source; Nasgem / 68 Jr

TABLE II.5

# RURAL POVERTY ESTIMATES BASED ON PER CAPITA INCOME PER ANNUM

**************************************			vocal analysis		( Pe	r Cent)	
Year	Below Rs.225	Below	Below Rs.250		Rs.300	Below Rs.350	
iea;	(Talat,1975)	(Talat, (Mujahid, 1978)		(Talat, Mujahid, 1975) 1978)		Talat,1975	
1963/64	33.5	56.5	29.2	67.4	41.6	60.0	
1966/67	15.6	30.8	40.6	48.8	55.8	61.4	
1968/69	23.2	36.9	Street sylt	63.8	Re-Serves	76.6	
1969/70	21.0	35.6	39.5	61.1	52.6	81.8	
1970/71	11.6	28.6	ow Prusib	50.1	Iron leto	79.2	
1971/72	19.3	41.6	d Facing Fling	64.8	vert botter	87.0	

Sources: Talat Allauddin / 5 7 and Mujahid / 60 7.

TABLE II.6

# ESTIMATES OF RURAL POVERTY BASED ON MINIMUM NUTRITIONAL REQUIREMENTS

(Per Cent)

Year	Below Poverty Line I (95% of minimum of 2,100 calories)		Below Poverty Line II (92% of minimum of 2,100 calories)		Below Poverty Lise III (90% of minimum of 2,100 calories)	
	House- holds	Popu- lation	House- holds	Popu- lation	House- holds	Popu- lation
1963/64	79	72	62	54	54	45
1966/67	73	64	63	52	55	44
1968/68	74	64	63	53	56	46
1969/70	76	68	56	46	45	36
1970/71	79	71	58	47	48	38
1971/72	82	74	65	55	54	43

Source: Naseem / 63 7.

depicting caloric requirements of different age groups. The authors nave used the 1979 Household Income and Expenditure Survey. Their study, however, suffers from a deficiency. Because of the nonavailability, at the time of analysis, of data on quantities of food items consumed in 1979, they have used the 1971-72 survey data to arrive at food expenditures per adult. Cheema / 17 / has in a recent unpublished paper adjusted for this deficiency by using data on quantities of food items consumed based on the 1979 survey. These data were made available after the study by Irfan and Amjad. The main results of Cheema's  $\underline{/}$  17  $\underline{/}$  study are reproduced in Table II.7. It is difficult to compare these results with those obtained by the earlier studies because of the differences in the methodology employed to construct the poverty line. Cheema's estimates show that nearly 39 percent of the rural population and over 35 percent of the rural households can be classified as poor. The percentage distribution of poor population by provinces reveals that over 62 percent of the poor reside in the Punjab Province. It should be borne in mind that the Household Income and Expenditure Survey of 1979 did not cover a major portion of the NNFP. This would lead to a serious under estimation of the poor in NNFP in particular and the country in general.

# 5. CONCLUSIONS

There are serious data problems and differences in methodologies that make the comparison of results across studies difficult. The general patterns that emerge are a tendency towards a reduction in income inequalities during the green

TABLE II.7

TABLE II.7
POVERTY BASED ON HOUSEHOLD AND POPULATION -1979 INCIDENCE OF RURAL P

Provinces	Incidence of Rural Ho	Incidence of Poverty Based on Rural Households	ou	Incide	Incidence of Poverty Based on Rural Population	/ Based on
and Prince of the Control of the Con	Percentage Distribu- tion of All Households	Poor House- holds as Per- Distribu- centage of tion of PC Total	Percentage Distribu- tion of Poor Households	Percentage Distribu- tion of Population	Poor Popula- tion as Per- centage of Total	Percentage Distribu- tion of Poor Population
Punjab	60.67	36.95	63.82	59.70	40.60	62.29
Sind	17.64	33.66	16.91	18.54	39.48	18.81
N.W.F.P	15.64	29.94	13.32	15.90	31.71	12.96
Baluchistan	6.05	34,55	5,95	5.86	39.47	5.94
ALL PROVINCES	100	35.13	100	100	38.91	100

1979 Survey d Expenditure S m Household Income and E from on data Based 2017 1000 100 Source:

revolution period followed by an increase in income inequalities during the 1970's. The patterns in rural poverty reveal an increase during the 1960's followed by a decrease during the 1970's. It should be borne in mind that Mujahid's study  $\sqrt{60}$  Twhich revealed un increase in poverty, was based upon an adjustment of the basic data. Moreover the determination of the poverty line is crucial to the estimates of the numbers of poor.

The dominant source of income inequality was generally found to be crop income. Wage employment, non-farm enterprise and livestock were found to be inequality-mitigating factors. Remittances from overseas migrants were responsible for some alleviation of poverty. However, these could also have been serious contributors to income inequality.

# CHAPTER III

# DOMESTIC TERMS OF TRADE AND PUBLIC POLICY FOR AGRICULTURE IN PAKISTAN

Despite the critical role of intersectoral terms of trade for a host of public policy issues, the official statistical system in Pakistan is not designed to generate a statistical series of the terms of trade of the agricultural sector on a regular basis. A number of views expressed on Pakistan's agriculture appear to be based either on results of diverse studies that are now outdated or on a complete neglect of empirical evidence. This chapter analyses existing data that could be processed to calculate the terms-of-trade indices. An attempt is made to bring together the existing information on the movement of terms of trade for the agricultural sector for the period 1951/52 to 1983/84. The impact of changes in terms of trade on farm output, distribution of income and the efficient use of resources is also traced.

# 2. BRIEF REVIEW OF PREVIOUS STUDIES

A number of studies have been conducted in Pakistan about movements in terms of trade of the agricultural sector \$\int\_{18,31,43,54,55}\$\,\bigceptilon\$. Since there are considerable methodological differences, it would be useful to briefly review these studies. Two distinct categories of studies can be identified: (a) those studies which examine the movements in intersectoral terms of trade by computing implicit price indices from the national accounts data; and (b) those studies which evolve a set of

weights for different items traded between the agricultural and non-agricultural sectors. The study by Cheong and D'Silva / 187 falls in the first category while all the remaining studies belong to the second category.

Studies belonging to the first group are not demanding regarding the basic data. Published National accounts data and/or published indices of wholesale prices for agricultural and industrial goods are used to discern the trends in terms of trade. Cheong and D'Silva work out the terms of trade by using the estimates of GDP at factor costs at current prices originating in the agricultural and manufacturing sectors and their corresponding estimates at constant prices 1. The terms of trade worked out from GDP deflators suffer from serious limitations. The weights attached to different commodities are on the basis of production and not on the basis of marketed quantities of intersectoral sales and purchases. Furthermore, the commodities included in the analysis are not the ones actually traded between the agricultural and non-agricultural sector but are inclusive of many commodities which are in fact not traded between the two sectors. On these grounds, the findings regarding the terms of trade may be biased and may not reflect trends in relative prices.

The second group of studies attempts to rectify the limitations inherent in the study by Cheong. The pattern of trade is identified to include most major products and weights are estimated on the basis of the best available information regarding the sales and purchases of a

The net barter terms of trade of the agriculture sector are computed by dividing the GDP deflator of the agricultural sector by the GDP deflator of the manufacturing sector.

sector for which terms of trade are computed. The studies by Lewis and Hussain / 557, Lewis /547 and Gotsch and Brown /317 are identical in resp of the selection of commodities, choice of weights and use of prices. The study by Kazi /43 / uses the same concept of prices but differs in the way weights are assigned to different commodities and in the coverage of commodities in the intersectoral trade.

The commodity coverage of the prices paid indices computed by Lewis was spread over three functional groups of consumption goods, intermediate and related goods and investment and related goods. The number of items in groups identified above were respectively 14, 7 and 6. Since no information on intersectoral trade was available, Lewis estimated the value of intersectoral transactions through an indirect method. He derived the value of intersectoral transaction by estimating the availability of different commodities and apportioning it between the two sectors on the basis of different assumptions about the absorption of different commodities in two sectors. The net availability was defined as domestic supply plus imports minus exports. The absorption of consumption goods in the agricultural and non-agricultural sectors was determined by different assumptions about the consumption pattern of agricultural and non-agricultural population. The alternative assumptions were equal per capita expenditure, 10%, 25% and 40% less expenditure on non-agricultural consumption goods in the rural areas as compared with urban areas. For the intermediate and investment goods, a smaller share was allocated to the agricultural sector as bulk of such goods is assumed to be absorbed in the urban industrial sector and in public projects

The weights computed by Lewis were based on estimated production, sales and purchases for 1959/60. All terms of trade indices are thus representative of trends of relative prices of bundles of good transacted in 1959/60. Lewis had computed the terms of trade indices on the basis of of a number of alternative weighting schemes and found that the results were robust and the basic trends were insensitive to substantial variations in the weights. This finding is extremely important as a fixed base index with base year weights soon runs into difficulty as it does not capture the effects of the changing composition of trade on the price indices.

Kazi finds three problems with Lewis's method of analysis.

First, she argues that Lewis's weighting schemes were based on arbitrary assumptions about the absorption of goods in different sectors. Second, she objects to the inclusion of some investment and investment related goods on the ground that they are unlikely items for purchase by agriculture. Third, she points out that Lewis ignores items purchased by agriculture from the non-agricultural sector that are not produced by the large scale industrial sector. Expenditure on gas and electricity are cited by her in this regard.

<sup>2/</sup> The robustness found by Lewis for his method of analysis also obtains for the extended period of analysis to 1983/84. For details, the reader is referred to the Annexure to this chapter. For a ready reference on this point in the case of two alternative weighting schemes, the reader may see Table 3.1. It would be extremely useful to extend the analysis using a recent year's pattern of intersectoral sales and purchases as weights for the terms of trade. Unavailability of basic data is the main factor explaining our decision not to pursue the ideal course but to stick to Lewis's weights. The assumption that rural per capita expenditure on most commodities is 25% less as compared to urban per capita expenditure is supported by the evidence for 1979 from the Household Income and Expenditure Survey. However, the important point to remember is that a wide variation in the weighting scheme does not change the pattern of movement of terms of trade for the agricultural sector.

Before we examine the contribution made by Kazi, it must be pointed out that she has correctly identified the problems with Lewis's analysis but has not dealt with these problems very adequately. Using the Household Income and Expenditure Survey data on consumer expenditure and National Accounts data on the value of production of some agricultural and non-agricultural intermediate goods, she estimates the weights for different items for 1967/70. Like Lewis, she uses index numbers of wholesale prices as price indicators.

Derivation of weights on the basis of consumer expenditure entails a bias as has been shown in the case of India by Kahlon and Tyagi  $\sqrt{40}$ . Such data are based on retail prices. The weights derived on the basis of final consumption estimates tend to over-estimate the share of those commodities where — the difference between the retail and wholesale prices is large.

Kazi also faults by ignoring items of capital formation which were included, though in a somewhat crude manner, by Lewis in his computations. Since investment goods are becoming increasingly important in modernizing agriculture, this omission by Kazi is a serious one. The coverage of commodities for final use by agriculture was also narrower in Kazi's study. Our interest in the computation of terms of trade of agriculture is to get an idea of changing domestic incentives for the sector and in analysing the impact such incentives have on the agricultural economy. Inclusion of a large number of items for final consumption, intermediate use and capital formation in the prices paid indices is an absolute necessity. Since Lewis has an edge in this regard and his method of analysis was found to be

robust to a wide variation in weights, we use weights constructed by him for 1959/60 in the next section to trace the movements in relative prices for the agricultural sector for the period 1951/52 to 1983/84.

#### 3. MOVEMENTS IN AGRICULTURE'S TERMS OF TRADE

The terms of trade of agriculture relative to the industrial sector are an indicator of the profitability of agriculture and of the purchasing power of agricultural income. The intersectoral terms of trade are determined jointly by changes in the supply and demand for goods and services entering in the intersectoral trade, by changes in a whole array of macro policies in areas of taxation, trade and monetary economics and by commodity-specific incentive price policies. In this section, we present information on trends in domestic terms of trade and give a brief account of the factors that may have influenced the rural-urban terms of trade.

There are many different concepts of terms of trade. Table 3.1 presents information on three types of terms of trade 3.1. All numbers are three-year moving averages that attempt to smooth the series from yearly fluctuations. Net barter terms of trade are measured in two alternative ways. For the prices paid by agriculture, estimated purchases are used in both alternatives while for the prices received by agriculture, alternative weights are based on marketings and gross output of different agricultural goods. The use of different weights changes the magnitude of movements but does not alter the general pattern of movements in terms of trade.

 $<sup>\</sup>frac{3}{1}$  The reader is referred to the Annexure to this chapter, for details regarding the method of analysis.

#### TABLE III.1

#### TERMS OF TRADE FOR AGRICULTURE

Three Years Moving Average: 1951/52 - 1983/84

Years		erms of Trade	Income Terms of Trade	Single Factoral terms of Trade		eceived Re aid by Agr f		Food Crops Relative to Cash	Index of Agricultural Output
	Alternative 1	Alternative 2	ATTACK TO THE PARTY OF THE PART	1 6 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Consump- tion goods	Inter- mediate goods	Invest- ment goods	Crops	(Base 1959/60)
	1	2	3	4	5	6	7	8	9
1951-54	99.34	96.64	83.60	109.21	100.30	99.75	90.85	87.16	87.00
1952-55	91.60	91.59	81.86	102.07	90.78	95.62	92.93	94.88	89.67
1953-56	90.12	87.97	79.42	96.55	89.70	92.73	94.65	87.02	90.33
1954-57	94.16	91.17	82.78	95.83	96.08	92.12	92.74	84.62	90.67
1955-58	98.56	95.14	87.34	96.83	101.75	93.83	93.90	87.41	91.67
1956-59	100.64	98.16	92.25	97.18	104.63	97.29	95.45	92.22	94.00
1957-60	100.88	99.37	94.74	98.56	103.29	99.46	98.61	96.01	95.33
1958-61	103.44	103 11	102.44	99.39	105.23	102.21	104.06	98.43	99.33
1959-62	106.11	105./1	111.73	101.02	106.98	105.13	106.24	96.98	105.67
1960-63	107.99	106.80	116.60	102.20	109.66	105.66	105.39	93.44	109.33
1961-64	106.91	105.37	120.49	104.04	109.82	104.41	100.79	95.41	114.33
1962-65	105.93	106.98	126.40	107.58	109.14	104.41	98.00	96.44	118.00
1963-66	104.55	108.43	133.74	108.60	107.94	103.76	96.49	97.11	123.33
1964-67	104.46	113.60	146.01	115.47	107.51	105.71	96.11	106.33	128.33
1965-68	101.69	113.78	158.98	121.63	105.12	103.44	90.57	124.69	139.67
1966-69	99.37	113.02	172.41	128.09	103.39	100.50	85.11	137.00	153.33
1967-70	96.61	109.33	186.16	138.25	103.13	96.26	78.63	136.25	170.33
1968-71	97.74	108.56	191.44	141.84	106.26	95.85	76.54	126.31	176.00
1969-72	99.42	110.38	199.79	146.46	109.05	96.07	77.67	123.56	181.00
1970-73	102.38	112.26	204.05	147.03	112.10	99.30	80.34	120.73	181.67
1971-74	108.67	118.56	224.50	156.33	117.52	109.52	83.81	121.16	189.00
1972-75	109.72	121.17	230.84	156.79	117.96	119.59	77.48	127.64	190.33
1973-76	106.98	118.04	228.86	151.58	114.46	120.55	72.22	129.42	194.00
1974-77	108.84	114.77	225.23	145.85	108.92	124.77	68.71	126.06	196.33
<b>1975-</b> 78	109.23	119.54	243.90	153.95	111.54	134.26	76.33	123.45	203.66
1976-79	111.69	119.12	257.38	150.96	113.65	140.11	75.56	118.90	210.33
1977-80	105.57	115.84	255.86	145.01	111.99	123.84	71.86	121.94	222.33
1978-81	95.87	103.98	244.03	131.87	108.20	100.01	63.86	117.53	235.66
1979-82 1980-83	91.45 92.36	99.82 99.41	247.50 256.81	136.56 134.69	107.09 107.60	86.30 84.22	62.36 66.07	121.44	248.00

#### FOOT NOTES:

- 1. Weights for prices received by agricultural sector are the marketings and gross value of output of each of the commodities for Alternative 1 and Alternative 2 respectively. Weights for prices paid by the sector are estimated purchase of non-agricultural commodities by the agricultural sector. The absorption of consumption is determined on the assumption that per capita consumption in agriculture is 25 per cent less than the non-agricultural sector. For the years 1951-54 to 1961-64, data series have been taken from Lewis and Mushtag years 1957. For the remaining years the series were updated using Lewis-Mushtag methodology.
- Income terms of trade were obtained by multiplying net barter terms of trade
   (Alternative 1) with an index of agricultural output. The index of agricultural
   output is published by the Federal Bureau of Statistics in its Monthly
   Statistical Bulletins.
- 3. Single Factoral terms of trade were obtained by multiplying net barter terms of trade (Alternative 1) with the Factor Productivity Index. The productivity index is taken from Wizarat / 91/. An aggregate input index was calculated by weighting the labour index, livestock index and land index. The productivity index is obtained by dividing weighted input index by index of value added in agriculture.
- 4. Weights for prices received and prices paid are the same as given in note (1) for net barter terms of trade (alternative 1).
- 5. Weights for agricultural prices are the marketings of each of the commodities. Food crops consist of Rice, Wheat, Maize, Barley, Sorghum, Pulses, Potatoes and Onions whereas cash crops consist of Oilseeds, Cotton, Sugarcane and Tobacco.

Five distinct periods in the movements in the net barter terms of trade can be distinguished. The first period from the year 1951/52 to the years 1954/57 was that of deterioration in the terms of trade when these terms declined by about 9 per cent. The partition of the subcontinent in 1947 had disrupted the pattern of trade of agricultural and manufactured goods. The areas that constituted Pakistan were surplus in agricultural goods and had exchanged these goods for manufactured goods from areas that became India. A relative glut of agricultural goods and scarcity of manufactured goods explain to a large extent the downward movement of terms of trade for the agricultural sector in this period. Trade policy, adopted in Pakistan to deal with the foreign exchange crisis, was an additional important factor for the movement of terms of trade against agriculture.

The second period from 1954/57 to 1965/68 was one of rising relative prices for the agricultural sector as the terms of trade showed an improvement of about 29 per cent over this period. The spurt in the agricultural output and a relative slow-down of the manufacturing sector output were responsible for an improvement in the terms of trade. The introduction of subsidies on some selected farm inputs and the fixation of support prices for a few major crops in the early 1960s were responsible for effecting an improvement in the barter terms of trade.

Increases in wheat and rice output as a result of the green revolution and the mounting bill for the treasury, on account of subsidy for farm inputs, had convinced the government of the need to moderate the price

 $<sup>\</sup>frac{4}{}$  For a detailed analysis, see  $\frac{1}{}$  53.54,55,317.

increases for crops and to reduce the level of subsidies on farm inputs. The slight downward trend in the net barter terms of trade noticed for the period 1965/68 to 1968/71 is a consequence of the government's efforts to force agriculture to share its productivity gains with the rest of the society.

The devaluation of the rupee in 1972 and increases in the rate of subsidies on farm input in the early 1970s had imparted an upward trend to the indices of the terms of trade. The improvement in the terms of trade of about 10 per cent, between 1968/71 and 1975/78 can be explained

in terms of the changes in the rate of foreign exchange as well as adjustment of sectoral prices of agricultural inputs and outputs in response to changes in world prices. A deliberate policy of the removal of subsidies on farm inputs accompanied by an increase in the support prices for major crops has been in force since the late 1970s. The downward trend in the terms of trade since 1979 is, in part, a result of this deliberate policy choice. The examples given above, from the economic history of Pakistan illustrate the crucial role played by both macro economic policies and sectoral policy initiatives in the determination of trends in terms of trade.

Estimates of terms of trade at a more disaggregated level point to considerable differences in the pattern of price changes between consumption goods, intermediate goods and investment  $goods \frac{5}{}$ . The prices of investment goods have risen relatively and, for most years, the terms of trade are adverse relative to 1959/60. The trends in

It may be useful to note that the terms of trade for an entire sector, for selected groups of commodities, and intra-sectoral transactions address different analytical and policy issues.

terms of trade for intermediate goods and consumption goods are parallel to the overall sectoral terms of trade. The improvement in terms of trade for intermediate goods is higher than for consumption goods. The prices of food crops relative to those of cash crops show considerable variation through time. Relative food prices were low till 1965/68, rose and then again fell to a low level in 1976/79. A sharp rise in relative profitability for production of one type of crop relative to another category is also noted.

The trends in single factoral terms of trade are more or less parallel to the trends noted for the net barter terms of trade. The only difference is that the rise in single factoral terms was much sharper from 1963/66 to 1972/75.

The net barter terms of trade of the agricultural sector and of the food crop producers indicate incentives for agriculture and the food sub-sector respectively. The barter terms indicate that one group's benefits are the other group's losses and the extent of the loss/benefit is measured by the deviation from the unit level. The income terms of trade measure the purchasing power of a sector. The deviation from the unit level in the case of income terms of trade of a sector does not necessarily imply a worsening of the purchasing power of the rival sector.

The income terms of trade are defined as the ratio of value of sales by a sector to its average import price. Since no data series exist for the marketed surplus, we have measured the income terms as a product of net barter terms of trade and an index of agricultural output. The income terms of trade remained depressed till 1957/60 and show an

increasing trend afterwards. In fact, these terms register a decline for the period 1951/52 to 1955/56. The explanation of the trends observed in the income terms of trade lies in the movements of its two components, the net barter terms of trade and physical agricultural output. An increase in out. ut with no change in relative prices increases income terms, while a movement of the internal terms of trade against agriculture ceteris paribus reduces the income terms of trade. The agricultural sector was squeezed through the declining internal terms of trade during the 1950s. For the later years, the purchasing power of agriculture shows an increasing trend. This is mainly due to productivity gains of agriculture. Increases in physical agricultural output more than offset the impact of declining barter terms on the income terms of trade, for years identified above, when barter terms showed some declines.

#### 4. SELECTED ASPECTS OF THE IMPACT ON RURAL ECONOMY

Active price intervention showing variation in form and intensity generates many critical policy issues. Some of these issues are: impact and role of price incentives on farm production, efficiency of resource use and distribution of income.

#### 4.1 Price Incentives and Aggregate Farm Output

High prices, in theory, may have implications not only for efficient use of resources but can shift the production function upwards by price-induced technological and institutional innovations and infrastructural investment in rural areas. In order to examine the impact of terms of trade on farm output, the generally used linear

relation between the index of agricultural output, net barter terms of trade lagged by a year and supply shifter variables was estimated.

The equation estimated for the period 1951/52 to 1983/84 is as follows:

$$Q_t = -13.72 + .75 Q_{t-1} + .28 P_{t-1} + 0.07 Z_t + 1.12 T$$

$$(-0.63) (6.05)** (1.55)* (2.80)* (1.40)**$$

$$\overline{R} = 0.92$$

Where Q, P, Z and T are respectively an index of agricultural output, terms of trade, percentage of net sown area irrigated and time trend.

(Figures in the parenthesis are t-values of regression coefficients and (\*\*) and (\*) indicate coefficient significant at 1 and 10 percent level respectively).

The terms of trade have a positive effect on output. By the conventional significance criterion, the price coefficient is marginally significant. The short-run price elasticity of .18 and the long-run price elasticity of .72 calculated at mean values are in the range found for other developing countries.

The first shifter variable, the proportion of net sown area irrigated, captures the impact of price-induced technical change on farm output. The spurt in the installation of private tubewells, which began around 1959/60, has been attributed by some analysts to the profitability of additional water \[ \sigma 59 \]. The profitability of water was, in turn, linked with the pricing policy regarding both output and inputs of the agricultural sectors. Some analysts have demonstrated the link between the farm prices and public investment in agriculture \[ \sum 147 \]. Higher prices for the agricultural produce increase the financial rate of return

on agricultural projects and justify increased allocations for the agricultural sector. The significant positive coefficient of the irrigation variable shows the importance of price-induced innovations.

The coefficient of the time-trend variable measures the impact of autonomous technical change on farm output. Without minimizing the role of price-induced innovations, it can be argued that basic scientific knowledge is weakly related with prices and has its own growth momentum. Its beneficial impact on farm output is evident from the significance of the coefficient of the time-trend variable in the estimated equation.

#### 4.2. Trade Policy and Resource Use

The implication with respect to the allocation efficiency from the society's vantage point can be spelled out after we know the extent of correspondence of private signals transmitted to farmers with the short and long run social economic benefits and costs as measured by world border prices.

Gotsch and Brown /31/ have documented the pervasive impact of trade interventions in distortions of incentives for the agricultural producers in Pakistan. The nominal and effective protection coefficients for major crops show domestic prices of most crops as being lower than the world prices. In the case of most industrial goods, the domestic prices are higher than the world prices. In this sense, Pakistani policymakers have undervalued agricultural production. The disaggregated analysis by crops and by different time periods shows that incentives vary

rop and, for different crops, over time. The food crops were d while export crops were taxed. Sugarcane, wheat and maize ided considerable protection till about early 1970. Considerfood self-sufficiency seem to have determined this policy here is also a distinct oreak in the pattern of incentives it 1972/73. The devaluation of the rupee in May 1972 had the border prices. Abrupt and wide fluctuations in world commodity ring the 1970s and 1980s have imparted instability to of protection coefficients. Despite the instability observed, movement of the protection coefficients of most crops e value of one is evident during the last few years.

e evidence on the presence or absence of distortions arising policy is only the starting point in the evaluation of implications. The measurement of the cost of distortions in conomic welfare is the next logical step. Research on: is totally lacking in Pakistan. The estimation of demand curves as a basic input in the measurement of consumer and impluses is the first requirement. The question of whether much increased agricultural incomes and employment would ted if world prices had been adequately reflected in centives deserves a high priority for research.

#### ct on Distribution of Income

relationship between agricultural pricing policies and n is complex and has neither been modelled adequately nor sjected to detailed empirical enquiry. Some attention has

Marian Services

been given to the relative sectoral distribution issue and of personal incomes while none or very little attention hato the impact of pricing policies on the regional distributione.

Agricultural prices determine the income of the faffect the living standards of the people engaged in farmi professions as agricultural commodities form an important wage goods. A controversy rages among economists as to watransfer of income takes place between sectors or between agricultural producers and low-income urban and rural consagricultural producers and low-income urban and rural consagricultural producers and that in India high farm prices has income from urban areas to rural areas and that all groups areas have benefitted from high farm prices. Ashok Mitrabelieves that transfer has taken place from low-income urban consumers to high-income agricultural producers. The limithat we present for Pakistan provides some support to the that the pricing policy has primarily resulted in intersect of income.

The notion that high farm prices benefit the large burt the landless labour and small farmers is based on two (1) labourers depend on the market to purchase the wage go incomes of the wage labour and small farmers are independent prices. The evidence we present below indicates that these do not hold in their pure form.

Table 3.2 shows sources from which rural households obtain wheat flour. The reliance on the market for wheat and wheat flour shows considerable variation between provinces and, within each province, between farm and non-farm households. Own farming and wages in kind are the dominant sources for wheat flour for farm households and for all households. Even in the case of non-farm households these two sources are important. High farm prices imply an automatic increase in income for the component accounted for by wages in kind and own farming. The assumption that income and farm prices are independent is clearly violated. The fact that percentage of flour obtained through market for all rural households is 25 per cent in Punjab, 24 per cent in Sind and 48 per cent in N.W.F.P. shows that dependence on market is not high.

TABLE III.2

PERCENTAGE DISTRIBUTION OF WHEAT FLOUR BY SOURCES
FOR TYPE OF HOUSEHOLD AND PROVINCE

remat son	Type of Household	SOURCES OF WHEAT FLOUR										
na (grados	ligeresulved to laters	Own Farming	Wages in Kind	Open Market Wheat	Ration Flour	Open Market Flour						
Punjab	All Households	55	12	18	8	7						
	Farm Households Non-Farm Households	82 20	2 26	6 34	6 <b>1</b> 0	4 10						
Sind	All Households	73	2	6	1	18						
	Farm Households Non-Farm Households	89 7	- 9	4 16	odsi (1)	7 68						
N.W.F.P	All Households	42	11	1	-	47						
	Farm Households Non-Farm Households	50 31	9 13	1	-	41 54						

Source: The Survey of Wheat Markets conducted in 1982 by Pakistan Institut of Development Economics, Islamabad, and Centre for Development Planning, Erasmus University, Rotterdam.

The data on trends in rural real wages further casts doubt on the hypothesis that landless labourers may loose as a result of high farm prices. Guisinger and Hicks / 357 and Irfan and Ahmed / 38/ have provided a series of rural wages for selected years between 1952 and 1973 and for each year between 1973 and 1984. There was a pronounced upward trend in real wages between ,1952 and 1973. Real wages for casual workers in 1973 were higher by about 60% over the benchmark year of 1952. The series constructed by Irfan and Ahmad show declines in real wages between 1974-76, an upward level for years 1976 to 1981 and a slight downward trend since 1981. The close correspondence between real wages and net barter terms of trade for agriculture again shows the salutary effect of high prices for the rural income of landless labour in the agricultural sector. The conclusion that we reach then is that the interests of large farmers, small farmers and landless wage labour in rural areas are more or less identical as far as farm prices are concerned. It must, however, be noted that the similarity in interests of these groups obtains in the long-run after the price incentives have had their impact on the income of the poor through increased job opportunities. In the short-run high food prices impose a burden on the poor consumers. A role for targetted food subsidies for the benefit of the poor is obvious.

The incidence of high farm prices on urban income distribution has also aroused a controversy. Some believe that high food prices hurt mainly the low-income urban consumers. Brown / 147 has shown that in the case of Pakistan urban wage levels have responded fairly quickly to the prices of wage goods. If Brown is correct, high food prices may have more impact on urban profits than on real incomes of wage earners. In any case, the

analysis of high food prices needs to take account of the national food subsidy schemes that have a dual pricing system and funnel the bulk of rationed food to urban areas. Naqvi and Cornelisse / 61 / in their study of wheat marketing have shown that the rationing system as it has operated has discriminated against rural areas, especially rural N.W.F.P., Sind and Baluchistan. They have further shown that wheatmarket actors (millers, ration shop-keepers and the Food Department officials — all belonging to high or middle income classes) appropriate for themselves a part of benefits intended for the rural and urban poor. The difference between the domestic producers prices and consumer prices and the public expenditure on the administration of the rationing system require huge budgetary subsidies. The financing of these subsidies is generally regressive. The intervention by the government in the public distribution of wheat may have been beneficial to the middle and upper income consumers in the urban areas.

The impact of input pricing on income distribution and agricultural development has not been discussed so far. The case for the introduction of subsidies on modern inputs (fertilizer, pesticides, water) in early 1960s was built to familiarize farmers with the new innovations and to encourage them to use these inputs on a large-scale. There is a growing literature in Pakistan that shows that input subsidies may have outlived their original justification. Gotsch and Brown  $\angle$  31/ and Cheong et. al  $\angle$  18/ have pointed out that subsidies on water may encourage wasteful use of the scarce water supply. Subsidies on machines tend to displace labour and provide wrong signals to farmers for the use of capital-intensive technology. Since access

to inputs is largely determined by the size of holding, it is prising that farm subsidies benefit mostly the large and profermers in Pakistan. Since large farmers are not necessarily users of modern inputs, a policy that diverts inputs to small would maintain or increase the output. Research on the iden of institutional interventions that ensures larger supplies inputs to smaller farmers is of high priority. However, the point that needs to be stressed is the need for a reduction subsidies to encourage both farm production and improve incompation in the farm sector. Subsidized input prices at this subsidired in the farm sector. Subsidized input prices at this subsidired paying high market clearing prices for the subsidize which are in short supply.

#### 5. CONCLUSIONS AND POLICY IMPLICATIONS

We can be very brief in conclusion. Basic data or sales and marketings of goods and services entering in the trade need an improvement. Notwithstanding the weaknesses data, the preceding analysis sheds light on various policy-issues and has several interesting implications. These are summarized below:

The barter terms of trade of the agricultural security shown an upward trend over the entire period from 1951/52 to However, this trend conceals in it periods of considerable considerable increase and large fluctuations in the Met bar trade. There was a declining trend in 1950s, a sharp up

during most of the 1960s, large fluctuations in the decade of the 1970s and a declining trend since 1977/78.

- 2. The efficacy of price instruments as a source of agricultural development was noted. The aggregate farm output was positively related with the net barter terms of trade, irrigation ratio and the time-trend. Notwithstanding the importance of a positive price policy for agriculture, a case can be made for an active technology policy and an expanded programme of public investment benefitting the agricultural sector. The long-run viability and productivity of the Indus Basin implies vast public investments in agriculture. It should be noted that it is easier to extract surplus out of increased production than out of stagnant output. Increasing farm output requires a provision of gross resources in the form of irrigation, research, credit and other modern inputs. Starving agriculture of resources too soon may mean a large amount of foregone farm output.
- 3. A public investment programme of the type needed requires increased resource mobilization from the agriculture sector. The extent of improvement in the purchasing power of agriculture since 1953/56 is large. The farmers' ability to pay taxes and their capacity to pay for modern inputs has improved considerably. This fact should be clearly noted in debates on reducing farm subsidies or increasing the tax burden on agriculture. The agricultural sector in Pakistan has been taxed mainly by the trade policy. Direct taxes have been too low to be a major force. The farm subsidies on inputs have shown

considerable increases mainly due to explosive increase in the quantity of inputs used. Very little support can be marshalled for input subsidies from the vantage point of efficiency and agricultural development. Taxation of the agricultural sector via the trade policy is inefficient. There is a strong case for increased taxation of the agricultural sector through direct taxes on land and/or agricultural income and reduced levels or withdrawal of input subsidies.

- 4. The impact of agricultural price policies on both the output and input sides on income distribution has interesting and useful policy implications. That high farm prices benefit large producers is obvious. We had also found some support for the notion that high farm prices also benefit small farmers and landless labour. The farmers' capacity to hire labour is a function of the farm prices. Benefits from subsidized inputs tend to accrue in large measure to large producers. Reducing input subsidies would not harm the small farmer greatly but could release public resources that could finance investments for the benefit of small farmers. In this sense remunerative farm prices and low or no subsidies on farm inputs should increase the efficiency of resource use and the welfare position of the small farmer and the rural poor.
- 5. Incentives to producers could be given through attractive produce prices, subsidized prices for inputs, technological innovations and through investment in complimentary sectors. Detailed specific research on the relative benefits and costs of providing incentives through these policies are required. The theoretical and empirical aspects of determination of support prices for different crops that

provide just the right amount of incentives and maintain appropriate price relatives for different crops should be given a high priority in the research agenda. Incentives, measured by the rural-urban terms of trade, are the outcome of a host of interacting sectoral and macro policies. It is important to ensure that the commodity-specific price policies and the macro-economic policies are a consistent policy set in their impact on farm incentives.

inout sides on income distribution has a interesting and useful police and useful police and useful police devices to producers is obvious

also benefit such farmer, and landless jabour. The farmers conscribe to

input subsidies would not have the seal former greatly but could release

on re-well has control and overestanded each low on a cremet France

Ote: and the valiant position of the tasis former and the court pour.

Section 1/8 (business are produced a result of an include a recommendation of the contract of

and through investment to complimentary secture. Desailed specific

And the same required. The Should had a should be a same required.

And admin the court of the cour

#### ANNEXURE

This annexure provides the complete data-series from 1951-52 to 1983-84, that was used to construct the domestic terms of trade indices. An attempt has been made not only to update the Lewis and Musthaq / 55/ estimates, which covered the period 1951-52 to 1963-64, to 1983-84 but also to provide alternative measures that incorporate some of the criticisms levelled against the use of the net barter terms of trade index. Data for the period 1964-65 to 1983-84 are obtained from the published statistics of the Federal Bureau of Statistics and the Planning Unit of the Ministry of Agriculture.

There are twenty-two tables in this annexure. The last eleven tables (Table 12 to 22) provide the background information on price indices and alternative weighting schemes used to construct the terms of trade indices.

It should be noted that the Lewis and Mushtaq / 55/ study did not present indices for income and single factoral terms of trade. Moreover the study / 55/ provided only one measure, each, of the intrasectoral terms of trade for the agricultural sectors.

The general form and the specific weights of the price indices are obtained following Lewis and Mushtaq  $\sqrt{55}$  and their methodology is reproduced here.

The symbols used are defined below, and 1959/60 is used as base 100 for all individual price indices. The weights can be varied by letting s and b vary, i.e., by defining a different

weighting for the sale and purchase of each good by a sector in the base year, or by changing the base year.

#### DEFINITION OF VARIABLES

P <sup>m</sup> dij		index number sector i in )			(d)	price	(P)	manufactures
-----------------------	--	-------------------------------	--	--	-----	-------	-----	--------------

Pdkj is the index number of the domestic (d) price (P) of agricultural products (a) of sector k in year j.

Wsi is the percentage weight of manufactured goods sold by sector i in the base year.

wbi is the percentage weight of manufactured goods of sector i bought by the agricultural sector in the base year.

wsk is the percentage weight of agricultural goods (produced or sold, depending on specifications) of sector k in the base year.

wabk is the percentage weight of agricultural goods of sector k bought by the Non-Agriculture sector in the base year.

If the weights and prices are defined as in the Table above, then the weighted price indices for each year are as follows:

Index of prices of agricultural goods sold by the agricultural sector:

= 
$$\sum_{k} P_{dkj}^{a}$$
 .  $W_{sk}^{a}$  for each year j.

Index of prices of manufactured goods sold by the industrial sector:

$$\sum_{i}^{p_{dij}^{m}} \cdot W_{si}^{m}$$
 for each year j.

Index of prices of manufactured goods bought by the agricultural
sector:

= 
$$\sum_{i} P_{dij}^{m} = W_{bi}^{m}$$
 forteach years joost to xebut

Index of prices of agricultural goods bought by the manufacturing
sector:

= 
$$\sum_{k} P_{dkj}^{a} \cdot W_{bk}^{a}$$
 for each year j.

The domestic terms of trade of the agricultural sector for any year j, with the base year equal to 100, would be written

and the domestic terms of trade of the manufacturing sector for any year j, with the base year equal to 100, would be written

It is obvious from the above expressions for the terms of irade of each sector that even using the same (wholesale) price indices for each good whether bought or sold by a sector, that the Agriculture sector's terms of trade will not be the reciprocal of the Non-Agriculture sector's terms of trade, as long as there are leakages out of the intersectoral trade in the form of sales to

other domestic sectors or to or from the foreign sector. One further problem must be mentioned. Actual purchases by one sector of the other sector's output have not been used in computing the b weights, since data on such flows are unerailable. Instead we have divided the available supplies of each type of good on the basis of a reasonable proportion that would have been used in each sector, and calculated the percentage weights for purchases of each good on the basis of this estimated share of the total availability.

The present study in the absence of alternative data uses the same weights as used by Lewis and Mushtaq / 55 /. Like the earlier study, we also found that substantial variations of weights for the purchases of different commodities made little difference to the behaviour of the price indices.

An explanation of some of the major concepts is outlined below, followed by the Appendix tables.

- The net barter terms of trade of the agriculture sector is computed by dividing the GDP deflator of the agricultural sector by the GDP deflator of the manufacturing sector.
- 2. The income terms of trade are computed by multiplying the net barter terms of trade by the quantum index of agricultural output.
- The single factoral terms of trade are computed by multiplying the net barter terms of trade by the factor productivity index. In this case cropped area is the aggregate input index.
- 4. Intra sectoral terms of trade for manufacturing refers to the index of consumption goods relative to intermediate and investment goods.
- 5. Intra sectoral terms of trade for agriculture refers to the index of food crops relative to cash crops.

	名图 B. 在. A. B.
Table-A 3.1:	Domestic Terms of Trade for Manufacturing (Absolute)
Table-A 3.2:	Domestic Terms of Trade for Manufacturing (3 Years Moving Average)
Table-A 3.3:	Domestic Terms of Trade for Agriculture (Absolute)
Table-A 3.4:	Domestic Terms of Trade for Agriculture (3 Years Moving Average)
Table-A 3.5	Income Terms Of Trade for Agriculture
Table-A 3.6:	Single Factoral Terms of Trade for Agriculture
Table-A 3.7:	Terms of Trade for Agriculture: Alternate Method (Using Marketings as Weight for Agricultural Prices)
Table-A 3.8:	Terms of Trade for Agriculture: Alternate Method (Using GVO as Weight for Agricultural Prices)
Table-A 3.9:	Terms of Trade for Agricultural Alternate Method (Using Value Added as Weight for Agricultural Prices)
Table-A 3.10:	Intra Sectoral Terms of Trade for Manufacturing Sector
Table-A 3.11:	Intra Sectoral Terms of Trade for Agricultural Sector
Table A-3.12:	Price Indices for Industrial Goods
Table A-3.13:	Indices of Agricultural Whole-Sale Prices
Table-A 3.14:	Weights for Production, Marketings and Purchase of Agricultural Commodities.
Table-A 3.15:	Weights for Purchase and Production of Industrial Goods
Table-A 3.16:	Weighted Domestic Price Indices for Manufacturing
Table-A 3.17;	Weighted Domestic Frice Indices for Agriculture
Table-A 3.18:	Alternatively Weighted Domestic Price
Table-A 3.19:	Alternatively Weighted Domestic Price Indices for Manufacturing by Broad Categories.
Table-A 3.20:	Alternatively Weighted Domestic Price Indices for Agriculture by Major Groupings
Table-A 3.21:	Agricultural Inputs in Pakistan - 1953-54 to 1983-84
Table-A 3.22:	Aggregate Input Index, Value Added Index and the Total Factor Productivity Index 1953-54 to 1983-84 (Base Period 1959-60).

TANAMAN TANAMA

TABLE-A 3.1

DOMESTIC TERMS OF TRADE FOR MANUFACTURING SECTOR

	receive	d are w	e weight				es	are- weighe	d are we	ighed by	GVO and prices received						
	Value added	Gross value	Net avail-		cnase by			Value added	Gross value	Net avail-	Pu	rchase b	y Agricu	lture			
		of output	ability	Alt.1	A1t.2	Alt.3	Alt.4		of output	ability	Alt.1	Alt.2	Alt.3	Alt.4			
1951-52	90.48	89.03	91.14	87.05	88.80	88.53	88.57	99.88	98.28	100.61	96.09	98.02	97.72	97.7			
1952-53	103.07	101.03	101.93	100.38	101.82	101.29	102.03	100.39	98.40	99.28	97.77	99.17	98.66	99.3			
1953-54	117.98	116.57	110.77	115.96	114.69	114.91	115.00	118.55	117.13	111.13		115.24	115.46	115.5			
1954-55	112.93	113.15	111.20	114.26	311.86	113.62	111.26	115.83	116.06	114.05		114.73	116.53	114.1			
1055-56	105.04	104.48	105.33	105.55	106.65	105.27	107.19	109.46	108.88	109.76	109.99	111.14	109.70	111.7			
1956-57	99.75	99.56	102.90	99.02	100.69	100.44	100.38	102.80	102.60	106.04	102.05		103.51	103.4			
1957-58	98.48	99.19	100.12	97.37	97.47	98.18	96.74	101.98	102.71	103.67	100.82	100.93	101.66	100.1			
1958-59	100.23	100.46	100.80	99.89	99.97	99.69	100.00	101.25	101.48	101.82		100.98	100.71	101.0			
1959-60	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00		100.00	100.00	100.0			
1960-61	90.65	90.50	90.43	90.47	90.68	90.39	90.81	90.85	90.70	90.63	90.67	90.88	96.59	91.0			
1961-62	91.58	91.36	91.20	90.39	91.89	90.44	92.29	93.05	92.82	92.72	9 . 34	93.37	91.89	93.7			
1962-63	93.60	93.37	96.06	92.49	94.66	93.53	94.65	95.75	95.53	98.28	94.62	96.84	95.69	96.8			
1963-64	92.56	92.11	95.42	90.68	93.43	91.88	93.50	93.68	93.23	96.57	91.78	94.56	92.99	94.6			
1964-65	94.35	94.55	95.70	93.21	95.13	93.59	94.98	88.61	88.79	89.87	87.53	89.34	87.89	89.1			
1965-66	97.98	98.51	100.54	96.57	98.54	97.26	98.31	92.42	91.92	94.84	91.10	92.95	91.75	92.7			
1966-67	93.86	94.58	96.68	91.76	93.65	92.69	93.38	82.64	83.27	85.12	80.79	82.45	81.61	82.			
1967-63	105.04	105.65	108.17	100.55	103.30	102.44	102.71	90.42	90.94	93.11	86.55		88.18	88.			
968-69	106.88	107.16	111.81	101.69	105.80	104.49	105.26	95.92	96.17	100.35	91.26	94.95	93.77	94.			
969-70	103.28	103.62	108.22	36.65	101.51	99.71	98.49	92.32	92.63	96.74	86.39	90.74	89.13	88.			
970-71	101.31	101.57	107.58	94.03	99.80	97.64	103.30	92.14	92.39	97.85	85.53	90.77	88.81	93.			
.971-72	100.29	100.23	106.03	94.10	100.45	96.84	100.49	90.13	90.08	95.29	84.57		87.03	90.			
972-73	95.21	95.66	98.94	86.60	93.12	88.65	93.51	88.26	87.75	91.72	80.28		82.18	86.			
973-74	89.92	89.17	92.47	79.06	83.98	81.90	83.56	82.94	82.25	85.30	72.93		75.55	77.			
974-75	108.77	107.20	116.03	91.50	97.36	98.11	95.03	94.21	92.86	100.50	79.25		84.98	82.			
975-76	113.10	112.77	112.71	94.38	100.81	98.22	100.31	105.40	105.09	105.03	87.96		91.53	93.			
975-77	105.86	105.37	102.99	88.39	91.21	91.63	89.89	97.30	96.85	94.67	81.25		84.22	82.			
977-78	97.80	97.11	94.64	81.86	84.11	84.68	82.99	87.49	86.88	24.66	73.23		75.76	77.			
978-79 1979-80	108.38	105.64	105.94	88.44	93.86	89.72	93.58	100.25	97.72	87.60	81.81	86.82	82.99	86.			
1979-80 1980-81	123.00	118.47	121.80	96.50	109.55	98.61	109.63	112.89	TDR 75	111 00							
					106.30	96-11	100.08	110.25	105.49	108.67	88.05	97.35	87.06	107.			
					106.30	96-11	100.08	110.26	105.49	108.67	88.05	97.35	87.06				

Table-A 3.1 Continued

res 1,3,18 118		s paid are	weighed by Net re weighed by	Availability and		To the same	82.05			
YEARS	Value	Gross	Net	Pur	chase by Agric	culture				
	added	value of	avail- ability	Alt. 1	Alt. 2	Alt. 3	Alt. 4			
	106 63	104.90	107.40	102.58	104.64	104.32	104.3			
951-52	106.62		99.09	97.58	98.98	98.47	99.1			
952-53	100.19	98.21		117.01	115.73	115.95	116.0			
953-54	119.05	117.63	111 .78	118.78	116.29	118.12	115.6			
954-55	117.41	117.64	115.60	113.36	114.54	113.06	115.7			
955-56	112.81	112.22	113.13	105.00	106.77	106.51	106.4			
956-57	105.77	105.57	109.11	103.54	103.64	104.40	102.			
957-58	104.73	111.49	106.46	103.34	102.79	102.51	102.			
958-59	103.06	103.30	103.65		100.00	100.00	100.			
1959-60	100.00	100.00	100.00	100.00	91.13	90.84	91.			
960-61	91.10	90.95	90.88	90.93	95.18	93.67	95.			
1961-62	94.85	94.63	94.52	93.62	99.43	98.24	99.			
962-63	98.31	98.08	100.90	97.15	95.92	94.33	95.			
1963-64	95.03	94.57	97.96	93.10	90.95	89.48	90.			
1964-65	90.21	90.39	91.50	89.11	94.83	93.61	94.			
1965-66	94.29	94.80	96.76	92.94	80.70	79.87	80.			
1966-67	80.88	81.50	83.31	79.07		85.72	85.			
1967-68	87.89	88.40	90.51	84.14	86.44	90.92	91.			
1968-69	93.00	93.24	97.29	88.48	92.06	86.46	85.			
1969-70	89.56	89.85	93.84	83.81	88.02	86.15	91.			
1970-71	89.39	89.62	94.92	82.97	88.05	84.73	87.			
	87.74	87.70	92.77	82.33	87.88	80.33	84.			
1971-72	86.27	85.77	89.65	78.47	84.38	74.04	75.			
1972-73	81.29	80.61	83.60	71.48	75.92		78.			
1973-74	90.27	88.98	96.31	75.94	80.81	81.43	89			
1974-75	101.12	100.82	100.76	84.38	90.13	87.81	80			
1975-76	95.21	94.77	92.63	79.50	82.03	82.41	70			
1976-77	83.57	82.98	80.87	69.95	71.87	72.36	83			
1977-78	97.17		94.98	79.30	84.15	80.44	96			
1978-79	108.47		107.41	85.10	96.61	86.96				
1979-80	112.91		112.69	86.14	99.03	88.33	101			
1980-81			105.53	85.50	94.53	84.54	97			
1981-82	107.07		102.09	83.40	93.01	84.09	96			
1982-83 1983-84	107.26 106.29		99.74	82.60	90.76	83.29	93			

Table A-3.2

DOMESTIC TERMS OF TRADE FOR MANUFACTURING SECTOR

(Three Years Moving Average)

			are weigh			gand				es paid a	•			
YEARS	S - Pric	es recei	ved are w	eiched !	ру	200				es receiv			A	
	Value	Gross	Net			Agricul		Value	Gross	Net		ase by A		
	added	value of	avail- ability	Alt. 1	Alt. 2	Alt. 3	Alt. 4	added	value of	avail- ability	Alt. 1	Alt. 2	Alt. 3	Alt. 4
		output			-				output					
1951-54	103.54	102.21	101.28	101.13	101.77	101.58	101.87	106.27	104.60	103.73	103.47	104.14	103.95	104.24
1952-35	111.33	108.25	107.97			109.94			110.53	108.21	110.49	109.71	110.22	109.69
1953-56	111.98	111.40	109.10			111.27			114.02	111.70	114.57	113.70	113.90	113.79
1954-57	105,91	105.73	106.48			106.44			109.18	109.95	109.74	109.87	109.91	109.76
1955-58	101.09	101.08	102.78			101.30			104.73	106.49	104.29	105.28	104.96	105.70
1956-59	99.49	99.74	101.27	98.76	99.38				102.26	103.84	101.26	101.89	102.29	101.55
1957-60	99.57	99.88	100.31	99.09	99.15	99.29	98.91		101.40	101.83	100.58	100.64	102.29	100.40
1958-61	96.96	96.99	97.08	96.89	96.88		96.94	97.37		97.48	97.19	97.29	97.10	97.34
1959-62	94.08	93.95	93.90	93.72	94.19	93.61	94.37	94.63		94.45	94.17	94.75	94.16	94.93
1960-63	91.94	91.74	92.58	91.22	92.41	91.45		93.22		93.88	92.38			
1961-64	92.58	92.28	94.23	91.19	93.33	91.95						93.70	92.72	93.87
1962-65	93.50	93.34	95.73	92.13	94.41	93.00		94.16		95.86	22.75	94.92	93.53	95.08
1963-66	94.96	95.06	97.22	93.49	95.70	94.24		92.68		94.91	91.31	93.58	92.19	93.55
1964-67	95.40	95.88	97.64	93.49	95.77			91.57	91.65	93.76	90.14	92.28	90.88	92.18
1965-68	98.96	99.58	101.80					87.89		89.94	86.47	88.25	87.08	88.06
1966-69	101.92			96.29	98.50			88.49	89.04	91.02	86.15	88,12	87.18	87.79
		102.46	105.55		100.92		100.45	89.66		92.86	86.20	88.77	87.85	88.20
1967-70	105.07	105.48	109.40			102.21		92.89	93.25	96.73	88.07	91.54	90.36	90.30
1968-71	103.82	104.12	109.20			100.61		93.46		98.31	87.73	92.15	90.57	92.15
1969-72	101.63	101.81	107.28		100.58		100.76	91.53		96.63	85.50	90.59	88.32	90.77
1970-73	98.92	99.15	104.18	91.58	97.79	94.38		90.18	90.07	94.95	83.46	89.12	86.01	90.32
1971-74	95.14	95.02	99.15	86.59	92.52			87.11	86.69 87.62	90.77	79.23	84.68	81.59 80.90	84.70 82.03
1972-75 1973-76	97.97 103.93	97.34 103.05	102.48 107.07	85.72 88.31	91.49	89.55 92.74	90.70	88.47 94.18		92.51 96.94	80.05	85.24	84.02	84.29
1974-77	109.24	1.08.45	110.58	91.42	96.46	95.99	95.08	98.97	98.27	100.07	82.82	87.37	86.91	86.14
1975-78	105.59	105.08	103.45	88.21	92.04	91.51	91.06	96.73		94.79	80.81	84.34	83.84	84.45
1976-79	104.01	102.71	101.19	86.23	89.73	88.68	88.82	95.16		88.98	78.76	81.97	80:99	82.14
1977-80	109.73	107.07	107.47	88.93	95.84	90.98		100.21		94.69	81.21	87.54	83.09	88.14
1978-81	119.46	115.18	118.16		104.93		105.71	110.11		105.45	86.60	96.71	88.39	97.43
1979-82	123.81	118.69	122.74		109.37		111.19		108.77	112.48	88.68	100.22	89.75	101.89
1980-83	123.67	118.21	121.03		108.29		111.31	112.77	107.80	110.39	87.88	98.76	88.55	101.55
1981-84	120.77	115.29	115.79		104.84	94,90	108,10	110.25	105.25	105.69	86.48	95.71	86.63	98.67

Table-A 3.2 Continued

YEARS		e weighed by Net A		Purchase h	y Agricultur	e
ILAKO	Value Gross added value of output	availability	Alt. 1	Alt. 2	Alt. 3	Alt.4
	108.62 106.91	106.09	105.72	106.45	106.25	106.5
LJJL NA	112.22 111.16	108.82	111.12	110.33	110.85	110.3
1952-55		113.50	116.38	115.52	115.71	115.6
1953-56		112.61	112.38	112.53	111.56	112.4
1954-57		109.57	107.30	108.32	107.99	108.1
1955-58		106.41	103.75	104.40	104.47	104.0
1956-59	104.52 106.79	103.37	102.09	102.14	102.30	101.9
1957-60	102.60 104.93	98.18	97.88	97.97	97.78	98.0
1958-61	98.05 98.08	95.13	94.85	95.44	94.84	95.6
1959-62	95.32 95.19	95.43	93.50	95.25	94.25	95.4
1960-62	94.75 94.55	97.79	94.60	96.84	95.41	97.0
1961-64	96.06 95.76	96.79	93.12	95.43	94.02	95.4
1962-65	94.52 94.35	95.41	91.72	93.90	92.47	93.8
1963-66	93.18 93.25	90.52	87.04	88.83	87.65	88.
1964-67	88.46 88.89	90.19	85.38	87.32	86.40	87.0
1965-68	87.69 88.23	90.37	83.90	86.40	85.50	86.0
1966-69	87.26 87.71	93.88	85.48	88.84	87.70	87.6
1967-70	90.15 90.50	95.35	85.09	89.38	87.84	89.
1968-71	90.65 90.90		83.04	87.98	85.78	88.
1969-72	88.90 89.06	93.84	81.26	86.77	83.74	87.
1970-73	87.80 87.70	92.45	77.43	82.73	79.70	82.
1971-74	85.10 84.69	88.67	75.30	80.37	78.60	79.
1972-75	85.94 85.12	89.85	77.27	82.29	81.09	81.
1973-76	90.89 90.14	93.56	79.94	84.32	83.88	83.
1974-77	95.53 94.86	96.57	77.94	81.34	80.86	80.
1975-78	93.30 92.86	91.42	76.25	79.35	78.40	78.
1976-79	91.98 90.82	89.49	78.12	84.21	79.92	83.
1977-80	96.40 94.06	94.42	83.51	93.26	85.25	93.
1978-81	106.18 102.39	105.03	85.58	96.73	86.61	98.
1979-82	109.48 104.96	108.54		95.52	85.65	98.
1980-83	109.08 104.27	106.77	85.01 83.83	92.77	83.97	95.
1981-84	106.87 102.02	102.45	03.03			

Table-A 3.3 DOMESTIC TERMS OF TRADE FOR AGRICULTURE (ABSOLUTE)

	Gr	oss Valu	e of outp	out in Ag	ricultur	e Relati	ve to		Net Av	ailabilit	y in Agr	icultur	e Relati	ve to
fears	Value Added	GVO	Net Avail- ability	Alt.1	Alt.2	Alt.3	Alt.4	Value Added	GVO in Manuf- actur- ing	Net Avail- ability	Alt.1	Alt.Z	Alt.3	Alt.4
1951-52	100.00	101.75	99.39	104.07	102.02	102.33	102.28	G2 70	95.32	93.11	97.49	95.57	95.86	95.82
1952-53	99.51	101.62	100.72	102.28	100.83	101.35	100.62	93.79 99.81	101.82	100.92	102.48	101.03		100.82
1953-54	84.35	85.37	89.84	85.82	35.78	86.61	86.54	84.00	85.01	89.46	85.46	86.41	86.24	84.13
1954-55	85.33	86.16	87.68	85.33	87.16	85.81	87.53	85.17	85.01	86.50	84.19	85.99	84.66	86.45
1955-56	91.36	91.84	91.11	90.92	89.98	91.16	89.52	88,64	89.11	88.40	88.21	87.30	85,45	86.80
1956-57	97.28	97.47	94.30	97.99	96.37	96.61	96.67	94.54	94.73	91.65	95.23	93.66	93.89	
1957-58	98.06	97.36			99.08	98.37	99.82	95.49	94.81	93.93	96.58	96.48		93.95
1958-59	98.76		96.46	99.18				97.03	96.80	95.48	97.35	97.28		97.21
		98.54	98.21	99.10	99.02	60.50	98.99							97.25
1959-60	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00		100.00
1960-61	110.07	119.25	110.34	110.28	110.04	110.38	109.88	109.77	109.94	110.03	109.98	109.73		109.58
	107.47	107.73	107.85	108.89	107.10	108.82	106.65	105.42	105.68	105.80	106.82	105.07		104.62
1962-63	104.43	104.68	101.75	105.68	103.26	104.51	103.27	101.72	101.96	99.11	102.34	100.58		102.59
1963-64	106.75	107.26	103.55	108.96	105.76	107.54	105.67	105.23	105.74	102.08	107.41			104.17
1954-65	112.86	112.63	111.27	114.25	111.93	113.78	112.12	110.85	110.63	109.29	112.22	109.95		110.13
1965-66	108.20	107.62	105.55	109.78	107.59	108.99	107.84	106.05	105.48	103.35	107.60		106.83	100.70
1965-67	121.01	120.08	117.48	123.78	121.28	122.53	121.62	123.55	122.61	119.95	126.38			124.15
1967-68	110.60	109.96	107.41	115.53	112.46	113.41	113.11	113.76	113.12	110.49	118.85	115.59		116.35
1968-69	104.26	103.98	99.65	109.57	105.32	106.64	105.86	107.53	107.25	102.78	113.02	108.53		109.18
1369-70	108.32	107.96	103.37	115.75	110.20	112.20	113.59	111.66	111.29	106.56	119.32		115.66	117.09
1970-71	108.52	108.24	102.20	116.92	110.17	112.60	106.43	111.87	.111.58	105.35	120.53			109.71
1971-72	110.95	111.01	104.94	118.24	110.77	114.90	110.72	113.97	114.03	107.79	126.46		118.03	113.73
1972-73	113.29	113.95	109.02	124.56	115.83	121.68	115.35	115.91	116.59	111.54 119.62	127.44		124.49 135.06	118.02
1973-74	120.56	121.58	117.23	137.11	129.09	132.37	129.74	123.02 110.77	124.05	103.84	131.63	123.75		126.79
1974-75	106.14	107.69	99.50	126.17	118.58	117.58	121.49	98.90	99.19	99.24	118.51		113.28	111.50
1975-75	94.88	95.15	95.21	113.69	106.44	109.25 118.73	106.97 121.03	105.03	105.52	107.95	125.79	121.90		123.69
1975-77 1977-78	102.77 114.30	103.25	105.63 118.11	123.08	119.28	131.99		119.63		103.66		129.14	136.19	111.00
1972-70	59.33	102.33	102.05	136.55 122,24	115.18	120.50	134.69	102.91	120.51	105.28	132.96 126.11			
1979-80	83.58	91.96	89,45	112.90	99.45	90.51	99.38	92.19	95.72	93.03	117.50		114,99	103.44
1380-81	85,33	89.23	35.50	111.85	97.30	109.08	95.12	88.56	92.60	88.74	116.09	100,99	113.21	25.72
1651-85	90.69	94.80	92.02	113.57	102.72	114.86	100.09	93.43 93.23	97.62	94.76	110.95		314.01	100.01
7 70 7 30 7	95.19	89.22	89.49	109.55	100.84	116.42	103.49	04.04		1.000-000				

10.52 97.02 97.02 98.55 98.55 98.55 98.77 100.00 110.31 109.19 106.84 106.84 106.55 96.77 106.56 96.87 98.71 99.71 106.03 111.21 99.71 106.03 111.21 91.94 88.42 94.47 95.74 81.30 75.74 81.30 81.30 81.30 81.30	value acted
112.33 98.98 85.79 88.37 95.71 100.45 100.81 99.54 100.81 109.46 107.10 109.46 107.10 105.77 101.51 105.77 101.51 105.77 101.51 105.73 94.55 93.32 96.50 98.45 99.77 107.10 107.10 107.10 109.46 100.49 100.56 100.29 100.56 100.29 100.56 100.29 100.56 100.29 100.56 100.29 100.56 100.29 100.56 100.29 100.56	0.0
92.17 90.27 89 93 94 94 97.18 99 21 100 00 110 58 104 10 104 10 104 80 104 80 104 80 104 80 103 44 92.45 89 43 92.40 92.45 88 73 97 09 105 66 94.39 97 09 105 66 98 76.50 83.56	Net Avail- ability
114.88 99.62 86.24 87.52 94.74 100.98 102.70 100.00 110.64 108.12 110.27 107.29 103.55 108.98 99.45 98.34 106.34 106.34 106.34 106.27 115.48 126.48 126.48 109.29 103.62 103.53 113.13 113.13 113.13 113.13 113.13	Alt. 1
112:52 98:21 87.19 89:40 99:31 102:60 100:03 10	
98.72 87.02 88.01 94.99 99.56 101.86 100.31 100.00 108.45 110.57 106.92 108.84 107.89 97.62	(+
98.01 86.95 89.88 93.29 99.62 103.37 100.00 110.12 106.95 106.95 106.95 107.09 97.36 95.00 101.72 107.09 97.36 95.00 101.72 106.95 106.95 106.95 107.09 97.36 96.80 99.69 111.25 112.49 106.35 91.22 87.77 96.89 97.88	

STOCKERS OF STREET OF A SHORE SHORE STOCKERS OF STREET STR

.

7088-75 178-45 1071-10 700-80 1071-10 30-03 1004-30 30-03 7004-25 30-80 10029-20 40-30 10029-20 40-30 10029-20 40-30

Table-A 3.3 Continued

77

Alt.

4

10

Table-A 3.4

DOMESTIC TERMS OF TRADE FOR AGRICULTURE (3 YEARS MOVING AVERAGE)

		s Value of	Output in A						ailabilit						
EATS	value	GVO	Net		e by Acrie			Value	GVC in	Net		ase by A	oricit.	.re	
	Asced		Avail- ability	Alt.l	Alt.2	Alt.3	hlt.4	Added	Manufa- cturine	Avail- ability	Alt.1	Alt.2	Alti	Ait.	
1951-54	94.65	96.25	96.65	97.39	96.64	96.76	96.48	92.53	94.05	94.50	95.19	94.34	94.55	93.5	
LEE2-55	90.10	91.05	92.74	91.14	. 91.59	91.26	91.60	89.56	90.61	92.29	90.71	91.14	90.82	90.	
1953-56	67.35	87.79	39.54	87.36	87.97	87.86	87.90	85.94	86.38	66.12	85.95	86.57	Bc.45	85.	
1954-57	91.66	91.82	91.03	91.14	91.17	91.19	91.27	89.45	89.62	88.95	89.21	88.98	89.00	89.	
955-58	95.57	95.56	93.96	96.03	95.14	95.38	95.34	92.89	92.88	91.33	93.34	92.48	92.71	92.	
1956-59	98.03	97.79	96.32	98.76	98.16	98.09	98.49	95.69	95.45	94.02	96.39	95.81	95.74	96.	
1957-60	98.94	98.63	98.22	99.43	99.37	99.22	99.60	97.51	97.20	96.80	97.98	97.92	97.75	98.	
952-61	102.94	102.93	102.85	103.13	103.11	103.22	102.96	102.27	102.25	102.17	102.44	102.34	162.54		
959-62	105.85	105.99	106.06	106.39	105.71	106.40	105.51	105.06	105.21	105.24	105.60	104.93	105.61		
965-63	107.32	107.55	106.65	108.28	106.80	107.90	106.50	105.64	105.86	104.98	106.58	105.13	10€.21		
951-64	106.22	106.56	104.38	107.84	105.37	106,96	105.20	104.12	104.46	102.33	105.72	103.3	104.85		
9-2-65	108.01	108.19	105.52	109.63	106.98	108.61	107.02	105.93	106.11	103.49	107.52	104.93	10e.52		
9=3-66	109.27	109.17	106.79	111.00	108.43	110.10	108.54	107.38	107.28	104.91	109.08	106.55	10e 2		
964-67	114.02	113.44	111.43	115.94	113.60	115.10	113.86	113.49	112.91	110,86	115.40	113.08	114.57		
965-68	113.27	112.55	110.15	116.36	113.78	114.98	114.19	114.46	113.74	111.26	117.61	114.99	116.20		
365-69	111.36	111.34	108.18	116.29	113.02	114.19	113.53	114.95	114.33	111.07	119.42	116.05	117.26		
967-70	107.73	107.30	103.48	113.62	109.33	110.75	110.85	110.96	105.55	106.61 .	117.06	112.64	114.12		
968-71	107.03	106.73	101.74	114.08	108.56	110.48	108.63	110.35	110.04	104.90	117.62	111.93	113.91		
969-72	109.26	109.07	103.50	116.97	110.38	213.23	110.25	112.50	112.30	10€.57	122.10	113.65	116.59		
975-73	110.92	111.07	105.39	119.91	112.26	116.39	110.83	113.92	114.07	108.23	124.61	115.29	119.53		
972-74	114.93	115.51	110.40	126.64	118.56	122.98	118.60	117.63	118.22	112.98	131.27	121.34	125.86		
972-75	113.33	114.41	108.58	129.28	121.17	123.91	122.19	116.57	117.68	111.67	133.01	124.66	127.45		
573-76	107.19	108.14	103.98	125.66	118.04	119.77	119.40	110.90	111.86	107.57	130.03	122.14	123.92		
· 4-77	101,26	102.03	100.11	120.98	114.77	115.22	116.50	104.90	105.70	103.68	125.33	115.67	119.34		
975-7B	104.02	104.50	106.32	124.44	119.54	119.99	120.90	107.86	108.41	110.28	129.09	124.60	124.47		
\$75-79	105.63	106.89	108.60	127.29	119.12	123.74	123.75	109.20	110.53	112.30	131.62	126.62	121.98		
977-80	100.90	103.13	103.20	123.90	115.84	114.33	116.53	104.92	107.27	107.34	126.86	120.43	125.83		
976-81	91.25	94.51	92.33	115.66	103.98	106.70	103.34	94.55	97.96	95.70	119.90	107.78	117.50		
979-82	88.20	92.00	88.99	112,77	99.82	104.82	98.20	91.38	95.31	92.20	116.65	103,43	115.50		
9-3-83	88.74	91.08	89.00	111.66	99.41	110.86	96.72	91.73	95.96	93.82	117.65	104.76	116.80		
982-84	90.70	93.27	92.91	113.50	102.59	113.31	99.51	93.57	98.78	97.66	119.30		119.03		

Continued....

106.50 101.27 106.95 108.02 101.15 105.32 105.36 08.78 00.44 93.36 102,73 98.44 04.86 95.19 96.36 98.41 91.61 106.65 100.27 107.50 108.98 103,34 100.63 105.63 104.34 108.73 100.12 103.69 105.85 101.29 94.82 96.09 98.24 97,90 98.03 95.51 GVO Net availability 106.72 104.46 106.16 108.09 103.26 102.46 102.92 101.38 98, 45 99.70 96.32 93.22 91.59 91.43 95.11 94.02 97.16 94.35 98.67 99.05 94.05 85.13 81.52 90.67 Alt. 1 103.73 109.46 113.91 116.08 103.32 103.30 112.95 116.12 113.75 117.08 109.36 105.36 102.72 100.42 102.26 103.99 106.61 107.04 108.56 109,68 109.76 107.06 103.55 100.94 101.26 94,41 89.50 91.13 105.57 99.47 00.25 PURCHASE BY AGRICULTURE Alt.2 107,99 106.11 103,44 100,88 100,64 101.69 104,46 104.55 105.93 106,91 109.72 105.57 111.69 109.23 103.84 106.98 108.67 102.38 98.56 94.16 90.12 91.60 99.37 99.42 97.74 96.61 91.45 Alt.3 108,85 106.24 102.82 100.72 100.58 100.41 105.85 106, 16 108.78 102.98 102.77 107.53 105.38 102.12 112,28 112,72 106.16 101.99 90.01 104.51 110.31 112.89 109,68 104.29 108.61 98.80 94.19 91.25 97.87 99.57 99,47 Alt.4 103.37 100.99 105.96 106.99 106.16 101.13 104.70 104.65 108.04 112.86 198.20 110.61 102.06 106,19 110.48 108.71 101.08 94.26 90.04 105.39 97.78 97.97 99.82 99, 29 99.28 95.28 89,96

ABLE-A 3.4 (Continued)

WARKETING IN AGRICULTURAL SECTOR RELATIVE TO:

### INCOME TERMS OF TRADE FOR AGRICULTURE ( 3 YEARS MOVING AVERAGE )

EARS	Relativ	e to Mar	dices Wei nufacturi	ing Price	e Weigh	ting ted by					ghted by GVO Relative to ices Weighted by					
	Value added	Gross value	Net avail-		ase by			Value added	Gross value	Net avail-	Pu	rcha	se by	Agricu	ltu	re
0 0		of output	ability	Alt.1	Alt.2	Alt.3	Alt. 4		oī output	авілісу	Alt.	1 A	lt. 2	Alt.	3	Alt. 4
951-54	81.93	83.30	83.30	84.26	83.60	83.82	83.53	80.18	81.48	82.04	82.4	0	81.77	81.	94	81.06
952-55	80.5	81.33	82.95	81.41	81.86	31.55	81.87	80.10	80.93	82.54	81.0	01	81.45	81.		80.80
953-56	78.8	79.21	80.87	78.84	79.42	79.30	79.36	77.56	77.97	79.62	77.6		78.19	78.		77.47
954-57	83.22	83.36	82.60	83.02	82.78	82.79	82.89	81.21	81.36	80.62	81.0		80.79	80.		80.90
955-58	87.72	87.70	86.20	88.17	87.34	87.54	87.54	85.26		83.78-	85.7		84.91	85.		85.10
956-59	92.14	91.92	90.52	92.82	92.25	92.18	92.56	89.93		88.35	90.5		90.04	89.		90.34
957-60	94.34	94.06	93.68	94.80	94.74	94.60	94.96	93.01		92.36	93.4		93.40	93.		93.61
958-61	102.47	102.47	102.40	102.65	102.44	102.74	102.48		101.82	101.75			01.90	102.		101.83
959-62	111.98	112.14	112.22	112.58			111.61		111.27	111.34			10.97	111.		110.74
960-63	117.29	117.54	116.53	118.35	116.60	117.93	116.48		115.66	114.68			14.85	116.		115.37
961-64	121.46	121.85	119.33	123.34	120.49	122.31	120.29		118.81	117.00	120.9			119.	-	118.68
962-65	127.61	127.81	124.69	129.53	126.40	128.33	126.45		125.38	122.31	127.0			125.		124.78
963-66	134.78	134.61	131.72	136.90	133.74	136.14	133.89		132.31	129.39	134.5			1.33.		131.57
964-67	146.53	145.77	143.00	149.02	146.01	148.26	146.35		145.18	142.54			45.44	147.		145.78
965-68	158.14	157.14	153.59	162.63	158.98	160.96	159.57		159.03	155.55			60.91	162.		161.53
966-69	170.72	169.81	1.64.71	177.52	172.41	174.21	173.20		174.43	169.35	182.3			178.		177.91
967-70	183.42	182.71	176.10	193.58	186.16	188.69	188,90		188.26	181.44			91.81	194.		194.63
968-71	188.48	187.94	179.17	200.94	191.44	194.59	191.44	194.39	193.77	184.73			97.13	200.		197.03
969-72		197.43		211.70			199.70	203.63	203.27	192.92			05.71	211.	03	205.60
970-73		201.91		213.00			201.56		207.34	196.75			09.54	217.		206.96
971~74		218.56		239.76			224.59		223.67	213.89			29.73	238.		244.33
972-75		217.97		245.66			232.78		224.17	212.77			37.45	242.		230.4
973-76 9 <b>74-</b> 77		209.68		243.01 236.72			231.45 228.58	215.02	207.25	208.71 203.60			36.78 33.22	243.		239.43
975~78		213.17		253.83			246.69		221.15	211.09			46.05	257.		255.89
976-79		224:75		267.65			260.06		232.42	222.12			59.20	269.		268.93
077-80		228.15		274.31			257.34		237.28	223.23			59.18	278.		267.6
:73-81		222.02		272.01	244.03	255.41	242.45		230.18	224.67	282.0	2 2	53.03	276.		251.3
179-82		228.22		279.69			243.53		236.42	228.67	289.7			286.		252.30
980-83		235.25	230.03	288.34	256.81 277.55		249.81		248.05 265.04	242.67 264.26	304.0		70.85	302.		265.40

Table-A 3.5 Continued

	th extracted parties of	AGRI CULTU	RE PRICE INDIC	ES WEIGHTED	BY VALUE A	DDED RELATI	VE
		TO MANUFA	CTURING PRICE	INDICES WELG		BY AGRICULT	
YEARS	Value added	Gross value of output	Net avail- ability	Alt. 1	Alt. 2	Alt. 3	Alt. 4
	the second secon		86.04	86.60	85,89	87.07	85.83
1951-54	84.18	85.58	83.03	81.46	81.93	81.60	81.94
1952-55	80.54	81.39	82.78	80.72	81.31	81.19	81.24
1953-56	80.69	81.11		85.73	85.48	85.49	85.58
1954-57	85.93	86.09	85.29	91.31	90.46	90.66	90.66
1955-58	90.84	90.83	89.27	95.18	94.59	94.51	94.92
1956-59	94.48	94.25	92.81	96.20	96.15	96.00	96.37
1957-60	95.74	95.44	95.05		102.94	102.38	102.87
1958-61	102.87	102.86	102.79	103.05	112.27	112.57	112.33
	112.70	112.87	112.94	113.32	118.02	118.79	118.08
1959-62	118.90	119.16	118.13	119.99	122.24	124.37	122.32
1960-63	123.51	124.34	121.34	125.42	124.99	126.90	125.04
1961-64	126.19	126.84	123.27	128.10	124.99	130.86	129.01
1962-65	129.87	130.18	126.86	131.93	128.87	135.87	134.42
1963-66	134.60	133.91	131.49	136.86	134.10	143.16	142.20
1964-67	140.97	140.08	137.03	144.92	141.67	153.23	152.34
1965-68	150.16	149.37	145.01	156.15	151,64		167.10
1966-69		161.62	155.75	171.26	164.67	166.86	172,29
1967-70	162.24	169.19	161.27	180.89	172.12	175.17	179.7
1968-71	169.67	177.79	168.72	190.65	179.92	184.57	183.80
1969-72	178.10		175.17	198.86	186.13	193.07	
1970-73	183.89	184.16	191.91	219.82	205.82	213.45	205.9
1971-74	199.30	200.33	188.10	223.12	209.11	213.99	210.7
1972-75	195.78	197.61	183.22	221.03	207.60	210.84	209.9
1973-76	188.61	190.23	178.27	214.95	203.99	204.74	207.0
197477	179.88	181.17		231.93	222.79	223.64	219.7
1975-78	193.80	194.77	198.16	244.18	234.79	237.48	231.6
1976-79	202.50	205.05	208.21	250.18	233.32	244.41	229.0
1977-80	203.05	208.06	207.91	250.74	225.02	245.69	223.5
1978-81	197.47	204.68	199.79		226.84	253.34	223.0
	200.40	209.03	202.19	256.21	238.72	266.13	232.1
1979-82	209.00	218.68	213.86	271.54	258,23		250.4
1980-83 1981-84	223.94	234.39	223,93	289.19	230,23	207.00	

-59-

Table-A 3.6

the name of the state of the st

# SINGLE FACTORAL TERMS OF TRADE FOR AGRICULTURE (Three Years Moving Average)

1981-84

YEARS	Relati	Price In ve to Ma					505 119 508 188	1979-81		Price Ind				
LAKO	Value added	Gross value of	Net avail- ability		Amrops	Agricult		Value added	Gross value of	Net avail- ability	13.05	SCOTE P	of Agric	Star V
		output		Alt. 1	Alt.2	Alt. 3	Alt. 4	[ Janks of ]	output		Alt. 1	Alt. 2	Alt. 3	Alt.4
1951-54	107.07	108.87	109.32	110.17	109,21	109,46	109.13	104.67	106.38	106.88	107.62	106.70	106.95	105.86
1952-55	107.39	101.48	103.37		102.07	101.72	102.07	99.92	101.00	102.87	101.12	101.58	101.23	100.81
1953-56	95.83	96.33	98.3	95.87		96.43	96.46	94.30	94.79	96.75	94.34	95.02	94.89	94.17
1954-57	96.33	96.51	95.74	96.05	95.83	95.85	95.94	. 94,03	94.20	93.46	93.75	93.72	93.55	93.65
1955-58	97.29	97.30	95.69	97.74	96.83	97.10	97.00	94.56	94.57	93.12	94.99	94.28	94.37	94.28
1956-59	97.07	96.82	95.37	97.77	97.18	97.11	97.51	94.73	94.50	93.20	95.43	95.02	94.79	95.17
1957-60	98.15	97.84	97.44	98.62	98.56	98.42	98.79	96.73	96.43	96.16			45	
1958-61	99.32	99.29	99.21	99.49	99.39	99.58	99.33	98.65	98.62		97.19	97.14 98.72	97.00	97.36
1959-62	101.15	101.29	101.35	101.67	A STATE OF THE PARTY OF	101.68	100.83	100.40	100.54	98.55 100.60	98.82		98.91	97.36
1960-63	102.72	102.94	102.03	103.66		103.27	102.01	101.06	101.28		100.92	100.27	100.92	100.09
1961-64	104.88	105.22	103.05	100.49		105.60	103.87	102.81	103.15	100.40	101.99	100.56	101.61	101.04
1962-65	108.61	108.79	106.12	110.23		109.21	107.62	106.52	105.15	101.02	104.40	101.90	103.52	102.50
1963-66	109.44	109.34	115.97	111.17		110.28	108.72	107.55	107.45	104.08	108.12	105.51	107.11	106.22
1964-67	115.89	115.30	113.26	117.84	CONTRACTOR OF THE PARTY.	116.28	115.73	115.38	114.79	105.08	109.25	106.72	108.38	106.84
1965-68	121.01	120.24	117.66	124.42		122.9	122.09	- 14 34 m - W		112.71	117.33	114.97	116.49	115.24
1966-69	126.84	126.15	122.53	131.85		129.42	128.67	122.42	121.65	119.00	125.91	123.07	124.36	123.54
1967-70	136.21	135.68	130.78	143.77		140.10	140.32	1,30.27	129.57	125.85	135.43	131.55	132.93	132.16
1968-71	139.82	139.41	132.9	149.09		144.37		1,140.33	139.79	134.74	148.14	142.45	144.35	144.58
1969-72	146.93	146.56	139.21	157.28		152.26	142.04	144.15	143.74	137.02	153.72	146.24	148.84	146.44
1970-73	145.23	145.48	138.03	157.05			148.41	7,151.28	151.01	143.34	164.14	152.86	156.78	152.81
1971-74	151.52	152.30	145.54	166.96		152.45 162.14	145.17	1 149.20	149.4	141.75	163.48	150.99	156.55	149.08
1972-75	146.76	148.16	140.68	167.29		160.43	156.38 158.08	155.08	155.86	148.95	173.06	159.98	165.93	160.03
1973-76	137.51	138.95		161.36		153.86	153.30	1,150,91	1.52.33	144.63	172.07	161.26	164.97	162.60
1974-77	128.45	129.64	127.29	153.72		146.44	148.04	142.45 133.27	143.70	138.23 131.81	166.94 159.05	156.81 153.80	161.28 153.80	158.60
1975-78	133.70	134.59	136.92	160.26		154.52	155.70	138.90	134.27	133.43	161.94	155.39	162.43	153.31
1976-79	135.69	137.66	139.86	163.93		149.10	150.77	136.05	139.61	130.81	161.52	152.23	160.79	
1977-80	130.72		135.24	162.09		142.06	142.23	130.05	136.14	126.85	161.03	147.02	161.06	
1978-81	120.16	126.07	123.12	153.96		134.54	127.33	121.54	126.49	122.42	156.36	the second of the second	152.84	
1979-82	119.50	125.87	121.76	153.87		146.14	132.48	125.03		126.15	159.85		158.02	
1980-83	120.19	123.42	120.58	151.28		150.22	132.46	123.03		120.15	159.85		158.02	
1981-84	121.14	124.58		151.56		151.34	132.84		130.91	130 37	159.55	141.00	1,70,21	150.172

Table-A 3.6 Continued

		AGRICULTURAL	MANUFACTURI	ES WEIGHTED BY	CES WEIGHTED	BY	
	Value	Gross	Net	Bar Islamate to	PURCHASE	BY AGRICULT	URE
	added	value of output	avail- ability	Alt. 1	Alt. 2	Alt. 3	Alt. 4
1951-54	110.22	112.03	112.40	113.41	112.38	112.63	112.32
1952-55	100.37	101.43	103.35	101.54	102.05	101.67	102.0
1953-56	98.17	98.68	100.65	98.19	98.88	98.75	98.7
1954-57	99.50	99.68	98.89	99.21	98.98	99.00	99.0
1955-58	100.80	100.80	100.14	101.25	100.32	100.60	100.5
1956-59	99.51	99.26	97.77	100.24	99.63	99.57	99.9
1957-60	99.61	99.30	98.89	100.10	100.05	99.90	100.2
1958-61	99.71	99.69	99.61	99.89	99.79	99.33	99.7
1959-62	96.03	101.91	101.98	102.31	101.40	101.66	101.4
1960-63	98.41	104.37	103.44	105.11	103.37	104.06	103.4
1961-64	100.92	107.36	104.80	108.30	105.57	107.40	105.6
1962-65	107.50	108.04	105.01	109.12	106.48	108.09	106.5
1963-66	105.50	105.77	103.07	107.18	104.69	136.31	104.8
1964-67	106.52	105.98	104.08	108.30	106.12	107.53	106.3
1965-68	107.93	107.24	104.91	110.93	108.45	109.59	108.8
1966-69	111.52	110.92	107.72	115.94	112.62	113.79	113.1
1967-70	120.45	119.99	115.64	127.18	122.27	123.91	124.1
1968-71	125.86	125.50	119.63	134.22	127.69	129.96	127.8
1969-72	132,31	132.07	125.35	141.64	133.69	137.11	133.6
1970-73	132.48	132.66	126.16	143.24	134.09	139.04	132.4
1971-74	138.84	139.52	133.66	153.04	143.29	148.62	143.
1972-75	133.17	134.41	127.97	151.68	142.14	145.52	143.
1973-76	124.99	126.07	121.38	146.41	137.52	139.66	139
1974-77	114 (0						

# OMESTIC\_TERMS OF TRADE FOR AGRICULTURE: ALTERNATE METHOD (Three Years Moving Average)

	1									(Usi	ng Mark	etings	as We	ight fo	r Agri	cultur	e Price	es)
	M OJ	cultura anufact hted <u>v</u>	uring F		dices		Matic	cultura facturi VO of	l Price	Indice ce Indic	s Relat ces Weig	ive to hted	Agric Manuf	acturing by N	Price g Price et Ava	Indice Indice Labili	es Weig	hted
YEARS	Consumption	Intermediate Goods	Investment Goods	Consumption & Intermediate Goods	Consumption & Investment Goods	Intermediate & Investment Goods	Consumption	Intermediate	Investment Goods	Consumption & Intermediate Goods	Consumption & Investment Goods	Intermediate % Investment Goods	Consumption	Intermediate Goods	Investment Goods	Consumption & Intermediate Goods	Consumption & & Investment Goods	Intermediate & Investment Goods
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)
1951-54 1952-56 1953-56 1955-58 1955-59 1957-60 1958-61 1959-62 1960-63 1961-64 1962-65 1963-66 1964-67 1965-68 1965-68 1968-71 1969-72 1970-73 1971-74 1972-75 1973-76	87.51 87.56 94.86 100.21 101.52 100.67 103.18 106.15 109.13 109.66 110.72 110.96 112.61 110.26 107.60 111.08 114.47 119.09 124.45 123.31	110.28	94.14 93.79 93.25 95.68 98.23 99.94 103.99 100.15 99.09 98.61 92.13 86.17 79.12 76.82 77.18 77.65 78.69 71.89 66.98 63.60 69.76 67.86	89.99 89.28 95.66 99.68 101.10 100.59 103.24 106.46 109.59 108.71 106.95 106.42 203.74 102.00 100.27 102.93 105.40 108.51 114.24 114.76 109.86 103.25 104.58	88.39 88.45 94.42 99.14 100.71 100.52 103.44 106.30 108.53 107.07 105.75 105.50 102.01 99.24 96.09 97.52 99.83	96.63 95.12 96.39 98.59 99.99 103.57 106.76 107.50 104.19 102.43 100.69 99.87 94.33 89.31 83.65 81.99 82.30 82.97 86.41 82.70 78.95 75.30 81.18 80.53	88.26 87.69 94.60 99.63 100.62 99.86 102.85 106.23 109.22 109.80 110.62 109.30	106.03 101.44 100.45 101.99 101.98 104.55 108.79 112.42 110.90 108.35 104.20 100.85 95.56 91.47 87.74 88.79 92.23	95.18 94.95 93.81 95.82 98.14 100.00 104.43 106.53 102.30 100.13 98.90 98.18 91.40 85.44 78.39 76.20 76.66 77.67 67.66 64.39 70.91 69.49	90.87 89.55 95.61 99.68 100.83 100.17 102.97 106.39 109.45 109.88 108.85 106.83 104.75 101.93 100.23 99.59 102.25 105.08 108.34 104.75 105.08 108.34 104.75	89.14 88.67 94.30 98.78 100.04 99.89 103.26 106.45 108.79 108.30 107.06 105.32 104.54 100.97 98.38 95.55 97.15 99.56 102.10 106.10	100.62 97.71 96.49 95.67 97.46 99.57 100.73 107.15 108.03 104.84 102.97 101.09 99.10 93.16 87.91 82.86 81.64 82.79 86.75 83.13 79.28	89.56 95.40 100.80 102.06 100.93 103.35 106.63 109.19 109.24 108.40 106.90 106.21 103.65 102.54 101.65 104.66 107.29 110.51 116.24 116.96 113.87 109.04 112.49 115.03	101.39 97.05 95.26 -5.48 96.98 98.68 102.25 107.60 110.33 108.19 106.71 104.30 103.47 99.23 95.74 92.61 93.42 96.13 103.39 105.84	92.76 94.48 92.67 93.92 95.50 98.62 103.72 106.38 105.57 101.26 98.30 97.03 96.90 97.40 85.91 77.34 78.52 84.72 78.34 73.03 69.48 77.16 77.33	91.49 95.69 99.24 100.62 100.39 103.14 106.93 107.50 107.50 107.50 107.50 107.50 107.50 107.50 107.50 107.50 108.91 109.91 110.99 108.81	91.40 91.04 93.99 97.78 99.16 103.60 106.69 107.81 102.96 102.96 102.96 102.96 102.96 103.60 104.22 102.96 104.22 102.96 95.34	95.54 95.54 95.15 92.87 93.67 98.68 103.16 106.63 106.63 106.63 106.63 106.63 100.10 100.

Continued Table -A 3.7

		A L T Iltural Pr Indices W		y Alterna	tive - 1	ufacturing of	Agricul Price I	tural Pri	ce Indices	Alternati	to Manuf	acturing
YEARS	Consump- tion Goods	Inter- mediate Goods	Invest- ment Goods	Consump- tion and Intermed- iate Goods	Consump- tion and Invest- ment Goods	Intermed- iate and Investment Goods	Consump- tion Goods	Intermed- iate Goods	Invest- ment Goods	Consump- tion and Intermed- iate Goods	Consump- tion and Invest- ment Goods	Intermed- iate and Investment
	19	20	21	22	23	24	25	26	27	28	29	30
1951-54	100.61	97.35	90.68	100.64	99.41	95.72	100.30	99.75	90.85	100.24	99.29	95.46
1952-55	90.95	96,13	92.68	92.03	90.12	94.49	90.78	95.62	92.93	92.36	90.95	94.49
1953-56	89.68	91.29	94.44	90.18	89.04	92.05	89.70	92.73	94.65	90.87	90.27	93.16
1954-57	95.74	88.73	92,56	95.49	95.55	90.11	96.08	92.12	92.74	95.76	95,33	92.06
1955-58	101.35	89.21	93.74	99.75	100.79	91.18	101.75	93.83	93.90	99.76	100.27	93.53
1956-59	102.86	93.55	95.34	101.67	102.35	94.89	104.63	97.29	95.45	101.04	101.97	96.61
1957-60	101.61	97.38	98.55	101.18	101.48	98.14	103.29	99.46	98.61	100.85	101.51	99.11
1958-61	103.68	102.59	103.72	103,67	103.85	102.82	105.23	102,21	104.06	103.05	104.04	102.68
1959-62	106.68	110.17	106.38	107.14	106.85	107.79	106.98	105.13	106.24	106.27	106.94	105.52
1960-53	109.21	116.36	105.55	110.09	109.29	110.60	109.66	106.66	105.29	109.19	108.98	106.23
1961-64	109.27	117.62	101.23	110.24	108.84	109.09	109.82	104.41	100.19	109.14	108.32	103.28
1962-65	102.22	120.11	97.26	109.15	108.28	107.81	109.14	104.34	98.00	108.18	107.12	101.70
1963-66	97.56	121.95	95.06	107.50	107.34	106.65	107.94	103.76	96.49	105.99	105.90	100.35
1964-67	93.23	126,62	94.01	106.76	106.99	107,82	107.51	105.71	96.11	105.14	105.48	101.06
1965-68	93.46	123.50	88.68	104.31	103.70	104.35	105.12	103.44	90.57	102.84	102.40	97.70
1966-69	92.33	119.16	83.40	102.83	101.55	100.18	103.99	100.50	85.11	101,13	100.23	93.88
1967-70	91.40	113.98	77.11	101.50	100,64	94.89	103.13	96.26	78.63	99.16	98.01	88.84
1968-71	94.18	115.25	75.09	104.19	103.08	94.37	106.26	95.85	76.54	101.03	99.78	87.66
1969-72	96.21	120.56	76.22	106.92	105.72	96.70	109.05	96.07	77.67	102.79	102.07	88.09
1970-73	98.53	137.43	78.85	110.05	107.58	103.76	112.10	99.30	80.34	105.06	104.48	90.59
1971-74	102.77	159.77	82.22	116.98	112.39	115,31	117.52	109.52	83.81	111.93	109.07	97.98
1972-75	102.94	172.92	75.01	118.79	119.91	117.87	117.96	119.59	77.48	115.36	107.21	99.70
1973-76	100.33	172.52	70.82	116.85	108.93	115.97	114.46	120.55	72.22	114.09	103.86	98.45
1974-77	96.15	176.58	67.36	112.37	103.80	115.24	108.92	124.77	68.71	111.15	99.23	98.28
1975-78	98.97	192.17	74.78	116.17	108.78	126.39	111.54	134.25	76.33	115.33	103.73	107.45
1976-79	100.95	199.15	74.93	118.80	109.96	128.97	113.65	140.11	75.56	118.39	105.09	110.07
1977-80	99.70	182,84	70.26	115.94	108.34	118.50	111.99	123.84	71,86	112.65	102.85	99.64
1978-81	96.38	155.28	62.35	110.06	101.54	101.43	108.20	100.01		102.92	97.46	83.59
1979-82	95.18	138.68	60.83	107.52	99.04	93.72	107.09	86,30	62,36	97.72	96.15	75.94
1980-83	95.32			107.71								
		134.13	64.44		100.23	94.50	107.60	84.22	66.07	97.09	97.14	76.47
1981-84	95.77	137.88	70.26	108,90	102.90	100.08	108.23	88.29	72.06	99.12	99.61	82.42

	-	AL	TERN	ATIVE	- 3							
	Agricu	Indices W	ice Indic	ac Dalati	vo to N.	ufacture			ALTER	NATIV	E - 4	
YEARS		THUILD ME	eighted b	v Alterna	tive - 3	of		ultural Pi	rice Indi	car Dolati	100 40 11	ufacturing
LAMO	1 g	07	1 4	1	A	1			leighted	by Alterna	tive - 4	orac cur mg
	Consumpt- tion Goods	ntermed- ate Goods	Investmen	and med-	and trment	Intermed- iate and investment	11.0	Goods		, 0	+	T-+
	1 5	ES	00 tt	ar	sump- n and estme	ntermed ate and nvestmer	900	300g	Goods	and med-	and tment	and tmen tmen
	no no	te te	5.00	ls un	So So Go	es	In s	0	st.	a a a	sump- n and estme	an tm
	5.5	E E	l u	Consump- tion and Intermed-	Consump- tion and Investme Goods	Inter	Consump- tion Good	ntermed- ate Goods	nves timen Goods	Consump tion an Interme	ion di nos di no	e and estme
	31	32	-	The second second		111111111111111111111111111111111111111	40	H	I I	Coltin	Consump- tion and Investme	Intermed- iate and Investmen Goods
		1 32	33	34	35	36	37	38	39	40	-	
1951-54	100.50	101.52	90.50	100.69	99.47	02.54	100 00			40	41	1 42
1952-55	90.84	97.55	93.45	91.85	91.85	93.54	100.20	98.14	90.66	100.07	99.34	95.98
1953-56	89.68	93.26	95.80	89.86	90.35	94.20	90.76	94.16	91.75	92.48	90.79	94.16
1954-57	95.42	90.35	93.22	95.17	95.28	93.42	89.81	91.61	92.80	90.59	90.08	92.58
1955-58	101.52	90.82	127.62	99.79		91.03	96.29	92.25	92.05	95.29	95.53	92.14
1956-59	103.14	94.89	95.03	101.75	100.08	92.00	102.10	94.09	93.87	99.22	100.85	93.93
1957-60	101.87	98.14	98.50	101.28	101.65	94.93	103.98	97.45	95.93	101.59	102.83	97.10
1958-61	104.08	102.44	104.11	103.71	101.25	98.30	102.56	99.52	98.69	101.47	102.14	99.31
1959-62	107.07	108.17	106.90	107.15	103.97	103.31	104.29	102.13	103.20	103.51	104.27	102.36
1960-63	109.75	112.47	106.16	110.10	107.12	107.49	107.04	104.87	105.72	106.25	106.98	105.06
1961-64	109.99	112.40	101.72	110.40	109.14	108.90	109.71	106.19	104.90	108.45	109.15	105.88
1962-65	108.93	111.46	99.30	109.41	108.32	106.35	109.89	103.82	100.74	107.69	108.72	103.30
1963-66	107.35	110.00	98.77	107.83	106.79	103.59	109.09	102.43	99.38	106.67	107.76	101.73
1964-67	106.27	111.06	99.56	107.06	105.43	102.70	107.85	100.58	99.27	105,20	106.70	100.30
1965-68	103.79	108.45	93.82	104.52	105.02	103.54	107.29	101.17	100.38	105.07	106.35	101.01
1966-69	102.60	104.77	88.16		101.86	100.06	104.93	98.99	94.75	102.77	103.46	98.13
1967-70	101.71	100.10	81.56	102.94	99.63	95.11	103.73	96.15	89.10	100.92	101.49	94.59
1968-71	104.80	1-0.55	79.45	104.12	97.40	89.24	102.86	92.03	82.29	98,76	99.53	89.81
1969-72	107.54	103.53	80.77	106.88	99.01	88.03	106.20	91.48	79.89	100.41	101.58	88.67
1970-73	110.19	114.06	83.70	110.13	101.40	89.99	108.82	91.41	80.59	102.00	103.97	88.81
1971-74	115.53	130.91	88.94	117.13	109.20	95.47 104.34	111.33	93.99	81.40	104.17	106.21	90.88
1972-75	115.92	143.04	96.90	119.04	107.49	102.47	116.07	103.30	84.21	111.06	110.36	98.70
1973-76 1974-77	112.96	143.53	92.99	116.99	104.24	99.53	113.84	112.76 113.72	81.07	114.80	108.73	102.61
1975-78	107.70	147.66	89.03	112.47	99.77	97.11	108.05	117.94	76.76 72.65	113.65	105.19	102.10
1976-79	129.22	160.22 166.59	83.72	116.17	104.59	107.59	110.47	127.06	76.06	111.10 115.46	100.41	102.86
1977-80	127,66	150.75	84.74	118.76		109.89	112.38	132.72	74.58	118.63	104.22	111.69
1978-81	107.91	125.63	74.66	115.77	104.26	103.34	110.72	117.04	68.44	112.12		114.50
1979-82	106.26	111.15	74.03	106.98	99.59 96.82	91.81	106.99	94.18	58.66	101.39	96.36	83.73
1980-83	106.78	108.33	79.89		100.60	87.55 90.86	105.84	81.02	56.36	95.48	94.77	74.43
	101.175	112.46	88.50	108.22	103.38	98.12	106.66	79.08 83.14	58.62	94.74	95.76	73.85
	1							77.77	02.30	97.25	97.61	77.99

DOMESTIC TERMS OF TRADE FOR AGRICULTURE: ALTERNATE METHOD (Three Years Moving Average)

							1			(Usin	ig GVO a	as Weigh	it for A	Agricul	ural Pr	ices)		
	Agrica Price	Indice	Price s Weigh	indices	Relati	ve to M	ant Ag	ri. Pri ice Ind	ce Indi	ces Relighted	by GVO	o Manf.	Agri. Indice	Price I s Weigh	ndices. ted by	Net Ava	to Ma	nf.Price
YEARS	.onsumption Goods	Intermediate Goods	Investment Goods	Consumptions satermediate Geods	Consumption & Invest- ment Goods	Intermediate & Investment Goods	Consumption Goods	Intermediate	Investment	Consumption & Intermed- iate Goods	Consumption &Investment Goods	Intermediate &Investment Goods	Consumption Goods	Intermediate Soods	Investment Goods	Consumption Jintermediate	Consumption & Investment Goods	Intermediate8 Investment Poods
1951-54 1952-55 1953-56 1953-56 1953-56 1955-58 1956-59 1957-60 1958-61 1959-62 1960-63 1961-64 1962-65 1963-66 1963-66 1963-66 1965-68 1966-69 1967-70 1968-71 1969-72 1970-73 1971-74 1972-75 1973-76 1974-77 1975-78 1976-79 1977-80 1978-81 1979-82 1980-83		109.54 100.18 90.29 94.43 96.68 98.55 103.07 108.06 110.61 108.43 107.83 105.87 107.05 104.89 102.67 98.76 96.87 97.99 100.30 114.65 110.49 105.42 107.56 112.95	91.71 94.05 91.61 92.62 92.35 95.81 98.47 103.52 103.78 102.63 98.26 101.22 102.83 107.17 102.95 93.03 89.59 85.34 85.70 85.18 85.92 79.17 73.75 70.16 76.33 74.43 67.24 55.31 53.20	94.98 89.98 87.11 92.62 96.22 98.59 99.09 102.82 105.81 107.98 107.75 109.76 110.90 115.73	92.92 88.38 86.30 91.41 95.70 98.22 99.1 103.02 105.65 107.07 106.17 108.13 109.67 114.73 114.10 112.86 108.73 108.31 110.84 111.92 115.22 112.24 106.26 99.87 102.62	99.84 99.10 94.41 92.12 93.04 96.02 98.50 103.15 106.10 106.07 102.44 103.49 104.46 108.54 101.60 94.71 91.08 91.38 91.38 91.18 87.01 83.15 88.85 88.34 80.01 66.09 61.85	90.10 87.22 93.25 96.18 98.13 98.37 102.44 105.57 107.75 107.97 111.14 113.38 119.30 119.97 120.61 118.70 120.43 124.23 127.40 132.58 133.47 127.02 119.84 120.25 124.22 124.40 121.02 120.72	108.11 110.91 109.04 109.36 108.01 109.58 106.83 104.04 99.38 97.45 98.58 101.11 110.13 116.08 111.18 105.35 105.67 109.98	106.28 105.14 100.60 101.20 102.62 106.70 102.21 97.20 88.78 84.65 85.12 85.18 86.47 79.90 74.48 71.03 77.61 76.25 69.53 58.14 56.30	105.74 107.97 108.04 109.89 110.75 113.86 114.01 113.98 112.67 113.57 116.66 118.81	94.36 59.14 86.52 91.29 95.36 97.57 98.42 102.85 105.80 107.33 106.49 108.11 109.20 113.67 112.93 111.89 108.11 107.91 110.53 111.97 113.34 107.83 101.95 105.19 106.60	106.60 103.09 104.03 104.87 107.66 104.10 100.00 93.80 90.19 90.65 90.80 94.68 91.65 87.36 83.17 88.75 88.48 80.68 67.44	91.17 87.41 92.36 97.31 99.52 99.42 102.93 105.98 107.73 107.41 109.48 110.85 115.49 115.98 116.61 115.00 116.24 119.12 121.18 126.83 129.17 125.64 120.55 123.06 124.54 119.00 118.64	104.93 101.32 94.84 92.27 92.17 94.57 97.21 101.85 106.93 108.85 106.37 107.78 108.21 112.45 110.98 108.91 104.85 103.72 105.39 112.79 116.91 113.83 112.79 118.67 125.46 114.14	104.18 99.56 99.33 100.67 105.39 102.16 97.73 89.91 85.92 87.19 89.06 92.46 86.22 80.36 76.60 84.42 84.81 79.62 70.16 68.90 73.25	112.74 112.49 110.08 110.20 111.77 112.90 119.04 122.60 120.06 116.37 117.46 121.60 108.08 102.74	108.44 103.07 101.37 103.30 105.06 109.79 106.63 102.66 98.19 104.53 105.63	102.35 103.88 108.96 106.35 102.65 95.95 92.40 93.49 94.51 99.48 95.81 91.76 88.30 96.20 97.66 90.72 77.99 73.58

		ALTERNAT	TIVE - 1							ALTE	RNATIVE .	- 2	29.1
YEARS	Manufac by Alte	tural Proturing Property	rice In	ices Rela dices Wei	tive to ghted		Man by	ufact Alter	tural Pri turing Pr mative -	ice Ind	ces Rela	tive to ghted	
	Consumption Consumption Goods	intermed-	Investment	Consumption & Inter- mediate Goods	Consumption & Invest- ment Goods	Intermed- iate and Investment Goods	~	Consumptior	Intermed- iate Goods	Investment Goods	Consumption & Intermed- iate Goods	Consumption &Investment Goods	Intermed- iate and Investment
1951-54 1952-55 1953-56 1954-57 1955-58 1956-59 1957-60 1958-61 1959-62 1960-63 1961-64 1962-65 1963-66 1964-67 1965-68 1966-69 1967-70 1968-71 1969-72 1970-73 1971-74 1971-74 1971-75 1971-78 1971-78 1971-78 1971-78 1971-80 1971-80 1971-80	97.73 90.98 87.62 92.79 97.95 100.31 100.09 103.26 106.04 107.76 107.45 102.89 100.91 101.42 104.56 104.99 103.40 104.60 106.81 108.04 112.15 113.69 110.64 109.33 104.54	97.93 96.03 89.24 85.96 86.12 91.24 95.95 102.19 109.47 114.77 115.64 122.45 126.66 137.74 138.13 135.54 129.00 128.00 133.84 150.49 174.10 191.07 190.33 194.99 210.38 210.38 210.70 168.41	88.64 92.57 92.24 89.62 90.48 93.02 97.12 103.30 105.73 104.16 99.54 98.22 98.58 102.20 99.12 94.88 87.30 83.42 84.63 86.46 89.73 83.65 77.92 74.27 81.84 82.19 77.13 67.62	120.67 127.63 131.22 128.92 124.22	96.54 90.14 86.87 92.51 97.29 99.82 99.96 103.43 106.19 107.83 107.01 109.39 111.35 116.32 115.99 113.86 114.49 117.24 117.97 122.64 117.97 122.64 117.97 120.57 118.86 110.14	93.30 94.37 89.95 87.28 88.02 92.55 96.70 102.41 107.12 109.11 107.26 108.89 110.65 117.27 116.71 113.96 107.41 104.83 107.36 113.71 125.74 130.08 127.83 127.20 138.36 141.44 130.10 110.01	99 99 100 100 100 100 101 111 111 112 122 123 130 126 127 127 127 127 127 127 127 127 127 127	7.41 0.80 7.54 2.02 8.23 2.05 1.74 4.79 6.31 8.18 7.98 0.23 1.94 6.91 7.64 8.68 8.27 6.68 8.27 6.29 9.20 9.20 9.20 9.20 9.20 9.20 9.20 9	97.02 95.50 90.61 89.23 91.71 96.01 99.09 101.81 104.49 105.24 102.66 105.42 107.68 114.32 108.96 106.47 106.65 108.85 119.44 132.38 133.21 137.97 147.05 153.66 136.01 108.45	88.82 92.82 92.44 89.79 90.59 93.13 97.19 103.6 105.59 104.02 99.10 99.01 100.09 104.48 101.24 96.83 85.03 85.03 85.24 88.99 91.47 85.27 79.46 75.76 83.54 83.97 78.79 69.27	97.36 93.35 88.69 92.73 96.30 98.53 99.34 102.63 105.62 107.72 107.31 109.20 109.88 114.35 115.08 115.02 112.21 114.12 115.20 122.12 127.50 125.94 122.90 126.20 129.79 123.59 111.63	96.48 90.95 88.29 92.50 96.99 99.45 99.99 103.62 106.28 107.53 106.51 108.20 109.82 114.69 114.56 114.50 110.90 110.83 113.32 114.51 109.63 113.47 115.20 112.81 105.72	93.05 94.36 91.03 89.14 90.29 94.23 97.65 102.28 104.89 104.83 101.55 102.72 104.11 109.90 109.28 106.80 100.56 97.37 97.30 99.32 106.88 110.10 108.53 117.64 120.72 109.42 90.65

			ALTERNAT	IVE - 3				1	LTERNAT	IVE - 4		
	Agricu Manufa	Itural Pr	ice Indic	es Relati	ve to		Agricul	tural Pi				p) h
YEARS	Consumption	Intermed.	Investment	Consumption & Intermed- iate Goods	Consumption & Invest- ment Goods	Intermed- fate and Investment Goods	Consumption	Intermed-	Investment	Consumption & Intermediate Goods	Consumption & Invest	Intermed intermed Investment
1951-54 1952-55 1953-56 1953-56 1955-58 1956-59 1957-60 1958-61 1959-62 1960-63 1961-64 1962-65 1963-66 1964-67 1965-68 1966-69 1967-70 1968-71 1969-72 1970-73 1971-74 1972-75 1973-76 1974-77	97.61 90.87 87.52 92.87 98.00 100.59 100.35 103.66 106.61 108.48 108.35 110.00 111.30 115.55 116.15 116.68 115.08 115.39 119.39 120.84 126.07 128.03 124.64 119.07 138.77	98.78 97.43 91.09 87.44 87.60 92.54 96.69 102.04 107.49 110.51 112.54 114.09 120.81 121.31 119.18 113.30 111.68 114.95 124.96 142.70 158.19 158.46 163.15 175.45	88.53 93.33 93.56 90.24 90.66 92.72 97.08 103.69 106.24 104.77 100.02 100.38 102.51 108.23 104.87 100.30 92.34 88.26 89.69 91.78 97.06 108.17 103.80 99.62 91.60	97.80 91.86 87.70 92.15 96.33 99.23 99.76 103.29 106.49 108.61 108.55 110.47 111.80 116.41 116.95 117.07 114.80 115.63 118.66 120.76 127.80 131.50 129.08 124.34 127.09	95,66 91,05 88,18 92,25 96,61 99,75 103,54 106,46 107,68 106,50 107,86 109,34 114,19 113,95 113,31 110,21 109,97 112,57 114,21 119,16 118,62 114,92 110,22 114,42	91.32 94.08 91.28 88.16 88.80 92.61 96.86 102.89 106.82 107.45 104.57 104.61 106.53 112.62 111.88 108.20 101.03 97.19 99.92 104.64 113.81 112.94 109.58 107.10 117.75	97.31 90.78 87.65 93.23 98.56 101.40 101.01 103.86 106.37 108.23 108.05 110.17 111.84 116.68 117.43 117.96 116.37 117.74 120.80 122.08 127.31 129.15 125.60 119.47 120.83	95.42 94.04 89.50 89.37 90.84 95.05 98.03 101.72 104.24 104.78 102.08 103.42 104.31 110.05 110.77 109.37 104.17 101.61 101.49 103.04 112.65 124.84 125.67 120.42 129.17	88.57 91.65 90.63 89.13 90.60 93.49 97.26 102.79 105.08 100.61 102.04 103.05 109.12 105.90 101.36 93.18 88.75 89.49 89.29 91.93 89.46 84.67 80.31 83.23	114.27 115.00 114.79 111.76 111.53 113.24 114.23 121.17 126.90 125.45 122.85	106.90 108.85 110.66 115.65 115.75 115.43 112.61 112.82 115.42 116.48 120.43 120.01 116.01	94.70 97.83 101.94 104.42 104.47 101.38 102.75 104.05 109.86 109.78 107.61 101.67 98.50 99.65 107.66 113.41 112.67 113.65

Table-A 3.9

DOMESTIC TERMS OF TRADE FOR AGRICULTURE: ALTERNATIVE NETHOD

(Three years moving average)

(Using Value added as Weight for Agricultural Prices)

				es Relat			Agricul turing			ices Rela								o Manufac- Availability,
74 73 75 75	Consumption Goods	Intermediate	Investment Goods	Consumption and Intermediate Goods	Consumption and Investment Goods	Intermediate and Investment Goods	Consumption	Intermediate	Investment Goods	Consumption and Intermediate Goods	Consumption and Investment Goods	Intermediate and investment Goods	Consumption	Intermediate	Investment Goods	Consumption and Intermediate goods	Consumption and Investment Goods	Intermediate and Investment Goods
- 07	127	(3)	165	(5)	(6)	(7)	(8)	123	(30)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	-
1951-24 1952-35 1953-56 1954-57 1955-58 1956-59 1957-50 1938-61 1939-62 1960-65 1961-64 1962-65 1964-67 1965-68 1964-67 1967-70 1968-71 1969-72 1970-73	91.09 87.09 84.01 39.61 94.03 100.45 101.54 105.89 104.71 105.94 103.70 109.73 113.15 121.97 124.73 127.26 125.41 127.20 131.12 138.98	111.96 109.00 98.67 93.45 91.78 95.02 97.78 103.05 73.91 108.86 106.29 105.74 104.05 106.53 106.00 105.45 101.75 99.88 100.90 103.00	89.89 93.58 90.20 88.15 89.77 93.53 97.05 102.84 103.05 101.06 96.32 99.28 101.04 106.66 104.00 100.66 92.30 87.99 88.23 87.48	92.77 89.54 85.70 90.40 93.53 96.23 97.65 102.14 105.02 106.27 105.63 107.65 108.99 115.22 117.33 119.14 116.87 117.86 120.48 122.20	92.39 89.55 84.90 93.02 95.87 97.59 102.34 104.87 105.39 104.08 106.05 107.78 114.22 115.33 115.91 112.02 111.68	97.62 98.50 92.97 69.94 90.44 93.86 97.08 102.48 105.32 104.43 101.50 102.66 108.02 106.55 104.34 97.60 93.93 94.11 93.47	92.25 87.87 84.15 89.37 95.78 96.95 101.77 104.79 106.05 105.84 109.00 111.42 118.78 121.29 123.87 122.30 124.16 127.90 130.83	116.95 113.78 95.97 94.27 97.08 98.36 103.44 107.30 109.16 106.89 107.25 106.15 109.94 107.95 106.85 102.39 100.48 101.50 103.84 101.50	90.60 94.61 91.32 88.68 89.90 93.46 97.11 103.31 105.50 103.52 98.50 99.25 100.84 106.19 103.25 99.81 91.44 87.27 87.64	93.54 89.55 85.96 90.34 93.54 95.97 97.27 101.88 104.95 106.26 105.91 107.77 108.84 113.33 115.24 117.08 116.08 117.09 120.12	93.07 89.60 85.11 89.09 92.68 95.23 76.99 102.16 105.64 104.39 106.03 107.32 113.17 114.14 111.39 111.25 113.79 114.99	99.59 100.04 93.99 91.24 91.44 94.79 97.79 103.11 72.37 104.94 101.05 102.03 103.05 107.12 105.17 102.70 96.44 92.99 93.33 93.25	95,89 90,74 85,99 90,12 94,59 97,14 97,98 102,25 105,19 106,03 107,38 108,94 114,98 117,26 119,77 118,49 119,84 122,65 124,46	102.57 100.81 93.40 90.11 99.59 92.31 95.81 101.19 106.13 107.16 104.27 105.70 106.34 111.86 108.03 106.05 106.79	87.01 97.12 90.82 87.58 88.11 90.96 95.81 102.61 104.95 102.59 97.61 97.42 98.92 104.91 103.23 100.36 92.62 88.58 89.77 91.46	97.10 93.31 87.88 90.44 93.12 95.77 97.46 102.05 105.48 106.37 105.15 107.20 112.32 113.97 115.54 113.62 113.62 113.62	93.13 90.94 87.45 88.81 91.71 94.41 97.07 102.49 105.20 104.71 102.04 103.26 104.95 111.24 111.68 111.36 106.19 104.51 106.36	92,46 94,96 91,50 87,74 87,88 90,89 95,82 102,07 105,19 103,84 99,52 100,38 102,08 102,08 102,08 103,46 107,47 105,41 98,65 95,26 96,15 97,05
1972-73	140.02	112.04	87.94	127.56	117.93	96.53	135.69	119.45	88.51	117.08	118.51	96.90	132.89	115.42	94.63	121.82	112.36	101.81
1973-76 1=75-72 1976-78 1976-78	133.12 124.70 373.11 127.93 178.81 124.97	114.30 109.19 111.57 116.80 107.92	76.27 72.66 79.20 77.00 69.97 57.33	125.46 118.35 118.66 122.49 120.82 113.30	109.92 103.45 106.45 106.99 103.68 94.86	90.00 86.11 92.17 91.36 83.25 68.51	131.43 124.16 124.74 128.44 129.37 125.47	109.13 109.60 113.72 104.90 88.09 78.37 73.68	77.03 73.55 60.51 78.86 72.35 60.26	115.42 119.26 119.82 123.55 122.09 114.79	111.55 105.61 109.12 110.27 107.52 99.35	90.35 86.13 92.06 91.50 83.94 69.90	130.01 124.88 127.65 130.37 129.52 123.36	117.78 116.78 123.07 129.71 118.70 96.12 61.30	83.10 79,31 87.58 87.71 82.83 72.72	124.24 120.54 121.81 125.70 121.25 212.03	106.19 101.69 108.44 109.23 107.73 98.31	94.90 91.42 99.78 100.99 94.37 80.63

Agri	culture	Price In Indices	dices Re	lative to	Manufac-	Agricu	Price I	ce Indi	ces Relat	tive to )	Manufac- native-2 of		Agrico Mfg. P	rice Ind	rice Indi	ces Helm	Alt -3
Cone umption Goods	Intermediate	Investment Goods	Consumption and Intermediate Goods	Consumption and Invastment Goods	Intermediate & Investment Goods	Consumption Goods	Intermediate Goods	Investment Goods	Consumption and Intermediate Goods	Consumption and Investment Goods	Intermediate and Investment Goods	Consumption Goods	Intermediate	Investment Goods	pur	Consumption and Investment Goods	Intermediate and Investment Goods
(20)	(21)	(22)	(23)	(24)	(25)	(26)	(27)	(28)	(29)	(30)	(31)	(32)	(33)	(34)	(35)	(36)	(37)
95.46	95.76	86.93	95.48	94.30	91.36	:95.14	94.86	87.11	95,12	94.28	91.12	95.33	96.60	86.86	95.53	94.44	89.49
90.55	95.53	92.10	91.59	89.73	93.89	90.36	95.01	92.34	91.90	90,52	93.89	90.43	96.93	92.86	91.42	90,61	93.59
86.20	87.91	90.79	86.61	85.47	88.60	86.12	89.23	90,99	87.28	86.68	89.56	86.10	89.81	92.09	86.29	86.75	89.88
90.55	83.96	87.49	90.24	90,28	85.23	90.94	87.13	87.65	90.51	90.07	86.96	90.63	85.48	88,10	89,93	90.02	₩.08
95,20	83.72	87.95	93.61	94,57	85.56	95,48	88.05	88.08	93.61	94.08	87.68	95.51	85.23	88.12	93.64	93.90	86.3
97.90	89,06	90.81	96.78	97.43	90.34	99,61	95.31	90.92	96.17	97.07	91.98	98.17	90,33	90.52	96.85	96.77	90.4
98.62	101.53	95.74	98.21	98.50	95.31	100.25	96,55	95.80	97.89	98,54	96.24	98.88	95.30	95.70	98.31	98.29	95.4
2.57	108.64	102.62	102.57	102.74	101.75	104.08	101.14	102.95	101,96	102,92	101.60	102.96	101.38	103.00	102.60	102.85	102.2
	112,93	104.95	105.69	105.40	106.32	105.52	103.73	105.09	104.84	105.49	104.12	105.61	106.69	105.46	105.69	105,66	106.0
6.06	113.36	102.56	106.99	106,13	107.40	106.47	103.59	102.42	106.03	105.83	103.19	106.57	109.19	103.15	106.90	105.99	105.7
5.34	119.11	97.58	106.26	104.91	105.14	105.85	100.63	97.15	105.20	104.42	99.54	106.03	108.33	98.05	106.42	104.44	102.5
0.91	124.46	96.33	103.09	107.29	106.79	108.11	103.39	97.11	107.10	106.12	100.74	107.89	110.37	98.46	108.35	105.78	102.5
9.18	137.17	96.88	105.53	109.43	108.74	110.00	105.82	98.36	107.99	107.93	102.31	10.37	112.12	100.74	109.87	107.45	104.69
01.01	139,63		115.59	115.79	116.77	116,40	114.50	104.01	113.85	114.19	109.43	115.04	120.31	107.74	115.90	113.68	112.14
05.70	139.20	101.74	118.00	117.25	117.97	118.94	117.01	102.30	116.35	115.81	110.47	117.42	122,63	105.97	118.24	115.19	113.0
07.85	132,91	97.43	120.12	118.62	117.03	121.48	117.41	99.43	118.13	117.09	109.68	119.84	122.40	102.99	120.25	116.38	111.1
06.54	132.04	89,95		117.31	110.66	120.22	112.26	91.72	115.61	114.26	103.61	118.56	116.73	95.14	118.28	113.55	104.09
07.85	7.000.00	40.000	118,32	118.04	108.08	121.69	109.77	87.67	115.69	114.26	100.40	120.00	115.15	90.99	119.22	113.38	100.8
9.97	137.80	86.01	119.31	120.71	110.54	124.65	109.77	88.79	117.50	116.68	100.70	122.93	118.36	92.34	122.18	115.91	102.8
10.96	178.13	87.13	122.21	121.15	116.76	126.24	111.79	90.47	118.32	117.67	102.00	124.11	128,30	94.24	124.02	117.29	107.43
14.77	196.61	88.78	123.93	125.51	128,66	131.24	122.22	93,62	124.97	121.82	109.38	129.02	146.00	99.33	130.78	121.95	115.46
16.96	196.99	91.83	130.61	127.13	133.80	134.02	136.30	87.63	131.22	121.70	113.27	131.72	162.81	111.44	135.30	122.01	116.12
		85.97	7.00		3500 000	701100		82.17	130.34	116.48	112.34	128.97	164.01	107.50	133.58	118.91	113.35
14.55	201.85	80.58	133.41	124.34	132.26	130.68	137.90		127.31	113.56	112.36	123.36	168.90	103,28	128,81	114.17	110.88
12.29	218.18	76.90	128.69		131.70	124.76		78.44	130.91	117.70	122,02	144.12	181.96	95.03	131.83	118.69	122.14
	225.79	83.89	131.63	123.49	143.52	126.55	152,50			100000	124.84	147.03	188,92	96,09	134,60		
14.40	208.76	84.99	134.64	124,70	146,27	128,80	158,91	86.84	134,21	119.12	113.83	146.07	172.18	92.76	132.07	120.66	124,6
13,70	174.53	80.24	132.25	123.64	135.35	127.74	141.50	82.07	128.55		93.95	121.34	141,19	83.94	123.30	109.42	117.98
08.37	156,86	70.09	123.75	114.18	114.01	115.67	112.37	71.80	115.71	109.60	85.90	120.17	125.72	83.74	120.99		101.01
07.64	152.10	68.80	121.60	112.01	105.97	115.11	97.61	70.53	110.52	108.74	4					108.88	96.83
08,08	155.77	73.10	122.14	113.66	107.18	116.00	95.51	74.96	111.08	110.15	86.73	121.08	122.85	90.64	121.37	111.43	100.87
08.21	153,87	79.36	123.05	116.25	113.06	122.30	99.71	81,39	111.98	112.54	91.96	121.42	127.05	99.95	122.28		110.82
											W					Contini	ed

Intermediate & Investment Goods	(43)	91,49	93,55	0	87,14	88,14	92,43	96,42	101,28	103,66	ထ	96,38	100,77	102,24	109,39	110,97	110,52	104,74	101,55	101,52	102,34	110,17	116,70	116,60	117,66	126,85	129,88	116,77
Consumption and Investment Goods	(42)	94,80	90,36	86,49	90,26	94.63	97.88	99,14	,1	•			105,76	108,74	115,14	117,02	118,55	116,03	116,32	118,85	119,63	123,26	123,46	120,04	114,95	118,26	119,26	116,49
Dns noitemens and Intermediate Boods	(41)	94,95	6		90.07	93,10	69.96	98,40		104,82	•	00	•	107,21	113,78	116,27	117,90	115,14	114,99	116,60	117,31	123,99	130,59	129,83	127,24	131,05		127,96
Investment Goods	(07)	86,82	7	93,59	91,38	95.44	91,36	95,86		•	0	~		101,27	10	10,701	104,09	00*96	91,50	92,14	91,70	60°56	92,01		83,19	86,35	- 4	78.21
ats ibamratnI Boood	(38)	93,29	93,56		87,26	88,29	92,76	96,61	101,06	•	1.03,14	•		102,51	109,58	111,98	112,26	107,32	104077	104,50	105,82	115,28	128,53	130,10	135,02	144,33	000	133,74
Cons <i>u</i> mption Goods	(38)		3	•	86.06		8,9	9.5	1	•	5	•	08°	6 60	116,16	118,72	121,16	119,90	121,40	124,38	125,38	130,29	132,87	129,97	123,77	125,34	127,36	126.28

Table-A 3.10

## INTRA SECTORAL TERMS OF TRADE FOR MANUFACTURING SECTOR (THREE YEARS MOVING AVERAGE)

YEARS	CONSUMPTION GOODS RELATIVE TO INTERMEDIATE AND INVESTMENT GOODS						
	Value added	GVO	Net avail- ability	Alt. 1	Alt. 2	Alt. 3	Alt.
1951-54	108.21	108.76	97.24	96.52	96.64	94.89	97.07
1952-55	113.92	114.55	105.49	104.51	104.70	104.37	104.32
1953-56	111.21	112.14	106.58	103.20	104.36	104.66	103.59
1954-57	100.88	102.66	97.93	94.77	96.36	95.49	96.23
1955-58	96.19	97.84	93.01	89.93	91.90	90.67	91.95
1956-59	97.12	98.99	93.58	93.27	92.33	92.07	93.39
1957-60	99.33	100.87	97.77	96.61	96.23	96.53	96.87
1958-61	100.33	101.30	99.77	99.09	97.59	99.21	98.12
1959-62	100.56	100.85	99.99	100.99	98.68	100.38	98.21
1960-63	98.50	98.90	97.86	101.25	96.86	99.19	96.49
1961-64	95.03	95.50	94.52	99.83	94.06	96.69	93.84
1962-65	92.53	93.60	93.45	106.44	95.13	95.11	93.28
1963-66	90.73	92.47	93.67	110.23	94.93	95.73	97.08
1964-67	88.71	90.39	94.38	115.77	95.92	97.45	98.21
1965-68	85.51	86.86	91.71	111.58	92.88	-96.32	97.54
1966-69	81.85	82.80	87.86	108.35	90.16	92.54	91.09
1967-70	77.84	79.06	83.44	103.87	86.20	87.81	87.38
1968-71	73.91	74.98	79.56	100.27	82.59	84.09	83.74
1969-72	71.77	72.98	78.40	100.52	80.80	83.70	81.63
1970-73	69.73	71.30	77.97	105.18	80.78	86.55	81.61
1971-74	69.47	71.42	78.46	111.88	83.21	90.18	84.38
1972-75	66.90	68.58	74.12	114.34	84.45	88.16	87.74
1973-76	67.65	68.75	72.99	115.51	85.99	87.98	89.72
1974-77	69.31	69.59	73.40	119.80	90.19	90.19	95.13
1975-78	73.94	73.65	78.04	127.42	96.07	87.46	95.83
1976-79	71.37	71.19	77.38	127.67	96.77	87.29	96.82
1977-80	64.24	64.51	72.47	118.29	88.44	82.50	86.69
1978-81	54.73	55.65	65.43	105.04	77.06	84.94	78.03
1979-82	51.60	52.65	2.02	98.48	70.94	82.39	70.34
1980-83	52.21	53.36	62.64	99.15	71.07	85.09	69.57
1981-84	55.05	56.23	66.17	104.49	75.22	91.31	73.11

Table-A 3.11

### INTRASECTORAL TERMS OF TRADE FOR AGRICULTURAL SECTOR

	FOOD	CROPS	RELATIVE	TO CASH	CROPS	
a mantelvet e	MARKETIN	igs	GROSS VAI	UE OF OUTPUT	NET AVAIL	ABILIT
TEARS	Absolute	3 Years Moving Average	Absolute	3 Years Moving Average	Absolute	3 Year Moving Average
1951-52	59.57	-	61.14	1901	65.15	_
1952-53	106.07	87.16	110.89	89.29	117.36	93.50
1953-54	95.84	94.88	95.83	96.33	97.98	99.2
1954-55	82.72	87.02	82.26	87.20	82.49	88.7
1955-56	82.50	84.62	83.50	84.91	85.77	87.7
1956-57	88.64	87.41	88.96	86.91	95.07	91.0
1957-58	91.09	92.22	88.27	90.92	92.42	96.3
1958-59	96.94	96.01	95.53	94.60	101.46	97.9
1959-60	100.00	98.43	100.00	98.05	100.00	103.0
1960-61	98.35	96.98	98.63	96.65	104.82	102.0
1961-62	92.58	93.44	91.32	92.46	101.34	101.2
1962-63	89.39	95.41	87.42	91.36	97.45	100.2
1963-64	104.27	96.44	95.33	89.95	102.09	97.3
1964-65	95.66	97.11	87.09	88.58	92.55	94.1
1965-66	91.40	106.33	83.31	95.84	87.78	100.2
1966-67	131.94	124.69	117.13	111.46	120.36	111.6
1967-68	150.73	137.00	133.94	121.41	126.93	118.1
1968-69	128.34	136.25	113.17	125.55	107.28	112.4
1969-70	129.67	126.31	109.55	110.85	103.10	105.4
1970-71	120.93	123.56	109.82	109.29	105.82	103.3
1971-72	120.07	120.73	108.49	108.64	101.10	102.6
1972-73	121.19	121.16	107.61	107.17	101.10	100.8
1973-74	122.21	127.64	105.41	111.96	100.47	104.5
1974-75	139.52	129.42	122.87	113.08	111.95	108.5
1975-76	126.54	126.06	110.95	111.11	113.10	106.7
1976-77	112.12	123.45	99.52	109.92	95.16	106.3
1977-78	131.70	118.90	119.28	108.70	110.72	100.5
1978-79	112.87	121.94	107.29	113.44	95.68	102.1
1979-80	121.26	117.53	113.74	109.51	100.06	97.2
1980-81	118.46	121.44	107.48	112.58	96.08	100.6
1981-82	124.60	83.15	116.53	113.97	105.87	103.1
1982-83	124.84	119.07	117.89	112.99	107.55	103.7
1983-84	107.76	<u> </u>	104.56	_	97.68	

*0.
S
-
-62
-
38
ment
1960-61
30
-
-
959-60
50
195
1958-59
60
33
n-4
00
5
1937-58
and.
-
1956-57
33
4-6
9
1
1955-56
good
103
1
954-55
-4
-
1953-54
3
000
~
3
33
CO's
2
-
33
-

84.64 79.19 61.33 93.66 32.80 113.03 86.41 86.41 86.41 87.60 82.80 63.76 102.69 87.20 54.10 54.10	121.71 110.92 87.93 88.40	100 18	
13. 19. 18. 26   97.50   103.73   91.61   100.00     13. 101.33   64.24   79.76   87.12   84.52     13. 13. 13. 14. 22.47   87.44   86.70   88.85     13. 13. 13. 12. 12. 100   116.93   117.03   111.69     86. 41   95.32   112.92   89.98   84.63   90.87     82. 40   90.79   92.47   87.44   86.70   88.85     82. 80   90.79   92.47   87.44   86.70   88.85     82. 80   90.79   92.47   87.44   86.70   88.85     82. 80   90.79   92.47   87.44   86.70   88.85     82. 80   90.79   92.47   87.44   86.70   88.85     82. 80   90.79   92.47   87.44   86.70   88.85     96. 13   89.72   83.31   91.27   82.07     96. 13   89.72   83.31   83.31   91.32   90.19     96. 13   89.72   83.31   83.31   91.32   90.19     96. 13   96. 28   91.70   80.59   90.37   96.67     96. 14   99.64   97.33   87.55   100.55   88.86     96. 15   84.37   76.52   74.64   74.64   126.56     96. 15   84.37   76.52   74.64   74.64   126.56     96. 15   84.37   76.52   74.64   74.64   76.55     96. 15   76. 26   77.32   87.55     96. 15   76. 27   76.55     96. 15   76. 27   76.65     96. 15   76. 27   76.65     96. 16   76. 27   76.65     96. 17   76. 27   76.65     96. 17   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18   76. 27     96. 18     96. 18   76. 27     96. 18     96. 1	121.71 110.92 87.93 88.40	100.10	
61.33 61.33 64.24 79.76 87.12 84.52 93.66 91.89 90.36 84.01 88.83 84.01 32.80 90.79 92.47 87.44 86.70 88.85 113.03 119.12 121.00 116.93 117.03 111.69 86.41 95.32 112.92 89.98 84.63 90.87 86.41 95.32 112.92 89.98 84.63 90.87 82.80 90.79 92.47 87.44 86.70 88.85 83.24 82.80 90.79 92.47 87.44 86.70 88.85 83.24 82.80 90.79 92.47 87.44 86.70 88.85 83.24 82.80 90.79 92.47 87.44 86.70 88.85 65.36 65.36 65.36 90.24 95.87 87.44 86.70 88.85 95.35 90.24 95.87 97.06 91.27 82.07 102.69 102.69 102.69 102.69 111.47 95.12 83.31 91.32 90.19 95.13 89.72 83.31 91.32 90.19 95.13 87.20 96.28 91.70 80.59 90.37 96.67 99.32 90.14 99.64 97.33 87.55 100.55 88.86 98.37 76.52 74.64 74.64 126.56 99.37 96.57 98.31 97.52 80.40 99.32 90.40 97.33 87.55 100.55 88.86 97.32 80.40 99.54 97.33 87.55 100.55 88.86 97.55 74.64 74.64 126.56 97.55	87.93 88.40	00 100.00	
93.66 91.88 90.36 84.01 88.83 84.01 32.80 90.79 92.47 87.44 86.70 88.85 113.03 115.02 112.92 89.98 84.63 90.87 86.41 95.32 112.92 89.98 84.63 90.87 82.80 90.79 92.47 87.44 86.70 88.85 83.24 82.80 90.79 92.47 87.44 86.70 88.85 83.24 82.80 90.79 92.47 87.44 86.70 88.85 83.24 82.80 90.79 92.47 87.44 86.70 88.85 83.24 92.47 87.44 86.70 88.85 83.24 95.75 82.07 102.69 102.69 102.69 102.69 111.47 87.44 86.70 88.85 83.87 83.23 90.24 95.87 95.13 83.31 91.32 90.24 95.87 95.13 83.31 91.32 90.19 95.13 87.20 96.28 91.70 80.59 90.37 96.67 93.32 90.14 99.64 97.33 87.55 100.55 88.86 95.70 84.37 76.52 74.64 126.56 97.92 97.92 80.40 99.32 97.97 97.52 87.45 97.92 80.40 99.32		93.04	
32.30 90.79 92.47 87.44 86.70 86.85 113.03 119.12 121.00 116.93 117.03 111.69 86.41 95.32 112.92 89.98 84.63 90.87 85.42 82.80 90.79 92.47 87.44 86.70 88.85 83.24 82.80 90.79 92.47 87.44 86.70 88.85 63.76 69.09 88.44 97.06 91.27 82.07 102.69 154.03 102.69 102.69 102.69 111.47 82.07 102.69 154.03 102.69 102.69 102.69 111.47 82.07 102.69 154.03 102.69 102.69 102.69 111.47 82.07 102.69 154.03 102.69 102.69 102.69 111.47 82.07 102.69 154.03 102.69 102.69 111.47 82.07 102.69 154.03 102.69 102.69 111.47 82.07 102.69 154.03 102.69 102.69 111.47 82.32 90.24 82.32 90.37 96.67 83.31 91.32 90.37 96.67 87.20 99.24 97.33 87.55 100.55 88.86 97.20 99.32 88.86 97.37 76.52 74.64 126.56 97.37 96.67 97.32 84.37 76.52 74.64 74.64 126.56 97.37 96.67 97.32 84.37 76.52 74.64 74.64 126.56 97.37 96.67 97.32 80.60 99.32	90.36 93.66	00 108,28	
86.41 95.32 112.92 89.98 84.63 90.87  86.41 95.32 112.92 89.98 84.63 90.87  82.48 61.20 72.49 57.80 54.26 58.42  82.80 90.79 92.47 87.44 86.70 88.85  63.76 69.09 88.44 97.06 91.27 82.07  102.69 154.03 102.69 102.69 102.69 111.47  53.23 46.18 49.99 53.94 66.34 67.84  96.13 89.72 83.31 83.31 91.32 90.19  54.10 57.74 68.01 61.56 70.52 85.87	93.28 93.31	19.76 00	
86.41 95.32 112.92 89.98 84.63 90.87  55.48 61.20 72.49 57.80 54.26 58.42  82.80 90.79 92.47 87.44 86.70 88.85 65.76 69.09 88.41 97.06 91.27 82.07 102.69 154.03 102.69 102.69 111.47  53.23 46.18 49.99 53.94 66.34 67.84 96.13 89.72 83.31 91.32 90.19 54.10 57.74 68.01 61.56 70.52 85.87	104.65 100,11	100.00 100.15 101.57	11.57 118.51
55.48       61.20       72.49       57.80       54.26       58.42         82.90       90.79       92.47       87.44       86.70       88.85         82.80       90.79       92.47       87.44       86.70       88.85         63.76       69.09       88.41       97.06       91.27       82.07         102.69       154.03       102.69       102.69       102.69       111.47         85.72       46.18       49.99       53.94       66.34       67.84         96.13       89.72       83.31       91.32       90.19         54.10       57.74       68.01       61.56       70.52       85.87         87.20       96.28       91.70       80.59       90.37       96.67         87.20       96.28       91.70       80.59       90.37       96.67         100.14       99.64       97.33       87.55       100.55       88.86         84.37       76.52       74.64       126.56         84.37       76.52       74.64       126.56         84.37       76.52       74.64       74.64       126.56         84.37       76.52       79.52       80.40       99.64     <	102.67 96.21	100.00 99.49 100.39	10.39 96.63
82.90 90.79 92.47 87.44 86.70 88.85 63.76 65.34 67.84 96.13 89.72 83.31 83.31 91.32 90.24 65.34	50.73 66.94	101.04	
82.80 90.79 92.47 87.44 86.70 88.85 65.76 65.76 59.09 88.41 97.06 91.27 82.07 102.69 102.69 102.69 102.69 111.47 85.32 46.18 49.99 53.94 66.34 67.84 95.87 53.23 46.18 89.72 83.31 91.32 90.19 96.13 89.72 83.31 83.31 91.32 100.00 54.10 57.74 68.01 61.56 70.52 85.87 66.24 67.55 110.90 87.20 96.28 91.70 80.59 90.37 96.67 84.37 76.52 74.64 74.64 126.56 88.36 84.37 76.52 74.64 74.64 126.56 99.32 84.37 76.52 74.64 74.64 126.56 99.32	85.92 87.07	98.18	
82.86 96.79 92.47 87.44 86.70 88.85 63.76 69.09 88.44 97.06 91.27 82.07 102.69 102.69 102.69 102.69 111.47 80.38 82.32 90.24 96.13 89.72 83.31 91.32 90.24 96.13 89.72 83.31 91.32 90.19 96.13 89.72 83.31 91.32 90.19 96.13 89.74 68.01 61.56 70.52 85.87 96.67 87.20 96.28 91.70 80.59 90.37 96.67 84.37 76.52 74.64 74.64 126.56 84.37 76.52 79.32 80.40 99.32 80.40 99.32	93.28 93.31	100.00 97.61 97.46	11.46 101.11
63.76 69.09 88.4F 97.06 91.27 82.07 102.69 154.03 102.69 102.69 102.69 111.47 80.88 82.32 90.24 53.23 46.18 49.99 53.94 66.34 67.84 96.13 89.72 83.31 83.31 91.32 90.19 54.10 57.74 68.01 61.56 70.52 85.87 	93.28 93.31	97.61	
5 102.69 154.03 102.69 102.69 102.69 111.47 80.88 82.32 90.24 84.18 49.99 53.94 66.34 67.84 96.13 89.72 83.31 83.31 91.32 90.19 96.13 89.72 83.31 83.31 91.32 90.19 96.13 87.20 96.28 91.70 80.59 90.37 96.67 100.14 99.64 97.33 87.55 100.55 88.86 84.37 76.52 74.64 74.64 126.56 84.37 76.52 79.32 80.40 99.32	87.94 99.00	85.59	
53.23 46.18 49.99 53.94 66.34 67.84 96.13 96.13 89.72 83.31 83.31 91.32 90.19 54.10 57.74 68.01 61.56 70.52 85.87 96.28 96.28 91.70 80.59 90.37 96.67 100.14 99.64 97.33 87.55 100.55 88.86 84.37 76.52 74.64 74.64 126.56 84.37 76.52 79.32 80.40 99.32 80.40 99.32	107.93 102.69	100.37	
53.23 46.18 49.99 53.94 66.34 67.84 95.87 84.86 13.31 83.31 91.32 90.19 100.00 54.10 57.74 68.01 61.56 70.52 85.87 100.14 99.64 97.33 87.55 100.55 88.86 84.37 76.52 74.64 126.56 84.37 76.52 74.64 126.56 84.37 76.52 74.64 126.56 84.37 76.52 74.64 126.56 84.37 76.52 74.64 126.56 84.37 76.52 74.64 126.56	91.21 91.36	86.91	
53.23 46.18 49.99 53.94 66.34 67.84 95.87 96.13 89.72 83.31 83.31 91.32 90.19 96.13 89.72 83.31 83.31 91.32 90.19 96.19 96.28 91.76 80.59 90.37 96.67 97.20 96.28 91.76 80.59 90.37 96.67 98.37 76.52 74.64 74.64 126.56 98.36 98.37 76.52 79.32 80.40 99.32			
53.23 46.18 49.99 53.94 66.34 67.84 96.13 89.13 91.32 90.19 96.13 89.31 91.32 90.19 96.13 89.21 87.26 87.20 96.28 91.70 80.59 90.37 96.67 96.28 97.33 87.55 100.55 88.86 84.37 76.52 79.32 80.40 99.32 84.37 76.52 79.32 80.40 99.32 90.47 97.33	95.87 96.90	100.00	
96.13 89.72 83.31 83.31 91.32 90.19  54.10 57.74 68.01 61.56 70.52 85.87	72.00 70.23	96.63	
54.10     57.74     68.01     61.56     70.52     85.87	91.50 91.19	97.50	
54.10 57.74 68.01 61.56 70.52 85.87  107.58 110.90  87.20 96.28 91.70 80.59 90.37 96.67  100.14 99.64 97.33 87.55 100.55 88.86  84.37 76.52 74.64 74.64 126.56  84.37 76.52 79.32 80.40 99.32	100.00 100.001	100.00	
87,20 96,28 91,70 80,59 90,37 96,67 100,14 99,64 97,33 87,55 100,55 88,86	97.76 103.49	90.29	
100.14 99.64 97.33 87.55 100.55 88.86 88.37 76.52 74.64 74.64 126.56 88.37 76.52 79.32 80.40 99.32	106.32 102.95	100.00 91.84 67.32	37,32 84.20
100.14 99.64 97.33 87.55 100.55 98.86 84.37 76.52 74.64 74.64 126.56 88.37 76.52 79.32 80.40 99.32	95.07 100.28	100.34	
100.14 99.64 97.33 87.55 100.55 88.86 84.37 76.52 74.64 74.64 126.56 84.37 76.52 79.32 80.40 99.32			
84.37 76.52 74.64 74.64 126.56	92,39 102,14	102.93	
84.37 76.52 79.32 80.40 99.32	126.56 83.07	92.26	
75 111 21 10 21 10 21 10 10 10 10 10 10 10 10 10 10 10 10 10	100.05 81.79	88,24	
10.411 01.15 01.18	1133 99.51	100.00 105.94 101.0	10,01 106.39
81.07	81.07 99.83	95.17	
94.46	98.18 99.15		

	 	-			000																					-7	4.
1972-73	167.35	151.57	158.73	189.05		2	116.75		125.7	57.	è	213.7	ai		90	161.99	2	19	'n	65	c i	(2)	(Q)	07	m	31 1	
1571-72	S	2.0		153.19		1.0	5.1	17	111.15	1.1	36.2	261.75	51.5		5.65	148.93	12.6	31.8	16.7	35.6	9.09	3.69	50.5	35.4	12.3	()	0
1:76-71	•		113.96			156.58	102.94	88,44	101.14	176,99	16.8	134.07	42.3		17	142.51	22	0.	7.1	90	37.	57.	13	88	2	1.5	26.
1969-70	co	1	109.78	4		0	101.78	.5	95.95	.0.	-	117.29	4	2 12 12 12 12 12 12 12 12 12 12 12 12 12	35.3	142.51	21.8	31.0	52.3	9.06	32.2	1.1.	89.9	88.2	12.3	42.7	25.3
1968-69	•	•	115.78	•		150.97	102.04		104.48	181.52	09.2	114.13	29.7		30	130.55	20	31	55	91	.37	130.6	2.7	9.0	111.37	3.7	6.
1967-08			108.74						108.01		•	104.24			en	125.96									108.41		
1966-67		140.05	130.33			141.12	101.89	•	114.12	•		103.90				113.88		5	ci	-	0	9	6.	9	1.	9	ó
1965-66	•	•	118.86	•		0	8	1	103.86	1-	8	109.93	2.7			99.34											
1964-65	127.12	112.70	101.12	113.10		•	•	•	98.98	97.90	ထ	99.17	-		51	89.88	.93	6.5	90	15	53	95	98	71		33	51
1										tot a	 -				******												

American security of the particular security of

3 1978-79 1979-83 1980-81 1981-82 1982-83 1	290.14 313.33 406.67 474.78 409.94 292.92 314.44 314.44 314.44 312.63 304.57 304.53 304.53 304.53 285.96 405.05 424.44 477.68 551.08 597.40	572.14 645.36 740.78 775.84 776.15 296.67 311.53 318.15 347.39 403.69 153.59 184.60 212.33 229.51 253.98 281.17 366.84 398.10 410.01 433.06 870.59 1252.88 1314.11 1167.37 1372.93	395.01 .67.34 523.01 562.78 459.58 281.92 434.54 456.89 454.21 447.33 337.73 331.57 416.02 427.09 544.17	334, 26 498, 94, 566, 89 587, 72 665, 60 544, 41 645, 54, 706, 59 717, 68 716, 58 715, 58 213, 29 272, 60 317, 39 305, 07 358, 20 175, 96 196, 98 246, 21 251, 93 228, 18 394, 43 524, 21 617, 80 581, 66 764, 16 127, 87 138, 13 146, 12 152, 97 162, 50 412, 81 618, 52 844, 39 879, 67 970, 49 1032, 70 1325, 92 1605, 87 1626, 79 1726, 47 576, 10 680, 70 700, 89 590, 69 482, 78 589, 38 693, 55 730, 73 616, 16 593, 87 364, 05 423, 46 453, 26 476, 25 390, 56 282, 46 283, 88 461, 34, 499, 13 472, 83
51-77-191	250.4 250.3 253.1 283.1 173.2 173.2 368.86 393.10	397.9 463.7 304.8 317.7 180.8 172.4 263.8 314.0 800.0 355.9	276.0 286.4 254.7 253.6 355.29 369.94	257.6 254.5 428.9 446.2 214.0 222.1 145.8 141.2 475.9 484.4 110.6 112.4 337.15 379.80 613.7 613.0 552.50 577.2 457.0 473.4 286.8 296.7 279.2 279.2 350.23 347.81
1974-75 1975-76	226.5. 250.5 247.04 283.1 262.14 173.9 302.38 334.38	323.51 365.0 198.20 270.7 171.1 164.7 210.8 236.4 445.8 747.7	249.14 741.5 253.62 150.6 284.85 3.1.84	193.15 13.6 162.56 13.9 213.0 1.0.0 151.99 168.0 319.66 432.4 87.96 104.0 314.24 370.09 464.60 586.0 637.60 516.9 431.7 427.8 242.62 268.0 261.9 284.4
1973-74	184.33 190.77 148.59 242.65	201.60 157.95 123.0 141.0	196.63 213.7 109.0	193 15 162 56 171 62 106 56 193 20 77 17 236 04 331 05 30 60 101 5 162 0

Table-A 3.13

# 1:DICES OF AGRICULTURAL WHOLFSALE MYSSTIC PRICES

	1951/52	1952/53	1953/54	1954/55	1955/56	1956/57	1957/58	1938/39	1959/60	1950/61	1961/62	1962/63	1963/64	1964/65
FOOD CROPS			-SRST						4 11					
Rice Theat Maize Marley Forghum Pulses Potatoes Inions	67.74 76.92 65.12 64.52 57.14 72.55 60.00 61.11	74.19 115,38 113,95 112,90 119.05 78.43 67.50 44.44	79.03 89.74 69.77 70.97 76.19 80.39 67.50 74.07	70.97 69.23 51.16 51.61 52.38 50.98 75.00 62.96	74.19 79.49 67.44 67.74 71.43 50.98 65.00 59.26	93.55 94.87 86.05 77.42 54.76 86.27 77.50 64.81	109.68 94.87 79.07 83.87 76.19 115.69 72.50 61.11	101.61 94.87 90.70 96.77 88.10 98.04 90.00 46.30	100.00 100.00 100.00 100.00 100.00 100.00 100.00 100.00	112.90 115.38 86.05 103.23 109.52 111.76 115.00 73.04	103,23 107,69 83.72 93.55 92.86 131.37 77.50 77.78	111.30 102.56 \$1.40 \$3.87 78.57 123.53 90.00	125.81 112.82 83.72 166.45 92.86 110.61 107.50 68,52	96.2 122.5 96.3 117.4 92.0 124.3 95.4 40.3
B. CASH CROPS Oilseeds	102.99	120.90	92.54	58.21	92.54	954	129.85	130.90	100.00	100.00	102.98	10:.90	1193	
Jute Sugarcane	184.26	90.60	92.50	87.04	99.54	98.15	96.76	86.11	100.00	106.80	100.00	98.76	97.22	136.6
Tobacco C. LIVESTOCK C. LIVESTOCK	61.39	63,70	87.13	87.50 93.40	82.50 71.62	87.46	117.50 38.45	107.50 71.62	100.00	125.00	147.50 97.03	130.00	140.00	140.1
dilk dutteroil deat ool ides & Skins	83.33 70.20 87.95 55.49 36.76	81.67 69.11 90.60 67.45 46.53	86.67 73.01 90.66 77.65 47.52	81.67 79.10 92.46 80.87 62.71	75.00 70.67 94.21 166.29 66.67	78.33 74.83 88.57 99.43 72.61	81,67 .94.23 .92.39 89.77 74.26	86.67 92.36 98.31 84.47 78.22	100.00 100.00 100.00 100.00	106.67 99.68 104.40 96.78 86.47	105.00 95.94 111.69 98.10 76.57	106.67 93.60 110.67 96.02 90.86	105.00 92.51 117.42 109.47 76.90	110.30 102.9 119.1 100.5

Contd....)

Table-A 3.13 Continued

19621	19661 67	19671	1968/	1959/ 70	1970/	1971/ 72	1972/ 73	1973/ 74	1974/ 75	1975/ 76	1976/	1977/ 78	1978/	1979/	1980/	1981/	1982/	1983/
112.66 111.66 90.55 90.48 91.92	168.48 136.36 138.41 140.66	149.10 118.30 151.99 119.77	127.78 97.02 115.47 102.66	119.61 128.96 100.65 135.11 92.48	132.48 104.72 140.41 83.54	126.72 154.19 99.76	185.56 153.62 137.88 176.56 131.85	238.35 190.55 152.21 242.90 128.01	260.10 294.20 280.30 240.10 235.30	295.3 280.8 264.7 282.0 225.5	329.10 282.6 234.4 248.7 241.2	412.41 278.12 458.16 214.83	362.34 343.13 426.29 229.06	333.81 530.64 298.26	450.07 379.53 335.77 589.22 346.44	502.57 443.02 463.24 540.40 386.35	461.61 478.76 539.95 420.67	527.26 486.79 454.65 627.96 363.33
131.05 97.78 67.07	135.19	139.01 107.28 53.68	147.29 103.23 55.13	198.83 130.97 56.20	194.56 138.23 86.15	188.73 113.11 64.86	243.01 100.55 68.29	364.19 283.75 228.05	243.2 229.9 126.7	355.6 136.1 204.1	470.0 285.9 292.0	399.33	501.31 401.27 206.12	415.37	529.12 416.02 217.35	517.84 425.35 218.94	430.16	540.29 435.02 245.24
159.61 107.39		144.36- 95.77		184.45 114.46		174.24	185.76 184.51	267.08 264.67	353.2 235.3	364.1 292.8	399.4 397.7			321.99 456.07	389.10 426.60	402.51 513.43	402.51 533.38	
140.15		81.14 111.97	81.14 105.96	81.14 113.57	83.74 113.75	34.15 138.48	101.87 157.91	114.21 147.42	123.1 223.0	158.5 221.1	171.0 238.4		152.47 392.49	439.63	235.29 455.54	245.25 441.72		258.88 444.79
114.24 105.07 125.70 89.23 89.36	112.98 133.69	116.96 158.13 59.78	134.94 118.36 168.82 67.66 146.58	140.08 133.38 170.06 79.81 138.29	150.65 148.13 186.65 77.24 135.97	160.00 147.42 196.79 68.04 164.41	189.61 158.18 198.93 97.93 214.86	264.61 253.03 272.72 147.51 225.37	327.9 272.1 366.5 115.2 247.6	382.0 310.7 417.1 141.8 271.1	494.4 368.6 490.4 182.4 321.9	510.45 510.05 127.86	410.65 521.86 212.80	480.41 450.95 579.05 241.46 383.67	496.68 678.93	565.71 620.32 723.02 175.27 299.39	780.33 159.87	698.07 639.45 825.98 173.04 429.09

Table-A 3.14

#### 1959/60 WEIGHTS FOR PRODUCTION, MARKETING, AND PURCHASES OF AGRICULTURAL COMMO-DITIES

		GVO	Marketings	Purchases by Non-Agriculture Sector net avai- lability
Rice	e			
	01 F and M	5:4	5.8	4.8
	02 Coarse	2.7	2.9	2.5
03	Wheat	23.9	13.7	32.3
04	Maize	2.6	0.9	1.2
05	Barley .	0.6	0.2	0.3
06	Sorghum	1.2	1.1	1.4
07	Pulses (lentils)	4.5	4.2	5.7
80	Potatoes	0.6	0.5	0.7
09	Onions	0.7	0.6	0.8
10	Oilseeds	5.6	3.3	3.4
11	Cotton .	8.6	16.3	4.4
12	Jute	0.0	0.0	0.0
13	Sugarcane	8.1	13.5	1.7
14	Tobacco	2.3	4.0	3.7
15	Milk	18.9	14.3	21.3
16	Butteroil	9.4	7.2	10.7
17	Meat	3.3	2.5	3.3
18	Woo1	0.7	1.5	0.2
9	. Hides and skins	0.9	1.5	1.6
	The second second			

Table-A 3.15
1959/60 WEIGHTS FOR PRODUCTION AND PURCHASES OF NON-AGRICULTURAL GOODS

		Value		Ne t.		rchases b
,	6.18 Th	added	GVO	availa- bility	Alter- native	Alter- native 2
0	Sugar manufacturing	2.8	3.6	2.7	3.5	2.2
1	Edible oils	1.9	5.6	4.5	4.3	1.7
2	Tea manufacturing	.5	1.6	4.3	7.9	6.1
9	Food mfg.,nec.	1.2	1.4	1.0	1.8	1.4
0	Beverages	.6	.5	.5	.9	.7
0	Tobacco manufacturin	g 5.6	4.6	3.1	5.7	4.5
0	Cotton textiles	42.2	41.2	21.2	38.9	30.3
4	Silk and artsilk textiles	3.2	2.6	2.9	5.3	4.1
0	Footwear	2.3	2.2	1.4	2.6	2.0
0	Wood and furniture	5	.4	.3		.4
0	Printing and publi- shing	2.4	1.6	1.4	2.5	2.0
0	Soap, perfume, etc.	2.4	2.7	2.0	3.6	2.8
1	Matches	.2	.1	.7	1.4	1.1
0	Miscellaneous mfg.	1.6	1.1	1.4	2.6	2.0
3	Jute textiles		.0	1.5	2.3	2.7
C	Paper manufacturing	1.4	1.3	2.7	.0	9.8
3	Leather manufacturin	g 1.1.	2.1	.0	.1	.1
С	Rubber and rubber pr	o3	.3	.9	.4	1.0
1	Fertilizer	.3	.4	.8	2.4	2.9
9	Chemicals	2.2	1.7	2.6	.0	.0
3	Medicines	2.6	2.1	3.2	5.8	4.5
Э.	Petrol and coal	4.3	3.8	7.4	2.3	5.4
3	Non-metallic mineral	s 6.0	4.7	3.9	.6	1.4
)	Basic metals	2.9	3.0	6.4	1.0	2.4
)	Metal products	3.4	3.5	3.2	.5	1.2
)	Machinery excluding elec.	2.4	2.1	8.8	1.4	3.2
)	Electric machinery	2.3	2.2	4.8	.7	1.7
)	Transport equipment	3.4	3.6	6.4	1.0	2.4

TABLE-A 3.16
WEIGHTED DOMESTIC PRICE INDICES FOR NON-AGRICULTURAL COMMODITIES

Vozna	Value	GVO	Net Avail-		Purchases	hy agricul	tu
Years	Added	GVO	ability	A7t. 1	Alt. 2	Alt. 3	A
1951-52	86.91	85.51	87.54	83.61	85.29	85.03	
1952-53	92.38	90.55	91.36	89.97	91.26	90.79	
1953-54	100.97	99.76	,94.80	99.24	98.15	98.34	
1954-55	87.14	87.31	85 80	88.16	86.31	87.67	
1955-56	86.89	86.43	87.13	87.31	88.22	87.08	
1956-57	92.16	91.98	95.07	91.49	93.03	92.80	
1957-58	98.83	99.54	100.47	97.71	97.81	98.52	
1958-59 .	95.51	95.73	96.05	95.19	95.26	95.00	
1959-60	100.00	100.00	100.00	100.00	100.00	100.00	ì
1960-61	98.31	98.15	98.07	98.12	98.34	98.03	
1961-62	99.74	99.50	99.39	98.44	100.08	. 98.50	i
1962-63	101.29	101.05	.103,96	100.09	102.44	101.22	1
1963-64	102.22	101.73	105.38	100.15	103.18	101.47	1
1964-65	102.81	103.02	104.28	101.56	103.66	101.98	1
1965-66	107.08	107.66	109.88	105.54	107.69	106.30	1
1966-67	112.75	113.62	116.14	110.23	112.50	111.35	1
1967-68	115,86	116.53	119.31	110.91	113.94	112.99	ı
1968-69	118.87	119.18	124.36	113.10	117.67	116.21	1
1969-70	120.85	121.25	126.63	113.09	118.78	116.67	1
1970-71	125.50	125.83	133.27	116.49	123.63	120.96	1
1971-72	132.54	132.47	140.13	124.36	132.75	127.98	1
1972-73	148.33	147.47	154.14	134.91	145.08	138.11	,
1973-74	188.86	187.29	194.23	166.07	176.39	172.02	1
1974-75	258.08	254.37	275.32	217.11	231.01	232.79	i
1975-76	315,26	314.34	314.16	263.09	281.00	273.79	2
1976-77	336.10	334.55	327.01	280.64	289.59	290.92	25
1977-78	350.97	318.51	339.63	293.78	301.84	303.91	25
1978-79	385,94	376.19	377,24	314.94	334.23	319,48	3.
1979-80	451.95	435.31	447.56	354.60	402.55	362.35	5
1980-81	505.83	483.75	504.84	385.89	443.60	395.69	
1981-82	523.19	500.55	515.67	417.82	461.94	413,11	
1982-83	557.07	531.82	530.23	433.14	483.08	436.75	
1983-84	600.56	572 00	563.57	466.74	512.82	470.59	

Table-A 3.17
WEIGHTED DOMESTIC PRICE INDICES FOR AGRICULTURE

YEARS	PRODUCT		Net Availability and	
		Gross Value of Output	Purchase by Manufac- turing Sector	
		output	, carrag sector	-
1951-52	96.05	87.01	81.51	
195 2-53	89.63	92.02	92.20	
1953-54	85.58	85.17	84.81	
954-55	77.16	75.23	74.22	
955-56	82.72	79.38	77.02	
.956-57	92.39	89.65	87.13	
957-58	100.35	96.91	94.37	
958-59	95.29	94.33	92.67	
959-60	100.00	100.00	100,00	
960-61	108.45	108.21	107,91	
961-62	108.91	107.19	105.15	
962-63	108.22	1.05.78	1.03.03	
963-64	110.44	109.12	107.57	
964-65	108.96	116.03	113.97	
965-66	109.29	115.86	113,56	
966-67	120.13	136.44	1.39,41	
967-68	110.30	128.14	131.82	
968-69	111.22	123.93	127.82	
969-70	117.01	, 130.90	134.94	
970-71	123.88	136.20	140.40	
971-72	132.16	147.05	151.05	
972-73	155.79	168.05	171.93	
973-74	210.04	227.70	232.33	
974-75	237.28	273.94	285.88	
975-76	278.74	299.11	311.78	
976-77	317.50	345.42	353.01	
977-78	358.87	401.15	419.98	
978-79	356.09	384.97	397.16	
979-80	367.44	400.33	416.67	
980-81	398.30	431.64	447.98	
981-82	430.90	474.51	488,65	
982-83	454.44	502.43	519.36	
983-84	507.57	547.88	565.04	

## ALTERNATIVELY WEIGHTED DOMESTIC PRICE INDICES FOR MANUFACTURING BY MAJOR GROUPINGS

(Continued)

		VALUE A	DDED		GRO	SS VALUE	OF OUTPUT			NE	T AVAILAR	ILITY
YEARS	Consum- tion	Inter- mediate	Invest-	TOTAL	Consum- ption	Inter- mediate	Invest- ment	TOTAL	Consum- ption	Inter- mediate	Invest- ment	TOTAL
1951-52	86.20	73.59	105.01	86.91	85.51	71.39	105.41	85.51	82.57	80.05	111.96	87.54
1952-53	93.36	78.32	95.59	92.38	91.66	74.42	94.58	90.55	88.76	86.54	99.08	91.36
1953-54	106.09	78.97	88.71	100.97	104.92	75.19	87.45	99.76	99.66	85.36	89.09	94.80
1954-55	89.87	72.84	83.69	87.14	90.31	70.91	83.05	87.31	89.15	76.89	84.09	85.80
1955-56	86.04	88.79	89.24	86.89	86.11	86.65	87.92	86.43	85.92	91.35	86.52	87.13
1956-57	90.07	94.98	97.41	92.16	90.31	92.34	97.83	91.98	89.38	97.36	101.94	95.07
1957-58	98.47	97.41	100.86	98.83	99.82	94.80	101.42	99.54	97.65	99.36	105.05	100.47
1958-59	95.20	98.10	94.97	95.51	96.21	95.21	94.30	95.73	95.25	100.24	94.96	96.05
1959-60	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00
1960-61	99.09	95.72	97.27	98.31	99.02	95.49	96.63	98.15	98.57	97.10	97.88	98.07
1961-62	99.93	96.45	107.08	99.74	99.77	96.56	100.30	99.50	99.10	98.03	100.52	99.39
1962-63	99.34	98.25	109.56	101.29	99.29	97.58	109.55	101.05	100.51	100.01	110.91	103.96
1963-64	99.45	102.47	111.21	102.22	99.26	101.31	110.94	101.73	100.26	104.92	112.86	105.38
1964-65	97.15	106.13	106.45	102.81	99.22	103.62	106.77	103.02	101.46	102.16	109.54	104.28
1965-66	99.64	113.82	114.11	107.08	102.30	111.03	114.92	107.66	105.94	108.23	116.57	109.88
1966-67	103.56	124.01	122.74	112.75	106.85	121.39	123.36	113.62	111.15	116.90	123.08	116.14
1967-68	104.95	124.57	133.04	115.86	107.71	123.52	134.60	116.53	110.62	117.39	133.15	119.31
1968-69	105.14	130.27	143.11	118.87	107.62	129.12	144.44	119.18	111.42	122.88	143.92	124.30
1969-70	104.56	133.20	152.47	120.85	107.32	133.01	153.50	121.25	110.99	125,20	149.83	126.83
1970-71	107.21	140.09	162.97	125.50	109.71	139.03	164.37	125.83	113.93	131.99	161.61	133.23
1971-72	113.29	149.32	167.74	132.54	116.19	147.99	168.59	132.47	122.68	142.20	163.55	140.1
1972-73	124.56	159.75	199.36	148.33	127.81	158.50	196.61	147.47	135.37	153.37	180.70	154.1
1973-74	159.58	182.43	265.40	188.86	162.99	182.15	262.61	187.29	166.83	181.60	241.05	194.23
1974-75	209.55	240.43	406.38	258.08	211.97	233.71	405.68	254.37	215.90	236.62	384.70	275.33
1975-76	265.73	322.62	439.55	315.26	266.76	327.50	432,18	314.34	263.43	298.15	396.69	314.1
1976-77	293.32	315.11	460.10	336.10	292.36	321.28	452.38	334.55	284.96	282.49	415.15	327.0
1977-78	310.10	333.28	466.88	350.97	307.43	339.46	459.75	348.51	298.46	300.67	423.74	339.6
1978-79	310.23	353.60	617.29	385.94	310.39	369.26	591.08	376.19	313.95	319.01	502.85	377.2
1979-80	338.82	487.84	749.02	451.95	337.26	499.57	712.25	435.31	343.38	453.29	591.86	447.50
1980-81	361.51	604.35	861.06	505.83	358.69	611.15	813.32	483.75	366.46	574.84	651.38	504.8
1981-82	388.74	615.75	846.91	523.19	385.42	623.27	795.88	500.55	390.39	588.35	651.81	515.6
1982-83	424.64	686.13	850.22	557.07	416.72	683.94	791.24	531.82	414.22	656.56	622.84	530.2
1983-84	464.93	723.24	910.13	600.56	456.87	730.20	845.03	572.00	454.78	684.26	650.84	563.57

ALTERNATIVELY WEIGHTED DOMESTIC TRICE INDICES DO

Years    Years   Con-   Sump-   Sump-	Principles of deliver distance	II.TURF	AGRICI	THE THE STREET STREET		110		Y	B		1.1	IASE	PURCE	-	Pilosop But William Sanda		
Table   Con-   Inter-   Inte				-	P M (mail of the little business	IVE 3	LTERNATI	A	A COMPANY OF THE PROPERTY OF T	IVE 2	LTERNAT	A		15	RNATIV	ALTE	
1952-53 89.43 93.54 99.20 89.97 89.88 94.52 98.88 91.26 89.56 93.51 96.77 90.79 90.03 95.24 99.37 1953-54 100.02 89.06 89.14 99.24 100.79 89.76 88.90 98.15 100.61 86.47 87.94 98.34 100.89 93.12 91.07 1954-55 88.61 80.04 84.14 88.16 88.06 79.81 83.97 86.31 88.30 78.87 83.60 87.67 87.77 79.99 84.68 1955-56 85.82 101.74 86.51 87.31 85.64 96.01 86.35 88.22 85.63 99.69 84.60 87.08 87.67 79.99 84.68 1955-57 89.06 104.40 102.23 91.49 88.29 88.80 102.06 93.03 89.01 102.39 103.15 92.80 88.62 98.49 100.74 1957-58 96.56 102.57 105.35 97.71 95.89 98.56 105.22 97.81 96.35 101.09 106.22 98.52 95.27 98.41 103.90 1958-59 94.42 101.03 94.92 95.19 90.55 98.69 94.84 95.26 93.90 103.03 19.00 100.00	Total	Invest	Inter- med-	sump-	Total	vest-	med-	sump-	Total	vest-	med-	sump-	Total	vest-	med-	sump-	YEARS
1952-53 89.43 93.54 99.20 89.97 89.81 94.52 98.88 91.26 89.56 93.51 98.77 90.79 90.03 95.24 99.37 1953-55 100.02 89.00 89.14 99.24 100.79 89.76 88.90 98.15 100.61 86.47 87.94 89.34 100.89 93.12 91.07 1954-55 88.61 80.04 84.14 88.16 88.06 79.81 83.97 86.31 88.30 78.87 83.60 87.67 87.77 79.99 84.68 1955-56 85.82 101.74 86.51 87.31 85.64 96.01 86.35 88.22 85.83 99.69 84.60 87.08 85.53 95.65 88.62 98.49 100.74 1955-58 96.56 102.57 105.35 97.71 95.89 98.56 105.22 97.81 96.35 101.09 103.15 92.80 88.62 98.49 100.74 1955-59 86.99 44.2 101.03 94.92 95.19 90.55 98.69 94.84 95.25 97.00 100.00 1	85.0	109.32	88.61	82.64	85.03	115,12	87.52	82.53	85.29	112.14	88.04	82.66	83.61	112.10	87.25	82.76	1951-52
1953-54 100.02 89.00 89.14 99.24 100.79 89.76 88.90 98.15 100.61 86.47 87.94 98.34 100.89 93.12 91.07 1954-55 88.61 80.04 84.14 88.16 88.06 79.81 83.97 86.31 88.30 78.87 83.60 87.67 87.77 79.99 84.68 1955-56 85.82 101.74 86.51 87.31 85.64 96.01 86.35 88.22 85.63 99.69 84.60 87.08 85.53 95.65 88.64 1955-57 89.06 104.40 102.23 91.49 88.82 98.80 102.06 93.03 89.01 102.39 103.15 92.80 88.62 98.49 100.74 1955-58 99.65 102.57 105.35 97.71 96.55 98.69 94.64 101.03 94.92 95.19 90.55 98.69 94.84 100.00	91.4							89.56		98.88	94.52	89.81		99.20	93.54	89.43	1952-53
1955-55 88.61 80.04 84.14 88.16 88.06 79.81 83.97 86.31 88.30 78.87 83.60 87.67 87.77 79.99 84.68 1955-56 58.82 101.74 86.51 87.31 85.64 96.01 86.35 88.22 85.63 99.69 84.60 87.08 85.53 95.65 88.64 1955-58 95.65 102.57 105.35 97.71 95.89 98.56 105.22 97.81 96.35 101.09 106.22 98.52 95.27 98.41 103.90 1958-59 94.42 101.03 49.92 95.19 95.55 98.69 94.84 95.26 93.90 100.00 10	98.4								98.15		89.76	100.79	99.24	89.14	89.00	100.02	1953-54
1955-56	85.8											88.06	88.16	84.14	30.04	88.61	1954-55
1956-57 89.06 104.40 102.23 91.49 88.82 98.86 102.06 93.03 89.01 102.39 103.15 92.80 88.62 98.49 100.74 1957-58 96.56 102.57 105.35 97.71 95.89 98.56 105.22 97.81 96.35 101.09 106.22 98.52 95.27 98.41 103.90 1958-59 94.42 101.03 94.92 95.19 90.55 98.69 94.84 95.26 93.90 100.13 94.31 95.00 93.16 98.66 95.77 195.96 100.00 10	88.5											85.64		86.51	101.74	85.82	1955-56
1957-58 96.56 102.57 105.35 97.71 95.89 98.56 105.22 97.81 96.35 101.09 106.22 98.52 93.27 98.41 103.90 1958-59 94.42 101.03 94.92 95.19 90.55 98.69 94.84 95.26 105.00 100.00 10	92.7														104.40	89.06	1956-57
1958-59 94.42 101.03 94.92 95.19 90.55 98.69 94.84 95.26 93.90 100.13 94.31 95.00 93.10 98.66 95.77 1959-60 100.00	97.0												97.71				1957-58
1959-60 100.00 1	95.2													94.92	101.03	94.42	1958-59
1960-61 98.48 95.69 97.90 98.12 98.19 98.52 97.99 98.34 97.92 96.70 97.44 98.03 98.13 98.77 98.49 1961-62 99.04 93.03 100.50 98.44 98.57 103.42 101.77 100.08 98.60 96.93 99.54 98.50 98.48 103.91 101.73 1962-63 100.59 91.28 111.00 100.03 90.17 103.47 111.02 102.44 100.14 95.85 110.68 101.22 100.18 104.09 110.97 1963-64 100.14 94.21 112.88 100.15 99.57 106.87 112.83 103.18 99.09 98.66 112.71 101.47 99.44 107.55 113.11 1964-65 122.72 87.48 112.98 101.56 100.55 110.68 7112.83 103.18 99.09 98.66 112.71 101.47 99.44 107.55 113.11 1964-66 116.78 88.09 120.22 105.54 144.65 106.27 117.61 107.69 105.79 100.72 113.64 106.30 104.76 111.07 112.44 1966-67 123.45 91.53 126.87 110.23 109.58 110.02 124.16 112.50 111.03 104.34 119.71 111.35 109.83 114.98 118.97 1968-69 123.27 95.76 137.11 110.91 108.87 112.22 134.44 113.94 110.44 108.46 129.85 112.99 109.16 117.21 128.17 1969-70 123.78 101.12 154.34 113.09 109.52 121.28 151.36 118.78 110.86 115.71 145.25 116.67 109.58 127.02 145.23 1970-71 126.66 104.11 166.50 116.49 111.38 127.63 163.47 123.63 113.53 120.22 157.57 120.96 112.04 133.98 157.12 1971-72 137.55 104.11 168.48 124.36 120.83 139.62 165.38 132.75 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1977-73 153.52 93.67 186.11 134.91 134.08 146.74 182.58 145.08 173.9 174.93 138.11 136.29 156.02 187.69 1975-76 297.64 168.82 409.08 263.99 263.72 255.64 400.80 288.100 127.65 232.79 215.89 200.98 235.16 1975-77 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 216.64 158.80 197.27 37.99 265.49 271.89 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 389.98 204.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 375.27 470.59 472.35 570.24 758.86 1988-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	100.0																1959-60
1961-62 99.04 93.03 100.50 98.44 98.57 103.42 101.77 100.08 98.60 96.93 99.54 98.50 98.48 103.91 101.73 1962-63 100.59 91.28 111.00 100.09 100.17 103.47 111.02 102.44 100.14 95.85 110.68 101.22 100.18 104.09 110.97 1963-64 100.14 94.21 112.88 100.15 99.57 106.87 112.83 103.18 99.09 98.66 112.71 101.47 99.44 107.55 113.11 1964-65 122.72 87.48 112.98 101.56 100.45 103.67 110.45 103.66 101.56 99.47 106.67 101.98 100.72 08.27 105.82 1965-66 116.78 88.09 120.22 105.54 14.65 105.27 117.61 107.69 105.79 100.72 113.64 106.30 104.76 111.07 112.44 1066-67 123.45 91.53 126.87 110.23 109.58 110.02 124.16 112.50 111.03 104.34 119.71 111.35 109.83 114.98 118.97 1967-68 123.27 95.76 137.11 110.91 108.37 112.22 134.44 113.94 110.44 108.46 129.85 112.99 109.1e 117.21 128.17 1968-90 123.38 10.12 154.34 113.09 109.52 121.28 151.36 118.78 110.86 115.71 145.25 116.67 109.58 127.02 145.23 1970-71 126.66 104.11 166.50 116.49 111.38 127.63 163.47 123.63 113.53 120.22 157.57 120.96 112.04 133.98 157.12 1971-72 137.25 104.11 168.48 124.36 120.83 139.62 165.38 132.75 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1972-73 153.52 93.67 186.11 134.91 134.08 146.74 182.58 145.08 137.26 117.39 174.35 138.11 136.29 156.02 187.69 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 220.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 299.63 238.64 423.84 1978-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.00 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.88 378.75 550.2 97.59 470.59 472.35 570.24 758.86 1988-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	98.4					97 44	96 70	97 92									
1962-63 100.59 91.28 111.00 100.03 100.17 103.47 111.02.102.44 100.14 95.85 110.68 101.22 100.18 104.09 110.97 1963-64 100.14 94.21 112.88 100.15 99.57 106.87 112.83 103.18 99.09 98.66 112.71 101.47 99.44 107.55 113.11 1964-65 122.72 87.48 112.98 101.56 100.45 103.66 110.56 99.47 106.67 101.98 100.72 108.27 105.82 1965-66 116.78 88.09 120.22 105.54 1.4.65 106.27 117.61 107.69 105.79 100.72 113.64 106.30 104.76 111.07 112.44 1966-67 123.45 91.53 126.87 110.23 109.58 110.02 124.16 112.50 111.03 104.34 119.71 111.35 109.83 114.98 118.97 1968-69 123.30 100.16 148.17 113.10 110.97 118.30 145.30 117.67 111.53 114.08 140.47 116.21 110.44 123.77 138.59 1969-70 123.78 101.12 154.34 113.09 109.92 121.28 151.36 118.78 110.86 115.71 145.25 116.67 109.58 127.02 145.23 1970-71 126.66 104.11 166.50 116.49 111.38 127.63 163.47 123.63 113.53 120.22 157.57 120.96 112.04 133.98 127.12 137.25 104.11 168.48 124.36 120.83 139.62 165.38 132.55 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1972-73 153.52 93.67 186.11 134.91 134.08 146.74 182.58 145.08 137.26 117.39 174.35 138.11 136.29 156.02 187.69 1973-74 190.01 112.92 248.58 166.07 165.38 164.44 233.68 176.39 1875-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.99 265.49 271.38 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.99 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 220.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.63 238.64 423.84 1978-89 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1988-84 48.87 315.26 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 470.59 472.35 570.24 758.86 1988-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	100.5																
1963-64 100.14 94.21 112.88 100.15 99.57 106.87 112.83 103.18 99.09 98.66 112.71 101.47 99.44 107.55 113.11 1964-65 122.72 97.48 112.98 101.56 '00.45 103.67 110.45 103.66 101.56 99.47 106.67 101.98 100.72 108.27 105.82 106.66 116.78 88.09 120.22 105.54 104.65 105.27 117.61 107.69 105.79 100.72 113.64 106.30 104.76 111.07 112.44 110.66-67 123.45 91.53 126.87 110.23 109.38 110.02 124.16 112.50 111.03 104.34 119.71 111.15 109.83 114.98 118.97 1968-69 123.30 100.16 148.17 113.10 110.07 118.30 145.30 117.67 111.53 114.08 140.47 116.21 110.44 123.77 138.59 1969-70 123.78 101.12 154.34 113.09 109.32 121.28 151.36 118.78 110.86 115.71 145.25 116.67 109.58 127.02 145.23 1970-71 126.66 104.11 166.50 116.49 111.38 127.63 163.47 123.63 113.53 120.22 157.57 120.96 112.04 133.98 157.12 1971-72 137.25 104.11 168.48 124.36 120.83 139.62 165.38 132.75 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1973-73 153.52 93.67 186.11 134.91 134.08 164.42 423.68 176.39 167.86 136.75 222.65 172.02 165.90 174.48 240.72 1974-75 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 217.42 166.13 217.65 232.79 215.89 200.98 325.16 1975-76 297.64 168.82 409.08 263.09 263.79 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 389.30 303.91 306.30 246.79 431.24 1977-78 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.86 393.98 303.91 306.30 246.79 431.24 1977-78 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.86 393.98 303.91 306.30 246.79 431.24 1977-78 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.86 393.98 303.91 306.30 246.79 431.24 1977-78 340.88 168.32 437.28 293.78 302.74 234.07 566.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 339.89 399.99 399.80 246.17 611.25 354.60 486.54 604.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 448.87 3	100.5																
1964-65 122.72 87.48 112.98 101.56 100.45 103.67 110.45 103.66 101.56 99.47 106.67 101.98 100.72 188.27 105.82 11965-66 116.78 88.09 120.22 105.54 144.65 106.27 117.61 107.69 105.79 100.72 113.64 106.30 104.76 111.07 112.44 1966-67 123.45 91.53 126.87 110.23 109.38 110.02 124.16 112.50 111.03 104.34 119.71 111.35 109.83 114.98 118.97 11967-68 123.27 95.76 137.11 110.91 108.37 112.22 134.44 113.94 110.44 108.46 129.85 112.99 109.10 117.21 128.17 128.17 12968-69 123.30 100.16 148.17 113.10 110.07 118.30 145.30 117.67 111.53 114.08 140.47 116.21 110.44 123.77 138.59 11970-71 126.66 104.11 166.50 116.49 111.38 127.63 163.47 123.63 113.53 120.22 157.57 120.96 112.04 133.98 157.12 137.25 104.11 168.48 124.36 120.83 139.62 165.38 132.75 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1972-73 153.52 93.67 186.11 134.91 134.08 146.74 182.58 145.08 137.26 117.39 174.35 138.11 126.29 156.02 187.69 11973-74 190.01 112.92 248.58 166.07 165.38 164.44 243.68 176.39 167.86 136.75 222.65 172.02 165.99 174.48 240.72 1974-75 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 217.42 166.13 217.65 232.79 215.89 200.98 235.16 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 340.66 160.20 428.32 220.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.83 238.64 423.84 1977-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1988-84 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1988-84 48.87 316.25 672.71 417.82 396.46 504.35 559.51 482 443.60 377.55 413.16 507.20 436.75 431.90 566.70 718.10 1988-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 448.87 431.75 522.79 470.59 472.35 570.24 758.86																	
1965-66 116.78 116.78 116.78 116.78 116.78 117.61 117.62 117.61 117.62 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 117.61 1	103.2															•	
1966-67 123.45 91.53 126.87 110.23 109.58 110.02 124.16 112.50 111.03 104.34 119.71 111.35 109.83 114.98 118.97 1967-68 123.27 95.76 137.11 110.91 108.37 112.22 134.44 113.94 110.44 108.46 129.85 112.99 109.16 117.21 128.17 1968-69 123.30 100.16 148.17 113.10 110.07 118.30 145.30 117.67 111.53 114.08 140.47 116.21 110.44 123.77 138.59 1969-70 123.78 101.12 154.34 113.09 109.32 121.28 151.36 118.78 110.86 115.71 145.25 116.67 109.58 127.02 145.23 1970-71 126.66 104.11 166.50 116.49 111.38 127.63 163.47 123.63 113.53 120.22 157.57 120.96 112.04 133.98 157.12 1971-72 137.25 104.11 168.48 124.36 120.83 139.62 165.38 132.75 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1972-73 153.52 93.67 186.11 134.91 134.08 146.74 182.58 145.08 137.26 117.39 174.35 138.11 136.29 156.02 187.69 1973-74 190.01 112.92 248.58 166.07 165.38 164.44 24.68 176.39 167.86 136.75 222.65 172.02 165.99 174.48 240.72 1974-75 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 217.42 166.13 217.65 232.79 215.89 200.98 325.16 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.63 238.64 423.84 1977-8. 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.86 393.98 303.91 306.30 246.79 431.24 1978-87 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 346.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 413.10 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	103.4																
1967-68 123.27 95.76 137.11 110.91 108.87 112.22 134.44 113.94 110.44 108.46 129.85 112.99 109.16 117.21 128.17 1968-69 123.30 100.16 148.17 113.10 110.07 118.30 145.30 117.67 111.53 114.08 140.47 116.21 110.44 123.77 138.59 1970-71 126.66 104.11 166.50 116.49 111.38 127.63 163.47 123.63 113.53 120.22 157.57 120.96 112.04 133.98 157.12 1971-72 137.25 104.11 168.48 124.36 120.83 139.62 165.38 132.75 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1972-73 153.52 93.67 186.11 134.91 134.08 146.74 182.58 145.08 137.26 117.39 174.35 138.11 136.29 156.02 187.69 1973-74 190.01 112.92 248.58 166.07 165.38 164.44 243.68 176.39 167.86 136.75 222.65 172.02 165.99 174.48 240.72 1974-75 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 217.42 166.13 217.65 232.79 215.89 200.98 325.16 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.33 238.64 423.84 1977-78. 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.80 393.98 303.91 306.30 246.79 431.24 1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	107.4																
1968-69 123.30 100.16 148.17 113.10 110.07 118.30 145.30 117.67 111.53 114.08 140.47 116.21 110.44 123.77 138.59 1969-70 123.78 101.12 154.34 113.09 109.32 121.28 151.36 118.78 110.86 115.71 145.25 116.67 109.58 127.02 145.23 1970-71 126.66 104.11 166.50 116.49 111.38 127.63 163.47 123.63 113.53 120.22 157.57 120.96 112.04 133.98 157.12 1971-72 137.25 104.11 168.48 124.36 120.83 139.62 165.38 132.75 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1972-73 153.52 93.67 186.11 134.91 134.08 146.74 182.58 145.08 137.26 117.39 174.35 138.11 136.29 156.02 187.69 1973-74 190.01 112.92 248.58 166.07 165.38 164.44 243.68 176.39 167.86 136.75 222.65 172.02 165.90 174.48 240.72 1974-75 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 217.42 166.13 217.65 232.79 215.89 200.98 325.16 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.63 238.64 423.84 1978-78 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.80 393.98 303.91 306.30 246.79 431.24 1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1988-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1988-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	112.1																
1969-70 123.78 101.12 154.34 113.09 109.32 121.28 151.36 118.78 110.86 115.71 145.25 116.67 109.58 127.02 145.23 1970-71 126.66 104.11 166.50 116.49 111.38 127.63 163.47 123.63 113.53 120.22 157.57 120.96 112.04 133.98 157.12 1971-72 137.25 104.11 168.48 124.36 120.83 139.62 165.38 132.75 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1973-74 190.01 112.92 248.58 166.07 165.38 164.44 243.68 176.39 167.86 136.75 222.65 172.02 165.90 174.48 240.72 1974-75 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 217.42 166.13 217.65 232.79 215.89 200.98 325.16 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.63 238.64 423.84 1978-78 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.80 393.98 303.91 306.30 246.79 431.24 1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	113.2																
1970-71 126.66 104.11 166.50 116.49 111.38 127.63 163.47 123.63 113.53 120.22 157.57 120.66 112.04 133.98 157.12 1971-72 137.25 104.11 168.48 124.36 120.83 139.62 165.38 132.75 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1972-73 153.52 93.67 186.11 134.91 134.08 146.74 182.58 145.08 137.26 117.39 174.35 138.11 136.29 156.02 187.69 1973-74 190.01 112.92 248.58 166.07 165.38 164.44 243.68 176.39 167.86 136.75 222.65 172.02 165.9° 174.48 240.72 1974-75 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 217.42 166.13 217.65 232.79 215.89 200.98 325.16 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.63 238.64 423.84 1977-78. 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.80 393.98 303.91 306.30 246.79 431.24 1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	117.0												113.10	154 34	101.12	123.78	
1971-72 137.25 104.11 168.48 124.36 120.83 139.62 165.38 132.75 122.41 124.16 158.96 127.98 121.14 147.4 160.44 1972-73 153.52 93.67 186.11 134.91 134.08 146.74 182.58 145.08 137.26 117.39 174.35 138.11 136.29 156.02 187.69 1973-74 190.01 112.92 248.58 166.07 165.38 164.44 243.68 176.39 167.86 136.75 222.65 172.02 165.90 174.48 240.72 1974-75 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 217.42 166.13 217.65 232.79 215.89 200.98 325.16 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.63 238.64 423.84 1977-78. 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.80 393.98 303.91 306.30 246.79 431.24 1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 366.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	115.2																
1972-73 153.52 93.67 186.11 134.91 134.08 146.74 182.58 145.08 137.26 117.39 174.35 138.11 126.29 156.02 187.69 1973-74 190.01 112.92 248.58 166.07 165.38 164.44 243.68 176.39 167.86 136.75 222.65 172.02 165.9¢ 174.48 240.72 1974-75 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 217.42 166.13 217.65 232.79 215.89 200.98 325.16 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.63 238.64 423.84 1977-78. 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.80 393.98 303.91 306.30 246.79 431.24 1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1988-84 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	127.9																
1973-74 190.01 112.92 248.58 166.07 165.38 164.44 243.68 176.39 167.86 136.75 222.65 172.02 165.90 174.48 240.72 1974-75 245.11 142.56 396.72 217.11 214.39 190.02 389.54 231.01 217.42 166.13 217.65 232.79 215.89 200.98 325.16 1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.63 238.64 423.84 1977-78. 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.80 393.98 303.91 306.30 246.79 431.24 1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	132.8																
1974-75	145.6																
1975-76 297.64 168.82 409.08 263.09 263.72 255.64 400.80 281.00 266.45 207.75 359.17 273.79 265.49 271.38 397.87 1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.63 238.64 423.84 1977-78. 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.80 393.98 303.91 306.30 246.79 431.24 1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	175.5																
1976-77 324.06 160.20 428.32 280.64 287.67 226.12 419.58 289.59 290.36 191.29 385.06 290.92 290.63 238.64 423.84 1977-78. 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.80 393.98 303.91 306.30 246.79 431.24 1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	225.4																
1977-78. 340.88 168.32 437.28 293.78 302.74 234.07 428.36 301.84 216.73 198.80 393.98 303.91 306.30 246.79 431.24 1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	279.63																
1978-79 357.57 191.39 519.07 314.94 317.86 281.30 507.00 334.23 315.88 232.31 441.37 319.48 321.59 297.53 542.84 1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	285.39																
1979-80 389.98 246.17 611.25 354.60 348.57 401.11 596.68 402.55 351.05 310.23 510.03 362.35 352.40 427.11 650.26 1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	297.8																
1980-81 417.85 305.12 682.88 385.89 371.67 486.69 666.32 443.60 374.58 378.71 559.02 395.69 375.88 518.44 739.50 1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	333.2																
1981-82 448.87 316.25 672.71 417.82 396.46 504.35 655.71 461.94 399.79 392.28 546.77 413.11 401.58 537.23 733.91 1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	402.8																
1982-83 480.14 335.12 640.69 433.14 424.95 532.22 624.83 483.08 427.75 413.16 507.20 436.75 431.90 566.70 718.10 1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	453.79																
1983-84 525.13 358.02 669.71 466.74 465.35 539.81 652.95 512.82 468.54 431.75 522.79 470.59 472.35 570.24 758.86	474.09																
	499.8						_										
	529.40	758.86	570.24	472.35	470.59	522.79	431.75	468.54	512.82	652,95	539.81	465.35	466.74	669.71	358.02	525.13	1983-84

	-10		ALUE ADDE	D j	GROSS V	ALUE OF	OUTPUT	NE	T AVAILA	BILITY			RICULTURE	td)
	Years	Consumption and Intermediale	Consumption and Investment Goods	Investment and Intermediate Goods	Consumption and Intermediale Goods	Consumption and investment Goods	Investment and Intermediate Goods	Consumption and Interme- diate Goods	Consumption and Invest- ment Goods	Investment and Intermediate	Consumption and Intermediate Goods	Consumption as and Invest-	Investment and Intermediate	
	-10 1-201 11	1_1_	2	3	4	5	6	7	8	9	10	11	12	
	1951-52 1952-53 1953-54 1954-55	91.83 103.88 85.73 86.65	88.35 93.92 103.50 88.63 86.53	89.79 89.53 85.31 79.45 89.47	83.48 89.80 102.20 86.65 86.29	86.94 92.19 102.43 89.06 86.34	87.10 87.03 85.45 78.16 87.77	81.88 88.25 97.18 84.17 87.32	89.09 92.23 96.44 87.41 86.06	97.26 94.99 87.90 81.30 88.83	82.63 89.50 100.06 85.35 87.43	83.28 89.81 103.00 88.20 85.68	97.60 96.69 89.09 81.47 96.89	
	1956-57 1957-58 1958-59 1959-60 1960-61	91.01 98.33 95.55 100.00 98.60	91.85 99.13 94.99 100.00 98.59	96.46 99.42 96.21 100.00 97.12	90.65 99.09 95.97 100.00 98.92	91.98 100.26 95.68 100.00 98.42	95.73 98.76 94.74 100.00 96.75	91.77 98.07 96.39 100.00 98.08	94 68 100 81 94 96 100 00 98 20	102.24 102.90 96.78 100.00 97.69	91.05 97.25 94.95 100.00 98.00	90.02 57.25 94.12 100.00 98.32	102.39 102.46 98.76 100.00 96.86	
	1961-62 1962-63 1963-64 1964-65	99.55 99.30 100.05	100.00 101.44 101.93 102.62	100.28 105.87 108.54 105.45	99.42 99.13 99.56 102.36	99.77 101.12 101.59 103.32	99.58 105.44 107.72 105.02	98.81 100.34 101.16 103.30	99.46 104.57 105.59 104.20	100.04 107.37 110.32 105.83	98.32 99.42 99.40 101.36	98.79 100.82 101.39 100.35	97.77 99.81 102.79 101.31	
	1965-66 1966-67 1967-68 1968-69	111.70 112.93	106.39 111.64 115.24 117.82	112.67 121.01 127.17 135.76	106.00 114.73 112.59 113.47		128.39 136.91	108.22 114.55 114.36 116.57	109.60 115.47 119.31 124.24	112.39 119.51 125.87 134.74	105.21 110.31 110.15 111.83	104.56 111.36 111.76 113.40	104.13 108.27 113.55 120.31	
	1969-70 1970-71 1971-72 1972-73 1973-74	112.99 116.05 124.80 138.00 169.50	119.20 123.87 130.42 148.65 189.96	142.15 151.57 159.49 184.73 228.18	113.84 116.91 124.09 138.39 220.67	119.56 124.42 130.55 147.67 188.28	153.09 160.04 1.83.5	117.01 121.15 132.45 4 145.79 174.04	128 24 133 16 139 33 156 54 195 50	139.16 149.43 154.87 172.50 214.69	111.52 114.49 122.79 136.29 162.84	111.28 116.87 125.11 140.24 174.41	123.06 129.58 132.95 134.00 161.26	
	1974-75 1975-76 1976-77 1977-78	220.77 283.88 305.01 322.00	263.94 314.49 343.03 357.98	330.88 381.31 391.51 401.64	219.20 281.94 302.32 317.92	260.17 307.54 334.34 348.78	330.85 380.22 392.31 403.08	225.07 277.86 288.41 321.05	285.15 313.43 335.08 348.55	326.30 350.84 360.97 372.67	209.85 256.95	227.67 272.93 302.25 301.20 336.93	252.11 247.92 255.33	ego potar
,	1978-79 1979-80 1980-81 1981-82	376.29 415.32	493.11	500.49 632.98 742.46 739.39	326.14 370.71 406.55 431.91	467.13 484.88	619.30 721.67 716.38	320.57 382.13 434.27 455.91 492.80		429.39 534.77 621.47 620.76	343.44 372.80 396.29	366.77 403.36 438.86 437.98	380.13 445.81	
	(Luble Co	rrd)			PURCHASE	BY AGRI	CULTURE		`	1				
	Years Years	te s imption nvest-	ment Goods and Investment and intermediate	Ę	(Alt	ment Goods Investment and	(-3)	ds	ment Goods	3				

2	(6	Iternati	ve-2)	PURC	(Alterna			Alternat		
	1		2	tion er- Goods	Arterna	and te	lon F- Gcods	Alrerbat	and te	\$19.600 F14894 F 10885 A 100 F
	5 7	Consumption and invest- ment Goods	Investment a Intermediate Goods	Consumption and Inter- mediate Goo	Consumption and invest- ment Goods		la . 00	5 t		
Years	Consumption and Intermedia te Goods	by ve	Investment Intermedia Goods	Inter-	Consumption and invest ment Goods	Investment Intermedia Goods		Consumption and Invest- ment Goods	Investment Intermedia Goods	
	Par est	F -	S	Consump and Inte	E - 5	STE	Sump Inte	umpt Inve	s tra	
	Consumand In	Const and ment	Inte	d d	Const and ment	Inter Inter Goods	d d a	P	od:	THE PROPERTY OF THE PARTY OF TH
dlagrain.			= = 3	Con	S & 5	== 3	Con	Consu	Inte	
	13	14	1 15	16	17	18	19	20	21	
1951-52	83.22	84.89	97.98	82.76	84.84	104.64	83.42	84.38	95.26	
1952-53	90.15	90.95	96.92	89.16	90.71	97.66	90.70	91.00	96.98	
1953-54	99.65	99.30	89.27	100.12	99.06	88.18	99.21	99.87	90.38	
1954-55		87.57	81.12	86.16	87.76	82.29	83.55	87.48	80.90	
1955-56		85.71	93.47	87.49	85.50	92.49	88.79	85.86	94.34	
1956-57		91.13	99.89	91.11	91.31	102.78	92.07	90.53	98.96	
1957-58		97.51	100.76	97.03	98.09	103.80	96.36	96.33	99.61	
1958-59		93.77	97.49	94.88	93.92	97.02	95.03	93.18	98.04	
1959-60		100.00	100.00	100.00	100.00	100.00	100.00	100.00	100.00	
1960-61		98.15	98.31	97.98	98.20	97.09	98.36	98.09	98.71	
1961-62		98.70	102.49	98.32	98.18	98.32	100.38	98.67	103.43	
1962-63		101.98	105.96	99.43	102.04	103.83	101.54	101.61	105.61	
1963-64		101.78	108.8€	98.98	102.30	106.15	102.27	101.06	108.75	
1964-65 1965-66		107.33	101.07	101.07	102.46	106.31	103.34	101.36	107.69	
1966-67		106.69	111.42	104.98	107.17	107.61	106.97	105.78	111.34	
1967-68		111.99	116.00	109.96	112.56	112.71	111.63	110.98	115.84	
1968-69		113.15	120.60	110.07	113.86	119.85	111.94	111.63	119.21	
1969-70		115.90	128.38	111.93	116.72	128.27	115.10	114.17	127.03	
1970-71	118.83	120.51	132.34	111.62	116.97	131.48	115.71	114.32	130.97	
1971-72		128.53	150.33	122.74	121.80	140.15	119.70	118.02	139.00	
1972-73		144.50	162.43	135.99	143.87	142.73	130.34	126.36 142.82	150.25	
973-74		180.17	190.73	162.63	179.74	182.63	169.17	178.90	163.01 186.61	
1974-75		243.97	254.94	209.25	243.78	272.78	210.70	237.97	237.98	
1975-76		285.22	302.48	257.11	282.92	288.60	267.64	283.12	296.46	
1976-77		309.18	289.66	274.62	306.84	294.64	272.46	308.00	276.47	
1977-78	286.61	324.01	297.61	288.25	321.01	303.04	285.53	322.96	284.32	
1978-79	312.54	349.75	356.06	306.55	342.05	344.05	313.28	351.02	347.63	
1979-80	371.85	382.86	469.01	844.52	379.25	417.04	378.72	391.99	471.58	
1980-81	414.80	420.98	549.97	375.19	407.32	475-18	425.99	424.19	562.30	
1981-82	438.17	440.33	559.10	398.50	425.89	474.94	449.27	445.71	578.18	
1982-83	468.91	459.29	568.65	425.38	441.93	463.60	479.28	469.95	596.53	
1983-84	497.14	497.51	581.54	462.63	478.23	480.64	502.46	510.37	609.71	WAS ASSESSED AS THE CONTRACT OF THE PARTY OF

#### 97.41 110.35 1957-58 99.85 100.35 86.31 96.91 97.62 105.62 86.01 94.37 109.62 86.10 1958-59 95.94 98.97 88.67 95.29 95.26 99.72 89.16 94.33 95.14 93.77 88.97 92.67 100.00 100.00 100.00 100.00 100.00 100.00 100.00 100.00 100.00 1959-60 100.00 100.00 100.00 111.56 106.43 103.53 109.41 102.93 108.45 109.88 111.40 103.71 108.21 107.91 1960-61 111.25 105.97 116.04 102.18 107.19 107.47 106.05 101.72 105.15 108.91 1961-62 106.70 115.25 101.24 103.02 105.73 103.45 101.53 108.22 102.04 116.72 102.44 105.78 102.09 103.03 1962-63 115.73 110.64 116.06 102.03 111.74 109.45 101.32 107.57 101.51 110.44 109.12 1963-64 120.08 115.16 108.96 114.59 131.58 107.73 116.03 116.75 126.15 108.43 113.97 1964-65 120.98 126.47 106.81 115.86 110.21 125.55 111.41 113.56 108.48 130.21 110.98 109.29 1965-66 113.48 124.16 110.12 150.05 128.10 117.21 153.96 127.92 117.49 139.41 1966-67 115.77 115.36 120.13 136.44 152.75 138.66 103.52 126.16 140.83 110.95 126.51 131.82 128.14 1967-68 145.47 96.51 123.17 110.30 127.82 133.01 103.64 130.57 111.22 126.06 111.39 132.51 123.93 126.64 118.04 133.32 1968-69 132.79 128.79 135.06 127.63 140.40 134.94 130.90 1969-70 140.64 108.46 137.64 117.01 130.72 119.32 139.83 136.20 132.92 121.03 151.56 152.08 140.40 1970-71 142.53 117.86 148.42 123.88 146.51 144.91 159.33 151.05 144.45 133.14 158.25 127.43 147.05 1971-72 153.01 155.20 132.16 165.82 154.09 180.38 168.05 165.60 163.79 182.95 171.93 178.42 155.79 1972-73 182.44 150.54 217.04 215.98 259.63 232.33 216.76 205.64 258.60 227.70 1973-74 242.72 198.61 253.60 210.04 275.37 224.12 309.27 277.43 247.81 310.59 273.94 285.88 1974-75 284.24 203.73 300.33 237.28 287.00 253.76 358.45 311.78 1975-76 286.33 258.06 357.23 299.11 306.92 242.54 346.74 278.74 307.02 308.50 447.12 308.58 324.28 448.60 345.42 353.01 317.50 334.33 298.19 433.58 1976-77 404.55 365.37 333.39 455.67 401.15 459.77 419.98 1977-78 426.34 323.71 443.25 358.87 397.66 397.16 1978-79 394.58 430.03 356.09 367.37 342.41 438.65 384.97 369.72 386.42 440.94 349.60 400.33 382.18 381.94 475.17 416.67 334.52 474.21 1979-80 416.54 343.52 463.06 367.44 380.47 404.24 420.71 503.16 447.98 431.64 452.87 382.29 487.86 398.30 406.10 377.85 503.81 1980-81 459.36 394.29 553.04 474.51 455.97 430.68 552.96 488.65 430.90 399.56 531.71 1981-82 497.87 473.12 439.91 519.36 609.43 1982-83 513.59 411.41 584.03 454.44 475,80 403.59 608.99 502.43 565.04 470.47 665.82 547.88 490.84 502.47 678.03 1983-84 533.70 495.26 650.19 507.57 491.94

66.81 81.22

90.95 102.24

89,72

74.92

81.48

77.38

78.66

75.23

79.38

89.65

67.19

75.24

91.46

81.45

87.72

96.20

81.07

75.59

78.11

74.22

77.02

87.13

1954-55

1955-56

1956-57

67.46

74.33

90.42

81.55

90.10

102,01

80.89

80.23

79.21

77,16

82.72

92.39

AGRICULTURAL INPUTS IN PAKISTAN - 1953-54 TO 1983-84

Years	Value Added by Agriculture (Million Rs.)	Land Cropped Acres(in Mil)	Land (Net Area Sown in Millions)	Land (Culti- vated Area in Millions)	Agricultural Labour Force (in millions)	Livestock (Million Heads)	Tubewells (in Nos.)
	122126 13712		100 772 107	130 38 - 1863	de Crezent e l		
1953-54	4532	32.77	29.94	38.39	.6,21		27.04 129.
1954-55	4320	32.80	29.30	37.86	6.32	6.5	990
1955-56	4406	34.32	30.44	38.70	6.38	6.1	1300
955-57	4502	35.00	31.22	39.56	6.84	6.1	1600
957-58	4578	34.42	31.01			8.2	1900
959-59	4322	35.41		40.14	7.48	8.4	2200
959-60	4775	36.29	31.92	40.04	7.86	8.4	3300
960-61	4709		32.31	40.80	7.93	6.8	4600
961-62	5127	36.72	32.78	44.76	8.97	5.9	8000
952-63	5486	37.69	33.70	44.23	9.05	7.0	13000
963-64	5638	38.21	34.04	44.55	9.25	7.5	18400
964-65	6018	37.40	33.14	45.30	9.89	7.0	25000
955-66		414	34.98	46.26	10.10	7.5	31600
966-67	5993 6421	38.41	34.42	47.54	10.21	8.3	40207
967-68	7484	40.54 41.85	35.16	47.60	10.38	7.9	51327
968-69		41.85	36.76	48.00	10.49	7.6	62163
969-70	7924	40.12	35.21	47.57	10.38	7.3	72149
970-71	8916	41.45	35.92	47.53	10.33	7.0	72223
971-72	8463	41.07	35.68	47.47	10.73	6.6	89143
972-73	8843	41.01	35.42	47.16	10.86	7.9	99373
	8951	41.82	34.75	47.23	11.00	8.0	109514
273-74	9429	45.15	37.52	47.87	11.24	8.1	
974-75	9134	42.90	36.48	48.29	11.54	8.2	120483
75-76	9672	44.51	37.20	48.98	11.86		144278
76-77	9864	44.98	37.32	49.31	11.98	8.3	150087
77-78	10077	45.69	37.45	49.28	12.32	8.4	155864
73-79	10338	47.69	38.08	49.37	12.80	8.4	161114
979-80	11191	47.49	38.33	49.87	12.84	8.5	166985
930-81	11588	47.76	18 23	10 7G	12.04	8.8	176517

1c . 3.22

Agar egate Input Index, Value Added Index and the Total Factor Productivity Index 1953-54 to 1983-84

( Base Period 1959-60 )

	Value Anded	abor	-PATT		LAND	INDEX	DSING	AGGR EGA	TE INPUT	INDEX USING	DIM FAC	CTOR PRIDD	-5500
Years	Injex	Index	stock	Capital	Cropped	Net Area	Cultivated	1 Cropped	Net Area	Cultivated	LIVITY	KECKI	
			Ind ex	Index	Ar ea	SWI	Area	: Area	SWI	Ar ea	I I	II	III
(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)	(6)	(10)	(11)	(12) (1	13)	14)
10	94,91	78,31	IO.	91.2	90.30	95.66	94.09	83,36	84.05	84.47	113.85 112	.92	112,36
554	2.16	79.70	9	86.1	90.38	89.06	92.79	83.59	83.68	84.30	108.23 108.	11	107.3
1955-56	92.27		89.71	86.5	94.57	94,21	94.85	85.31	S	85,39	108.16 108.	23	108.36
956	2	86.25	0	115.9	96.44	96.63	96.96	92.72	92.78	92.88	101,68 99	9,45	99.3
957-15	ന	94.32	(43	119.0	94.85	95.98	98,38	97.38	97.72	98.42	98.44 98	8.11	97.43
1958-59	00.9	99.12	123,53	120,4	97.57	98,79	98.14	101.17	101.53	101.34	99.81 99	9:46	99.66
1959-60	S	100,001	0	100,00	00.0	100.00	100.00	100.00	100.00	100.00	100,00 100	00.0	100,001
19-0961	98,62	113,31	101.47	105.8	1.18	101,45	109,70	112.89	108.86	108.94	87,36 9	90,59	90,5
1961-62	07,3	114.25	102.94		3.86	104.30	108.41	111.10	111,23	112,22	96 69 96	6.53	95.4
1962-63	114.39	116,64	110.29		5.29	105,35	109.19	114.56	114.58	116.07	100.29 100.	0.27	98.9
1963-64	18.0	124.72	102.94	128.9	103.06 1	102.57	111,03	118,84	188.69	121,18	99,35 9	9.48	200
1964-65	26,0	127,36	110.29	144.3	0.61	108.26	113.38	124.44	123,75	125,25	101.28 101	84	100.6
9	125.51	128.75	122.06	166.5	5.84	106.53	166.52	126.46	126.67	129,61	99,25 9	80.6	96.8
1966-67	34.4	130.89	136.18	175.2	1.71	108.82	166,67	130,47	129.62	131.93	103.06 10	3.74	101.9
1967-68	56.7	132,28	11.76	184.9	5.35	113.77	117.65	133.51	133.05	4	117.39 117	7.80	116.3
1968-69	55.9	130,89	107.35	193.6	0.55	108.97	116,84	131.50	141.04	ers	126,20 119	9.60	117.5
1969-70	86.7	CO	102.94	189.5	4.22 1	11.17	116.49	132,53	131,63	133,20	68	141.85	140.1
1970-71	77.2	135,31	90.76	2.05.7	3.17	110.43	116.35	137.10	136.30	138.04	129.27 130	0.03	128.3
1971-72	85.1	136,95	116.18	236.8	3.01	109.62	115,59	141.69	140.70	142.45	130,70 131	1.62	130,0
1972-73	8	138,71	117.65	251.2	5.24	107.55	115.76	145.08	142.82	145.23	129,20 131	1.25	129.
1973-74	97.4	141.74	119.11		4.41 1	16.12	117.33	151,38	148.94	149.29	130.44 13	132.58	132,2
1974-75	91.2	145,52	120,59	298.5	8.21 1	12.91	118,36	155.53	153.97	155,58	122.99 124	4,24	122.9
1975-76	202,55	149.56	122,06	307.4	122,65 1	15.13	120.05	160.27	158.06	159,50	126.38 128	8.15	126.9
1976-77	06.5	151.07	123.53	316.5	3,95	115.51	119.63	162,61	160.13	161,34	04	129.07	129.
1977-78	11.0	155,36		2	5.90	115.91	120.78	165.46	163.53	164.96	-	29.05	127.3
1978-79	21	161,41	126.47	333.2	1,41	117.86	121.00	172.86	168.88	169.80	24	128.20	127.5
979	234.	161.92	129,41		0.86	118.63	122.23	174.77	171.18	172.23	.10	0	136.0
980	1.4	172.13	130.88	362.5	131.61 1	.18.32	122.03	182,67	178.77	179,86	132,85 13	35.75	134 . 9
81-3	251.	CO	32.		134.69 1	90.67	122.70	187.42	182.82	183.89	134.37 13	37.75	136.9
98		185,37	135.29	00	136.87 1	19.59	123,11	194.53	189,45	190,48	40 1	32 . 37	132.1
100	51.	89	00	395.2	139.07 1	125.12	m		193.43	134,44	125,43 1.	37.38	129.

#### CHAPTER IV

# INTRA-SECTORAL PARITY ISSUES IN PRICING OF AGRICULTURAL OUTPUT AND INPUTS

#### 1. INTRODUCTION

The terms of trade discussed in the previous chapter are an indicator of the profitability of the agricultural sector and of the purchasing power of goods and services of agricultural income. After the initial allocation of resources to the agricultural sector, the further allocation of these resources within the sector is influenced by the relative levels of costs, prices and the income from different agricultural commodities and other related farm activities. Most planning agencies set specific targets of production for different crops. Agricultural price policy has important implications for the achievement of these planned targets. An appropriate relationship needs to be maintained between costs and prices of, and the income from different commodities if planned targets are to be fulfilled.

In Pakistan, there has been some controversy that the prices of agricultural crops fixed by the government are not in parity with the costs and that producers of some crops are unduly handicapped. The purpose of this chapter is to present evidence on various notions of inter-crop, input-output prices and income parity for a number of crops and evaluate the price policy from the view point of parity in costs and prices of and the income from different crop activities.

the base year are also presented.

### 2. COSTS OF PRODUCTION

The farming community generally maintains that it looses money in the farming business and argues for fixing prices of agricultural output at levels that cover 'costs of production'. The policy-maker needs reliable figures on costs of production as a basis for judicious price policy. It is important to understand, however, that theoretically there is a schedule of costs and outputs and that these schedules vary by farm and by agro-climatic zones and are a function of the price of output. Data problems further compound the situation as reliable farm surveys on costs of production are often not conducted. Valuation of family labour and land also creates problems in the estimation of costs of production.

The state of the art in the estimation of costs of production data for different crops in Pakistan is in its first stage of development. The Agricultural Prices Commission established in 1981 has conducted some large scale farm surveys but their findings have not been released to outside analysts. The earlier studies conducted by the Planning Commission and the Ministry of Agriculture on costs of production of different crops are based on small samples. The findings of these studies, in the light of the sampling and other problems mentioned above, should be taken with the proverbial pinch of salt.

Table IV.1 provides information on cost of production and price-cost ratios for five major crops, viz. wheat, rice, cotton, sugarcane and maize for selected years between 1969/70 and 1980/81. The index numbers of costs of production per maund with 1969/70 as the base year are also presented.

TABLE TV.1

COST ( PRODUCTION AND PRICE-COST RATIOS FOR SELECTED CROPS

Years	Ţ	at	Paddy Coarse	Paddy Fine	American Cotton	Sugarcane	Ма
			CO	ST OF PI	CODUCTION	(Rs. per mauno	i)
1969-70	3	13	25.57	33.47	53.64	2.71	14
	(	3)	(100)	(100)	(100)	(100)	(1
1975-76	2	80	30.53	44.45	86.09	5.48	35
	(	5)	(119)	(133)	(160)	(202)	(2
1976-77	3	14	29.21	45.05	115.03	4.70	43
	(	5)	(114)	(135)	(214)	(173)	(2
1977-78	4	30	29.80	46.69	132.51	4.99	47
	(	3)	(116)	(148)	(247)	(184)	(3
1978-79	4	<b>i</b> 9	28.41	52.47	143.09	5.60	46
	(	<b>;</b> )	(111)	(157)	(267)	(207)	(3
1979-80	4	37	27.90	53.52	111.65	7.11	48
	(	))	(109)	(160)	(208)	(262)	(3
1980-81	4	57	38.79	73.32	117.82	8.41	51
	(	1)	(152)	(219)	(220)	(310)	(3
			PR	ICE-COST	RATIO		
1969-70	0.		1.27	0.97	2.08	0.70	1
1975-76	1.		1.94	1.33	3.37	0.54	0
1976-77	1.		2.26	1.46	3.42	0.67	0
1977-78	1.		2.15	1.29	3.02	0.63	0
1978-79	1.		2.41	1.31	3.89	0.58	0
1979-80	1.		2.59	1.35	4.45	0.52	0
1980-81	1.		2.33	1.23	4.10	0.51	0

Note 1: Cos of production includes expenditure on: (i) prepara
(ii Seed bed preparations, (iii) Seeds, (iv) Sowing c
(v) anuring (vi) Fertilizing (vii) Watering (viii) Pl
(ix Water charges, (x) harvesting, (xi) threshing, an
2; Fig as in parenthesis are the index numbers with base p

Source: Stu as on Cost of Production of Crops, Ministry of Food, and poperative, Government of Pakistan, (various issues

lows some measure of confidence in the reliability de-

The table shows that the costs of production for all crops have increased considerably during 1970s. However, quite divergent rates of increase in the costs are found for different crops. The percentage increase is highest for maize (252 per cent) which is followed by sugarcane (210 per cent), wheat (174 per cent), cotton (120 per cent), paddy fine (119 per cent) and paddy coarse (52 per cent). Like the cost of production, the price-cost ratios for wheat, coarse paddy, fine paddy and American cotton show an increasing trend. It shows a near price-cost parity for wheat and maize crops. The ratio is above the unit-level for wheat, coarse and fine varieties of paddy and cotton. The price-cost ratios for sugarcane and maize show a downward trend and are let than one for all the years, except the ratio for 1969/70 for maize, which was 1.12.

#### TRENDS IN PRICES OF INPUTS

Fixed resources, owned or rented, are not critical determinants of the allocation between different farm activities and/or of the level of output of different crops. The prices of variable inputs in conjunction with produce prices are important datum for farmer's decision in this regard. Table IV.2 shows the index numbers of prices of six important inputs with 1969/70 as the base year.

The average percentage increase in the prices of agricultural inputs between 1969/70 and 1963/84 comes to 342. During the same period the percentage increase in prices is highest for pesticides (678 per cent) and lowest for fertilizer (267 per cent). The percentage increase in prices of all other inputs falls in the range indicated above. The effect of sharp increases in the prices of agricultural inputs is reflected in the increases in the costs of production of different crops shown in Table IV.1. This finding allows some measure of confidence in the reliability of data on cost of production of crops.

### TUMBERS OF PRICES OF AGRICULTURAL INPUTS

ransport harges		Pesti- cides	Electri-	Ferti- lizer	See	Pears
(7)	(6)	(5)	(4)	(3)	(2)	1
100	100	100	100	100	100	1969-70
130	321	100	100	134	124	1970-71
144	328	100	100	128	135	1971-72
168	383	1.75	100	1.00	176	1972-73
201	399	183	110	222	216	1973-74
252	420	280	117	286	289	1974-75
266	445	400	143	256	283	1975-76
288	478	450	143	254	31.1	1976-77
333	557	300	143	243	342	1977-78
355	432	400	257	230	364	1978-79
375	409	480	314	248	380	1979-80
402	442	589	327	318	402	1980-81
452	515	642	354	340	444	1981-82
472	513	708	362	367	483	1982-83
500	51.5	768	379	367	484	1983-84
	515 513	642 708	354 362	340 367	444	1981-82 1982-83

Note: 1. Expen iture on each input relative to total expenditure on was used to assign weight to each of the inputs to calcu ate the aggregate input price index.

Colum Natio Feder (Vario

3 (2), (3), (5), (6) and (1) have been taken from al Accounts of Palistan (Product and Expenditure), Bureau of Statistics, Government of Pakistan, Issues).

Colum

(4) has been taken from Monthly Statistical Bulleti Bureau of Statistics, Government of Pakistan, (Vari

tivity of cross in the next section:

#### 4. TRENDS IN PRICES OF AGRICULTURAL COMMODITIES

The government announces the procurement and/or minimum support prices for a few selected crops every year on the basis of the recommendations of the Agricultural Prices Commission. For all other crops, prices are determined by market forces. Even in the case of crops for which the government intervenes, market forces play a dominant role in determining the actual prices received by the farmers. Table IV.3 presents data on producer prices for 12 crops for the period 1969/70 to 1983/84. Indices of farm prices for each of the 12 crops with 1969/70 as the base year are also shown in the table.

It would be seen from table IV.3 that the prices of all crops show considerable increase over the period of study. The weighted average price for all crops shows an increase of 334 per cent. The highest increase in prices is for cotton (533 per cent) and the lowest increase is for sugarcane (255 per cent). Excepting the case of cotton, it turns out that commodities for which government intervention obtains, show smaller increases in prices than crops in whose case government does not intervene in the pricing or marketing operations. The divergent rates of increase in prices of different crops have implications for inter-crop resource allocation. However, two additional considerations need to be taken into account to measure the incentives being provided to farmers for the production of different crops. The increase in the prices of crops needs to be examined in the light of both the increases in prices of major purchased agricultural inputs and changes in productivity of crops. The table shows that increase in the prices of purchased inputs (342 per cent) is much higher than the increase in the prices of n sugarcane, jowar, maize, tobacco and rape and mustard seeds. In case of wheat and weighted average of all the selected crops the increase in output prices was less than the increase in input prices, but the difference between them was relatively small. The increase in the prices of cotton, gram, barley, bajra and sesamum was much higher than average increase in the prices of agricultural inputs during the period. We now turn to the examination of increases in output prices to the cha in productivity of crops in the next section.

Table IV.3

PRODUCER PRICES FOR MAJOR CROPS ...

Rice	Wheat	Sugarcane	Cotton	Gram	Barley	Jowar	Bajra	Maize	Tobacco	Mustard_	Sesamue	OPORAL!
32.47	(100, 00)	(100.00)	111.57 (100.00)	28.18 (100.00)	14.37 (100.00)	14.44 (100.00)	16.95 (100.00)	16.35 (100.00)	122.43 (100.00)	39.68	41.99 (100.00)	(100)
30.94 95.29)	18.44 (105.79)	(110.00)	133.74 (119.87)	21.61 (76.69)	14.78 (102.85)	14.97 (103.57)	76.20 (95.58)	16.24 (99.33)	131-13 (107.13)	39.00 (98.28)	59.72 (142.22)	(105.74
28.93 87.43)	(113.31)	2.95 (155.26)	136.39 (122.25)	25.12 (89.14)	17.02 (118.44)	18.44 (127.70)	18.55 (109.44)	18.35 (112.29)	130.87 (106.89)	39.30 (99.04)	55.99 (133.34)	(117.66
38.22	22.10	3.66 (192.63)	166.22 (148.98)	32.88 (116.68)	18.85 (131.18)	22.13 (153.25)	21.35 (125.96)	21.61 (132.17)	144.57 (118.08)	44.49 (112.12)	70.92 (168.90)	(139.62
47.74	28.67 (164.49)	(112.10)	261.85 (234.70)	35.50 (125.98)	26.84 (186.78)	16.20 (112.19)	25.08 (147.96)	20.19 (123.49)	146.47 (119.64)	(154.28)	80.89 (192.64)	(149.20
52.07 60.36)	44.27 (253.99)	(120.00)	232.92 (208.77)	41.21 (146.24)	26.43 (183.92)	29.75 (206.02)	50.39 (297.29)	37.21 (227.58)	206.98 (169.06)		107.32 (255.58)	(214,92
59.13)	42.51, (243.89)	2.95 (155.26)	289.85 (259.79)	40.98 (145.42)	31.28 (217.68)	28.44 (196.95)	50.05 (295.28)	35.20 (215.29)	215.60 (176.10)		105.19 (250.51)	
65.92	41.58 (238.55)	3.17 (166.84)	393.69 (352.86)	44.16 (156.71)	27.40 (190.68)	30.46 (210.94)	43.22 (254.99)	31.13 (190.49)	227.02 (185.43)	91.41 (230.37)	188.21 (281.52)	
64.05 97.26)	55.51 (318.4)	3.17 7) (165.84	400.15 (358.65)	75.55 (268.10)	37.33 (259.78)	29.30 (202.91)	51.55 (304.13)	36.69 (224.40)	239.60 (195.70)	103.43 (260,66)	129,23 (307.76)	
68.57	54.61 (313.31)	3.25 (171.05)	557.15 (499.37)	62.15 (220.55)	45.73 (318.23)	31.39 (217.38)	55.58 (327.91)	43.90 (268.50)	276.11 (225.52)	139.68 (352.02)	178,80 (425.81)	(319,85
72.19	56.29 (322.95)	3.73 (196.31)	496.60 (445.10)	68.08 (241.59)	51.88 (361.03)	40.01 (277.08)	76.00 (446.38)	41.43 (253.39)	305.78 (249.76)	166.11 (418.62)	216.09 (514.62)	(327.97
90.29	57.11 (327.65)	4.29 (225.79)	483.31 (433.19)	149.08 (529.03)	64.87 (451.43)	43.82 (303.46)	88.09 (519.71)	44.57 (272.60)	421.20 (344.03)	115.83 (291.91)	147.22 (350.61)	(340.09
00.63	(382.50)	4.63 (243.68)	508.25 (455.54)	210.00 (745.21)	60.02 (417.68)	48.79 (337.88)	76.03 (448.55)	(376,21)	392.76 (320.80)	97.15 (244.85)	120,75 (287.57)	(380,50
02.28	69.47 (398.56)	4.78 (251.58)	530.79 (475.75)	182.68 (648.26)	59.91 (415.91)	53.11 (367,80)	102.46 (604.48)	63.64 (389.24)	360.06 (294.09)	104.85 (264.23)	139.12 (331.32)	(399.15
02.91	73.27 (420.37)	4.85 (255.26)	706.61 (633.33)	137.29 (487.19)	67.90 (472.51)	44,42 (307.62)	107.02 (631.39)	57.67 (352.72)	398.54 (325.52)	143,63 (361,97)	186.64 (444.49)	(434.46

registered for the years 1969/70 and 1983/84. However, very

rape seed and mustach seeds the partty-indices were lower to

all years excepting 1978/79 and 1979/80 when high values of

### 5. PARITY INDICES OF PRICES OF AGRICULTURAL CROPS AND INPUTS

The relationship between prices received by farmers for different crops and the prices paid for agricultural inputs as measured by the parity indices is shown in Table IV.V.

The overall parity between prices of agricultural produce and inputs was very near—for the years 1969/70 and 1979/80. However, the parity index in 1970/71 falls to as low as 46.65. It increases to the figure of 80 in 1971/72, falls to 52.39 in two years and shows consisten recovery till 1979/80 when it reaches the figure of 100.12. After a fall, the index recovers to 98.30 for the year 1983/84. The overall impression is that of drastic declines in terms of trade in some years but gradual and sustained increases in the terms of trade in favour of farmers over most years.

The parity indices between the prices received and prices paid for crops reveal a lot of divergence. In the case of cotton, gram, barley and bajra price-cost parity indices were much higher in later years which is a reflection of the fact that output prices had outpaced input prices by a wide margin. In the case of rice, wheat, sugarcane, jowar, maize and tobacco the parity indices had remained lower than the base year for all years of the study. In the case of sesamum near parity was registered for the years 1969/70 and 1983/84. However, very high values of the parity index were found for 1978/79 and 1979/80. In the case of rape seed and mustard seed, the parity indices were lower than 100 for all years excepting 1978/79 and 1979/80 when high values of the parity index were registered.

	Mileat	Rice Wheat Sugarcane	Cotton	Gram	Barley	JOWar	Bajra	Maize	Торассо	Rape Seed &	Sesamum	ATT
	- 1											2000
00.001	5	100.00	100.00	100.00	100.00	100.00	100,001	100,001	100,001	100,00	100.00	100.00
48	48.67	50.61	55.15	35,28	47.32	47.70	43.98	45.70	42.29	45.22	65,43	48.65
55	52.21	71.55	56.33	11.58	54.58	58.85	50.43	51.74	49.26	45.64	61.44	81.87
46.62 50	50.20	76.26	58.98	46.19	51.93	60.67	49.87	52.33	46.75	44.39	66.87	55.28
51,631 5	57.76	39.37	82.42	44.24	65.59	39.40	51,96	43.37	42.01	54,18	67.65	52,39
49.14 7	77.84	36.78	63,98	44.82	56.37	63.14	91.11	69.75	51.81	62.41	78.33	65.87
56.27 7	75.36	47.97	80.27	44.93	67.26	60.85	91.23	66.52	54.41	64.87	77.40	69,61
59.63 7	70.07	49.01	103,65	46.03	56.01	61,96	74.90	55.95	54.47	19.19	82,69	70.59
52,74 8	85,14	1 44.60	95.89	71.68	69,45	54.25	81.31	59.99	52.32	69.69	82.28	76.03
Q1	95,80	52.30	152,68	67.43	97,30	66.46	100.26	82.10	68.95	107.63	130.19	97,80
	98.59	59.93	135,88	73.75	110.21	84.59	136.88	77.35	76.25	27.80	157.10	100.12
74.30	87.55	60.33	115.76	141.36	120.63	81.09	138.87	72,84	91,93	78.00	93.69	90,88
74.52	91.97	58,59	109.53	179.18	100,43	81.24	107,85	50.46	77.14	58.87	69.15	91,49
71.87	90.93	3 57.40	108.54	147.90	95.12	83.92	137,91	88.81	67,10	60.29	75.59	91.07
71.75	95.17	57.79	143,38	110.30	106.97	69.64	142.94	79.85	73.70	81,95	100.63	98,36

Share of gross income from each crop in total gross income was used to assign weights to producer naise indicates of the crops to calculate named indicates indicated the crops to calculate named indicated the crops indicated the crops to calculate named indicated the crops indicated the crops to calculate named indicated the crops to calculate named indicated the crops indicat te 1:

## 6. TRENDS IN GROSS INCOME OF CROPS AND INTER-CROP GROSS INCOME PARITY INDICES

Gross income per acre at market prices for 12 selected crops is shown in Table IV.5. The gross income has shown the maximum increase of 554 per cent in the case of gram and the minimum increase of 78 per cent in the case of tobacco during the period 1969/70 to 1982/83. The ranking of crops in descending order is bajra (535 per cent), wheat (471 per cent), cotton (469 per cent), maize (381 per cent), barley (343 per cent), rice (270 per cent), jowar (262 per cent), sesamum (255 per cent) rape and mustard seed (217 per cent) and sugarcane (108 per cent)

These relative trends in gross-income by crops reflect public policy in the area of pricing of major crops which was designed to reduce the disparity in the value productivity of major crops. Sugarcane, an inefficient user of resources from the society's perspective, was the most profitable activity for farmers in 1969/70. By 1982/83, the disparity in the value productivity of rice, wheat, cotton and maize as compared with sugarcane was considerably reduced. In fact, cotton and rice were earning more income per acre than the sugarcane. The other important finding is that the prices of crops whose physical productivity increased markedly due to widespread adoption of high-yielding new varieties were relatively restrained with a view to pass on part of the benefits of technological advances to consumers.

The crop calendar in Pakistan is complex and a farmer can choose from various combinations of crops. There are two main crop seasons known as <u>Rabi</u> and <u>Kharif</u>. In the <u>Rabi</u> season, crops are sown if fall and harvested in spring. The main crops in this season are wheat, gram, barley and rape and mustard seed. The main crops in the <u>Kharif</u> season are rice, cotton, jowar, bajra, maize and tobacco. These crops

Table IV.5

-	ំ សូល (1)	WHEAT (2)	SUGANCANE (3)	COTTON .	GRAM (5)	BARLEY (6)	JOHAR (7)	BAJRA (8)	MATZE (9)	TOBACCO (10)	RAPE AND MUSTARD SEED (11)	SESAMIN (12)	ALL CR DR
.73	521 45 (100,00)	221.33 (150,00)	886,31 (100,00)	368.35 (100,00)	166.64 (100.00)	103.10 (100,00)	90,66 (100.00)	87.85 (100.00)	182.50 (100,00)	360,92 (100,00)	228.97 (100.00)	163.64 <sup>4</sup> (100.00)	(2030
71	(90,65 (94.09)	216,79	924.68 (93.05)	453.37 (123.08)	126,60 (75.97)	103.32 (100.21)	95,11 (104.91)	83,09 (94.57)	197,39 (108,16)	265.16 (73.47)	223.03 (97.40)	210.53 (128.65)	(95%45
72	457.11 (93.41)	254.4 (114.96)	1157.76 (130.63)	534.79 (145.18)	143.92 (86.36)	121.15 (117.51)	12::,92 (135.58)	95,44 (108,63)	221,98 (121,63)	243.92 (67.58)	228.27 (99,69)	204.09 (124.72)	}, {111,7
73	652.51 (125.14)	218.43 (135.01)	1477.55 (165.71)	629.45 (170.88)	193,73 (116,26)	135.72 (131.64)	144,88 (159.80)	115,06 (130,96)	256,62 (140.61)	224.45 (62.]9)	246.27 (107,55)	260.27 (159.05)	(105.89
74	640.47 (1:1,18)	388.69 (175.34)	854.96 (95.46)	1014.10 (2/5.28)	211.82 (127.11)	197.41 (191.47)	112.68 (124,28)	130.29 (148.29)	265,46 (145,46)	238.25 (66.01)	275.36 (120,26)	320,98 (196,15)	(190,3
15	814,56 (156,21)	633.75 (285.35)	785.52 (88,63)	788.25 (213.99)	245.52 (147.93)	202.34 (196.26)	192.79 (212.65)	266.57 (303.40)	491.29 (269.20)	371,36 (102.89)	405.65 (177.16)	470.71 (287.65)	(263.68
16	981.55 (188.23)	655.36 (296.10)	1169.26 (131.92)	872.27 (236.80)	249.98 (150.01)	237.15 (230.02)	181.94 (200.68)	267.19 (304.11)	493.47 (270.39)	249.90 (69.24)	534.24 (233.32)	449.27 (274.54)	(285.9)
.77	(214:45)	(291.53)	1295.46 (146.16)	995.70 (270.31)	283.97 (170.41)	211.65 (205.29)	)92.84 (212.71)	224.82 (255.88)	413.13 (226.37)	355.74 (98.56)	565.49 (246.97)	506.67 (309.62)	(298.0
79	1078.70 (206.85) 1200.94 (230.13)	791.8.7 (257.75) 681.12 (398.10)	1264.02 (142.52) 1678.15 (189.34)	1353.37 (367.41) .510.76 (410.14)	457.68 (274.65) 296.15 (177.72)	293.21 (284.39) 359.21 (348.41)	173.55 (191.43) 182.93 (201.77)	277.15 (315.44) 289.85 (329.90)	497.77 (272.75) 585.07 (320.59)	361.63 (100.20) 472.22 (130.84)	641.94 (280.36) 867.34 (378.80)	576.92 (352.55) 812.07 (496.25)	(365.1) (434.0
30	1237.06 (237,23)	954.29 (431.16)	1554.01 (175.3J)	1883.45 (511.32)	(122.83)	417.41 (404.85)	255.43 (281.74)	406.85 (463.07)	561.06 (307.43)	518.58 (143.68)	999.93 (436.71)	964.57 (589.45)	(502.9
81	1575.11 (302,83)	(017.72 (459,82)	1815.45 (206.07)	1776.93 (482.40)	646.24 (387.81)	478.12 (463.74)	277.32 (305.89)	503.36 (572.91)	629.97 (345.19)	713.81 (197.77)	761.83 (332.72)	651.85 (398.34)	(525.0
82	(353.63)	1131,51 (511,23)	1944,18 (219,35)	1861.59 (505.39)	722.22 (433,40)	463.01 (449.09)	303.51 (334.78)	401.06 (456.48)	839.39 (459.94)	681.81 (188.91)	645.43 (281.88)	520.49 (318.07)	(605.2
163	193°. iu (370,33)	1264.10 (571.3a)	1843.47 (207.99)	2095.45 (568.87)	1089.13 (653.58)	457,00 (443,25)	327.90 (361.68)	558.06 (636.17)	878.26 (481.24)	641.22 (177.66)	725.41 (316.81)	581.38 (355.28)	(691,8

El. Share of gross income from each crop in total gross income was used to assign weights to producer price indices of each of the crops to calculate overall weighted index of prices received by the agricultural sector.

figures in parenthesis are the index numbers with base year 1969-70.

maize and coston is measured. Colton is sore profitable as corporatelling

rice, except in 1973/76 while cutton is

of relative profitability, the ranking of summer crops is

of relative profitability, the ranking of susses crops is said

As mentioned previously, supercase is an emusi crop

and legister team of the control of

relative profitability of sugarcane to rice accent society at cotton and to wheat and maire show that sugarcanels advant-

are sown in the beginning of summer and harvested in the fall. Sugarcane is an annual crop. The reliance on relative prices of competing crops as a measure of relative profitability can be deceptive if differential growth in the physical productivity of crops occurs. A better measure in such situations is that of inter-crop gross income parity. Such indices for various crop combinations are given in Table IV.6.

Within the <u>Rabi</u> season, the gross profitability of wheat is compared to, two crops, gram and rape and mustard seed. The gross profitability of wheat is higher by different magnitudes relative to its competing crops. Relative to gram, the parity index is less than 100 only in 1982/83 while for all years the index is higher than 100. Similarly, wheat profitability is higher relative to rape and mustard seed for all years except 1979/80. The sharp rise in profitability of gram and rape and mustard seed in one year is due to the sharp rise in prices of these two crops. For the remaining 13 years of the study, the sustained and high rise in physical productivity of wheat explains high levels of its profitability.

Within the <u>kharif</u> season, the relative profitability of rice, maize and cotton is measured. Cotton is more profitable as compared with rice, except in 1973/74 while cotton is more profitable compared with maize for all years of this study except 1974/75 and 1975/76. In terms of relative profitability, the ranking of summer crops is cotton, maize and rice.

As mentioned previously, sugarcane is an annual crop and competes for the use of land with both summer and winter crops. The indices of relative profitability of sugarcane to rice and wheat, to wheat and cotton and to wheat and maize show that sugarcane's advantage in the

Table IV.6.

CHOICES OF INTER-CROP GROSS INCOME PARITY AND GROSS INCOME-INPUT PRICES PARITY

2000	4	111	Gross	Income	Parity	Index of			
מש	Wheat to	Wheat to	Rice to	Rice to Maize	Cotton to Maize	Sugarcane to	Sugarcane to Wheat and	Sugarcane to	Gross Income-Input Prices Parity Index
		Mustard	n	100	77.00	Wheat	Cotton	Maize	
	×.			3:					
01-6961	100.00	100.00	100.00		100.00	100.00	100.00	100.00	100.00
1970-71	128.87	100.52	76,45	86.99	113,79		81.89	90.75	43.91
1971-72	133.12	115,32	64.34		119,36		97.60	110.72	51,49
1972-73	116.13	125.53	73.23		121.53	-	105.90	121.20	42.31
1973-74	137.94	145.81	58,55		189,25		40.57	59.60	66.84
1974-75	193.57	161.63	73.00		79.49		36.75	31.81	80,81
1975-76	197.39	126.91	79.49		87,58		50.92	46.37	88.36
1976-77	171.08	118.04	79.33		119.42		52.52	55.77	87.56
1977-78	130.26	127.60	56.30		134.71		39.20	44.66	111.64
1978-79	224.00		56.11		127.93		46.68	52.15	132.50
1979-80	350,88	98,73	46.40		166.32		36,43	46.72	134.40
1980-81	118.57	138,20	62.78		139.75		43.48	50,50	140.30
1981-82	117.96	181,36	71,83		109,88	16	43.21	44.94	145.76
1982-83	87.42		65.10		118.21		36.51	39.21	157.84

early years of study was rapidly eroded as the inter-commodity gross income parity moved against sugarcane vis-a-vis all its competing crop combinations.

The gross income-input prices parity indices given in the last column Table IV.6 show that the parity index falls drastically in 1970/71 and becomes as low as 42.31 in 972/73. The steep decline is due to sharp increase in the prices of inputs without comparable increases in the prices of agricultural commodities and in the physical productivity of crops. In fact for some of these years, agricultural output also fell drastically. However, due to deliberate price policy of the government which had led to an increase in the prices of selected major crops and had resulted in increasing productivity through development programmes, the parity was restored by 1978/79. The index shows an increasing trend and was as high as 158 in 1982/83.

### 7. CONCLUSIONS AND POLICY IMPLICATIONS

The foregoing analysis has some obvious implications for public policies that need to be briefly highlighted. The relationship between increase in incomes and prices of inputs in the agricultural sector as measured by the gross income-inrut prices index is one of deterioration in early to mid-1970s but sustained and rapid rise since 1978/79. The incentives provided to sugarcane producers were drastically curtailed possibly to improve the allocative efficiency of resource use in the agricultural sector. There was also an attempt to pass on some of the benefits of technological advance in the high-yielding crops of wheat and rice to the consumers as their prices were somewhat restrained relative to certain other crops. In view of the shortage of pulses and oilseeds

in the cultiva high gi may nee especia

pi-digg deline

95 771

STOLL THE

M 100 二十十二 100

The same

HOTEL TO THE

Talder splin

Market action

13 Tabilita

Awes of Early Amos

yes things

Garage States of

puntry, and the inherent nature of risk involved in the ion of pulses and oilseeds, the present policy of maintain is income parity in favour of wheat vis-a-vis gram and oi to be revised. There is also a need to improve the data for the union areas of the country ly in the area of costs of production of different crops.

otives and housiver carais committee with each other PRINTER RESPONSE VALUE OF THE PRINT S PRINTED TO THE TERM OF THE PRINTED TO THE P

the crop and release as a fill subject of the goad was

off was abused the greath of accress to our still directly at

the confectional order than relative to the conceptors in

THE TRUE COURT SMAYOR ODECS CHANNESS OF THE THANK FOR THE TOP IN

is to see that the text of the manufacture of the seems which is the seems of the s

as more steamed branches and a steam more set not

ALL MENT SHIP SHAFTY CERTAINS HOLD

to the day was the first with the state of the day of t

to the Operation troubles reporter afterward there this no shreet

United the second of the second secon

efauther eat of accompany of accompany

on the cold of the cold of the same of the cold of the

#### CHAPTER V

# RESPONSIVENESS IN PAKISTAN'S AGRICULTURE:

ION

tent of farmers' response to relative price changes is a subject in many developing countries. Governments in itries have extensively used agricultural pricing policies east three functions (i) to accelerate the growth of itput; (ii) to accelerate or decelerate the growth of an in accordance with the priorities of the government; (iii) keted surplus of food crops to ensure better availability of ban areas of the country; and (iv) to improve terms of trade ural sector in relation to the industrial sector. These however, rarely compatible with each other. Market supply inter-crop substitution depends on inter-crop price the growth of aggregate output depends on the variations aral price index relative to the variations in the prices of ems and/or goods purchased by the farmers for consumption fore, the particular pricing policy to be followed depends s set forth by the government and empirical results of erent aspects of agricultural pricing. officing colicles

to formulate an effective pricing policy, reliable empirical the degree of price responsiveness of supply to the relevant is necessary. In this context three types of supply response

a stress of the country; what (iv) to imprise leads of the

worth, herely repetitionality give a time; the account

al states he killed to the analysis all sector is the

be distinguished viz. the supply response of the agriculneed as a whole, the supply response of individual crops and th sect response of marketed surplus. The main purpose of this pay supp 'eview and provide empirical evidence on these three types is t response. This will give policy makers an adequate under supp ig of the direction and magnitude of the effects (intended stan ided) of agricultural pricing policies. This chapter revi unin :ults of the empirical work on price responsiveness for Pa the ture. It also provides estimates of supply response of agri ite output to lagged terms of trade for agriculture and ot aggr ere areas decends on the porent shif

The second section of this chapter presents information resp siveness of aggregate output while the third section is co with creage response and the fourth with the responsiveness of mark ed surplus. The fifth section contains a summary of the m conc sions and highlights the areas for further research.

to apidential as principle

lot hores and in the

#### 2. AGGREGATE OUTPUT RESPONSE

The price responsiveness of aggregate output has receiv erable attention in the literature1. The knowledge of suc cons se is important for the formulation of policies regarding resp tion of public resources between the agricultural and the a 110 Itural sectors. It is possible to find negative or insigni agri pric responsiveness of aggregate output even in situations when dual crops exhibit significantly positive responsiveness t indi nts in relative prices. This aggregation problem is inhere move See for example [50], [51] and [84].

in any econometric exercise of this kind. There is, thus, a need to study price responsiveness at fairly disaggregated levels in order for the estimates to be used for effective policy making. It may be noted, however, that in a situation where one or two crops dominate the total agricultural output, the distinction between the responsiveness of aggregate and individual crops may not be so relevant. A number of addit. ional points need to be considered in the formulation and estimation of supply functions for the agricultural sector. In land scarce areas where there is multi-crop agriculture, the elasticity of supply of land is expected to be very low. Responsiveness of aggregate output in these areas depends on the potential for improving the yield through intensive application of current inputs and use of new techniques of production. In land abundant areas this physical constraint on price responsiveness is not present. Aggregate supply function based upon data from different regions need to specifically take this into account. Secondly, the availability of transport facilities as well as the level of monetization and commercialization of the region need to be explicitely considered for a discussion of price responsiveness to be meaningful. Unfortunately very little work has been done in this area and this is due largely to the dearth of necessary data in Pakistan.

Most of whatever little work has been done in Pakistan on price response relates to individual crops. The study of aggregate output has been very limited. Wizarat / 91 / conducted a study to investigate the role of factor productivity in accelerating the growth of aggregate agricultural output. She calculated a total factor productivity index as the ratio of the value added index to the aggregate input index. She found that stagnation in the agricultural sector

during the 50's was mainly due to the use of old technology, constraints on vital inputs and the disincentive effect of various government pricing policies.

In the absence of alternative empirical evidence and in order to get a rough idea of the effects of terms of trade for agriculture on aggregate output we have  $\epsilon$  imated the following equation. The equation has been estimated using the available terms of trade series for the years 1951-52 to 1983-84 presented in Chapter III.

$$Q_t = -13.72 + 0.75 Q_{t-1} + 0.28 P_{t-1} + 0.07 W_t + 1.12 T_t$$

$$(6.05) (1.55) (2.81) (1.40)$$

Adjusted R<sup>2</sup> = 0.98 ... annual and anul-augman set

Where the figures in paranthesis are estimated t ratios and

Q+ = Index of agricultural output; The second

P+\_1 = Lagged terms of trade for agriculture;

Wt = Index of water availability calculated as percentage of irrigated area in total cultivated area;

T = Time trend variable; and

 $Q_{t-1}$  = Lagged dependent variable.

The equation explains about 98 percent of variations in the dependent variable. The estimated results show that terms of trade movements have a positive effect on the growth of agricultural output. The short-run elasticity, calculated at mean values of the output index and the terms of trade index, is 0.18. The long-run elasticity is 0.73. These values of elasticities are small but suggest that farmers are responsive to changes in prices. These estimates should only be taken as rough indicators of the price responsiveness of aggregate

output. A number of alternative specifications in terms of choice of explanatory variables and lag structures were tried. However, these were rejected on the basis of low explanatory power and non-significance of the estimates. In the selected equation the water availability variable also turned out to be significant at the 95 percent confidence level, indicating that pricing policy is not the sole instrument to foster agricultural growth. The time-trend variable is also positive and significant but at the 85 percent confidence level indicating the role of technological change in increasing agricultural output.

The longer-term benefits from any dynamic effects of price incentives depend largely on the adoption of new technology and the development of an appropriate institutional framework. Agricultural price policy, therefore, needs to take into consideration technological and institutional factors.

The equation explains about 30 percent if verietrons to trade entered veriable. The extracted results show shall take of trade entered have a posterior effect on the growth of derical turn output. The shart-rue electicity, calculated at mash values of the satisful this

and the terms of trace index, it was small the supplet that there

which the evaluation of the orthogram temperature of States and the States of the stat

### 3. RESPONSIVENESS OF CROP ACREAGE

Considerably more work has been done on the acreage responsiveness of different crops to relative price movements. Most of the recent work is based on the Expectations-Adjustment model developed by Nerlove. Cummings / 197 estimated the model for Pakistan to seek answers to the fundamental question as to whether farmers respond to price incentives while deciding about acreage to be allocated to a crop. He included seven major cereal and cash crops viz. wheat, rice, barley, cotton, rape and mustard seeds, sesamum and tobacco in his analysis. The model was estimated at the national and regional levels. The detailed results of the regressions are presented in Appendix Table A-5.1. The results at the national level indicated positive price responsiveness of acreage for all crops except two i.e. sesamum and tobacco. The estimated price coefficients for wheat and barley were not significantly different from zero. The coefficients for the other crops were, however, significant. Supply elasticities obtained from the estimated price parameters are shown in Table V.1. The short-run and long-run supply elasticities estimated by Cummings for cotton varieties and for rape and mustard seed were notably higher than those for the less market oriented cereal crops.

Cummings also estimated the elasticities based on data for ten
Divisions and for selected districts of Pakistan. His estimates of
short-run and long-run acreage elasticities by Divisions and Districts
and for all the crops are also presented in the Appendix Tables A-5.2 and
A-5.3 respectively. For rice, the small supply elasticity indicated
nationally was also found in most of the divisions and districts.

Statistically insignificant negative estimates were found in some cases, For wheat, the national estimates like most of the geographically disaggregated estimates are positive but insignificant. However, there are some estimates that are significant but negative. For the major cash crop, cotton, positive price responsiveness is indicated. The degree of responsiveness is greater for American cotton which commands higher market prices. At the district level the estimated price parameter was found to be negative but insignificant for Hyderabad which is one of the most important cotton growing districts. For mustard seed, a fairly large and statistically significant positive elasticity was indicated at the national level and at the division and district levels. For Sesamum, which is a minor crop, a negative but statistically insignificant relationship between prices and acreage was found at the the national level and in four of the six divisions. The negative but statistically insignificant relationship at the national level was not confirmed by the results of disaggregated analysis at the district level.

The pattern of acreage responsiveness that emerges from the study by Cummings can be classified into low response, medium response and high response groups for the long-run acreage elasticities in the ranges of zero to 0.1, 0.1 to 0.4 and 0.4 to 0.7 respectively. It is evident that the response of foodgrain crops falls in the range of the low or medium category. The response of sesamum and tobacco is also low. Commercial crops such as cotton (American) and rapeseed and mustard have the highest acreage elasticities.

the still area round in host of the divisions and districts

Falcon  $\int$   $\frac{237}{}$  estimated a simple model of price responsive for cotton and  $\frac{1}{}$  leat crops. The following estimated equation we reported for the cotton crop.

$$A_t = -4$$
 .1 + 16.7  $\frac{1}{(2.4)} = -\frac{Price \text{ of } Cotton_{t-1}}{P_{t-1}}$   $\frac{1}{\sqrt{P_{t-1}}}$ 

$$R^2 = 0.70$$

$$R = 25$$

Where

 $A_t$  = Pe centage change in acreage under the crop fr m period t-1 to t.

 $P_{t-1}$  La ged weighted price of rice, bajra, jowar, ma ze and sugarcane.

A similar equati n was estimated for wheat where the explanatory variable was the relative price of wheat. In order to analyse y response in cott 1, the percentage change in cotton yield was su tuted as the depa ndent variable in the above equation. The coefficient of lag. ed relative price was positive for both wheat a cotton crops. Ti : coefficient of determination was 0.70 for the cotton equation idicating that seventy per cent variation in t dependent variab e are explained by the model. Falcon estimated positive short-rt | acreage elasticity of 0.41 for cotton crop ( Table V.1). He e timated the acreage response function for the wheat crop, sepay tely for irrigated and rainfed (barani) areas. For irrigated are sabout 25 per cent of the variation in the dependent variabl was explained by the price of wheat relative the price of suga cane, lagged by one year. The estimate of pric elasticity was low and varied between 0.1 and 0.2. For rainfed areas relative price could explain very little of the variation in the dependent variable. In these areas acreage under the crop was found to be highly correlated with the rainfall variables. Falcon's \( \frac{7}{23} \) estimates of cotton and wheat short-run elasticities are consistent with the findings of Cummings \( \frac{197}{197} \) although the estimates pertain to slightly different time periods. On the whole, Falcon's analysis suggests that farmers behave rationally and react to the changes in price level.

Acreage response functions were also estimated by Gotsch and Brown \_31\_7. The authors estimated a traditional Nerlovian distributed lag model for wheat, rice, cotton and sugarcane. They estimated the following equation for Central Punjab for the period 1934/35 to 1964/65 and for the Pakistani Punjab for 1960/61 to 1975/1976:

$$A_{t}^{c} = a + b P_{t-1}^{c} + d A_{t-1}^{c} - e W_{t-1}$$

Where

b = short-run supply response to price

 $\frac{b}{1-d}$  = long-run supply response to price.

e = short-run supply response to water.

 $\frac{e}{1-d}$  = long-run supply response to water.

 $A_t^c$  = Acreage under a crop in period t.

Pt-1 = Crop price in period t-1 relative to the weighted average price of production substitutes.

At = lagged dependent variable

Wt-1 = water availability in period t-1.

The implicit assumption in estimating the above equation is that the decisions of farmers about the acreage to be brought under each crop are based on the expectation of prices, which are formed on the basis of previous experience. The results of this study are also presented in Table V.1. The authors estimated separate equations for irrigated and rainfed areas for the wheat crop. The price elasticities for both the areas were low and possibly due to the subsistence nature of the crop. There were marked difference in price responsiveness between the irrigated and the rainfed areas. The price elasticity for cotton was about 0.3 which is slightly lower than the previous estimates of 0.4 found by Falcon  $\sqrt{23}$ . Sugarcane had the highest price elasticity of 0.44 indicating that farmers are highly sensitive to pricing policies while deciding about the acreage to be allocated to the crop. The authors concluded that there was substantial absolute acreage response due to changes in relative prices.

Gotsch and Falcon / 33 / analysed the role of new technology and pricing policies by applying a linear programming model. The authors noted that farmers respond to relative net revenue rather than to relative price because they interd to maximize profit. This distinction seems to be more important in situations where a new technology is introduced to foster agricultural growth. The authors derived(static) normative supply curves through linear programming by varying the price of wheat and calculating the net revenue associated with each price. The effect of new technology on output at current prices and the elasticity of farmers' supply response was traced by varying the price of wheat parametrically with alternative

TABLE V.1

CROP	PRICE	ELASTICITY	ADJUSTMENT	ADJUSTMENT COEFFICIENT	PERIOD	SOURCE
- Maria	Short- run	Long- run	Price	Area	2 11 1 2 11 1 2 11 1 2 1 1 1	
Rice	0.12	0.17	06.0	69.0	1949-68 1934-65 1960-76	Curminas Sostch & Brown Gostch & Brown
Wheat	0.10	0.22	0.70	0.45	1949.68	Curminos
	0.10 to 0.20 0.07	0.21			1933-58	Falcon
Wheat (Irrigated)	90.0	0.31			1934-65	Gostch & Brown
Wheat (Rainfed)	-0.09	-0.18			1934-65	Gostch & Brown
Wheat (Rainfed)	0.07	0.35		10.1	1960-75	Gostch & Brown
Barley	0.03	0.02	0.85	1.31	1951-68	Cumnings
(American)	0.40	3.62	1.10	0.86	1950-62 1934-65 1960-75	Cummings Sostch & Brown Bostch & Brown
Cotton (Desi)	0.41	0.28	1.10	1.49	1950-62	Cummings
Rape & Mustard	0.38	0.48	0.80	0.79	1951-67	Curmings
Seşamum	60.0-	-0.09	06.0	0.98	1921-67	Cumings
Tobacco	-0.13	-0.14	0.70	96.0	1951-67	Curminas
Sugarcane	0.14	1.57	r or	TO THE	The state of the s	Gostch & Brown

sets of assumptions about the technology and the availability of supplementary water. The study found that the new technology greatly increased the price sensitivity of the system. However, little could be accomplished without supplementary water.

It must be noted that most of the estimates of price elasticity discussed so far are less than unity and thus, inelastic by the usual definition of the elasticity concept. These estimates indicate an awareness and willingness on the part of individual cultivators to make changes between crops and to adopt a cropping pattern that tends to maximize the economic potential of the available resources.

defect surplies. Especies the direct approach where a relationship of catabilities, between the eartern decimal type of the crop. The statistical decimal worlded to desirate the crop. The statistical decimal value of the crop. The statistical decimal value of the crop. The statistical decimal value of the crops of the desirate to desirate the electricity of the content and the statistic of the content of the

# 4. RESPONSIVENESS OF MARKET SUPPLY

In Pakistan, like other developing countries, a large proportion of total output is retained for domestic consumption, feeds, seeds and payments in kind. A knowledge of the determinants of market supply is necessary for estimating the food availability for the urban sector and forecasting the required level of foodgrains imports in any one year.

In the case of crops which are almost wholly marketed the elasticities of output and market supply can be regarded as approximately the same. But in the case of those crops, a substantial part of whose output is retained by farmers, the responsiveness of the marketed surplus must be measured separately from the responsiveness of total output.

There are two basic approaches to estimating the price elasticity of marketed surplus. Firstly, the direct approach where a relationship is established, usually by regression analysis, between the marketed surplus and the relative price of the crop. The statistical data required to estimate the elasticity using the direct approach are extremely difficult to obtain. The elasticity is, therefore, often measured by applying the indirect approach. Various methods of indirect estimation have been suggested in the literature. A review of these methods shows that most of them are aimed at estimating the approximate range within which the elasticity may be expected to lie. Krishna \( \frac{1}{2} \) 50\_/ presented a framework for an indirect estimation of price elasticity of the marketed surplus of a subsistence crop. He started

with the simple notion that total output can be distributed betw consumption and sales. Thus, sales (M) will be identical to the difference between the output (Q) and the consumption (C), i.e.

$$M = Q - C$$

By differentiating (1) with respect to the price relative and pe ming a series of manipulations ne obtained an expression for the elasticity as:

$$e = rb_1 - (r-1) (g + mhk)$$

Where

e = Price elasticity of market supply

r = Inverse of the sale ratio i.e. M/Q

b<sub>1</sub> = Price elasticity of output of the crop

g = Price elasticity of consumption

m = Sales ratio (i.e. M/Q or 1/r)

h = Income elasticity of crop output

k = Ratio of the value of crop output to net inco

Krishma pointed out that not much reliance can be placed on such estimates of price elasticities of marketed surplus. He therefo suggested another method for estimating the elasticity. According this method the elasticity e is a product of the price elasticity output  $(b_1)$  and the output elasticity of the market supply  $E_{mo}$ .

Noshirvani and Behrman critically examined eq. 2 of Kri indirect method. Noshirvani incorporated an income effect and co Krishpa's estimate of the price elasticity as below:

$$e = rb_1 - (r-1) (g + hk + hk b_1)$$
 .....(

Behrman ..... also included other crops as a source of income and corrected the expression as

where  $b_2$  is the price elasticity of production of crops other than the subsistence crop. A comparison between Krishna, Noshirvani and Behrman shows the difference in their assumptions and derivation procedure. Therefore, the estimates of price elasticity from these three models will yield different results. The models of Noshirvani and Behrman are more relistic and involve fewer statistical problems. It is, therefore, preferable to use these two models while indirectly estimating the price elasticity of marketed surplus.

There are very few empirical studies, in the context of Pakistan, concerning the price response of market supply. A possible reason for this is the absence of time series data on marketed surplus and the small variance of food prices in any one year.

Qureshi \_ 78 / estimated eqs. (2), (3), (4) and (5) for three alternative values of m and r to compute plausible ranges of price elasticity of marketed surplus of wheat in Pakistan. The results of his estimation are summarised in Table V.2 below:

matrix to be an dentinary at least to appear on the deposit feet

volva es vellos entrescriprose en aperida elimporto

RANGES OF PRICE ELASTICITY OF MARKETED SURPLUS OF WHEAT

Eq. No.	m = 0.09 r = 11.11	m = 0.55 r = 1.81	m = 0.92 r = 1.09	$b_1 = 0.1 \text{ to } 0.2$ $E_{mq} = 1.273 \text{ to } 2.178$
	The second of			
2	Min. e 1.854	0.1899	0.102	mone winguity employer
	Max. e 4.141	0.064	0.240	tail tibe verberies
4	Min. e 2.143	0.090	0.009	entrelement articles.
	Max. e 5.229	0.603	0.340	and the "Add total sorigin
5	Min. e 2.321	0.103	0.023	the period beautiful tried on
	Max. e 5.437	0.620	0.242	
3	Min. e -	117 mg ne	ore it de	0.13
gh co're	Max. e -	- Vi -		0.44

Source: / 78 /.

The results suggest a positive price response of marketable surplus for likely values of the parameters used to estimate the elasticity. All of the models of indirect estimation showed estimates of the price response for market supply of wheat larger than the price response for wheat production. There may be various factors that have accounted for the positive value of the elasticity. Larger farmers, who are often assumed to behave 'rationally', may have positive elasticity due to a negative consumption effect arising from increases in food prices or the substitution of cheaper food items in consumption by peasants. Qureshi also estimated a market supply function for the wheat crop using cross-section data on marketable surplus. The results of his estimation are given in Table V.3. The results showed a positive price response for the entire sample i.e. owners and mixed categories of farmers. The response is negative for the tenants but the co-efficient is insignificant.

of production to be significant and greater than unity. This implies that marketable surplus for wheat will increase more than proportional to the increase in output. He estimated the price elasticity of marketable surplus to lie between 0.3 and 0.7. His results suggest that a price increasing policy will result in higher levels of marketable surplus through reduced family consumption and increased farm sales. The author admitted however, that the conclusions drawn were a 'first approximation' of the true picture, the nature and quality of data being the limiting constraints.

Another attempt to establish the link between the market supply and output was made by Khan and Chaudhry / 45 /. Since the marketable surplus is the surplus of production over the minimum consumption level, the resultant income is used for consumption and investment purposes. According to these authors the main determinant of marketable surplus is the level of output. They also expected rent to be an important determinant of marketable surplus and estimated the following equation;

$$M = 18.39 + 0.35 \times -0.21 R$$

$$(0.01) \quad (0.02)$$

$$R^{2} = 0.58$$

Where the figures in parenthesis indicate the standard errors of the estimated coefficients; and

M = Marketable surplus

X = Output of food per capita

R = Total rent payment per capita

The signs of the coefficients are consistent with the hypothesis. The output elasticity of marketable surplus at mean value's worked out to

be greater than unity (1.60). The model explained 58 percent of variation in the dependent variable.

The authors also tested the hypothesis that farmers market only that portion of their output which is enough to satisfy their minimum cash requirements. The hypothesis suggests that cash requirements of the farmers will be the main determinant of marketable surplus. In this connection the prices of food crops will be a relevant variable in order to determine marketable surplus. The following equation was estimated to test this hypothesis.

$$M = -12.14 + .35 X_0 - .19 R - .02 X_c$$

$$(0.01) (0.01) (0.02) C$$

$$R^2 = 0.58$$

Where the figures in parenthesis indicate the standard errors of the estimated coefficients; and

X = Output of cash crops per capita

 $\chi_0$  = Income from other sources

The signs of the estimated coefficients were found to be consistent with the hypothesis.

Most of the work on market supply responsiveness relates to the wheat crop while little or no work has been done to analyse price responsiveness of market supply for rice, maize, gram, pulses and other food crops. At present there is no time-series study available on marketed surplus of agricultural commodities as a whole. A positive step in this direction will be to plan studies in this area to establish the degree of generality of the findings established for other developing countries.

The main findings of the preceding analysis need no elaborate summary. However, some concluding remarks may be fruitful for a busy reader. The trends and phases of domestic economic incentives as measured by the net terms of trade for the agricultural sector and its various subsectors were analysed in the chapter on 'Terms of Trade and the Price Policy'. The impact that the domestic incentives may have had on the level and composition of agricultural output was the focus of attention in this chapter.

Shifts in the relative prices of crops were frequent and large.

The empirical evidence that the acreages of different crops vary systematically in response to intercrop price movements is abundant and shows that cropping patterns are a function of prices. The price elasticities for food crops are lower than those for cash crops, but even in these cases positive price responsiveness is indicated.

While acreage response functions have been subjected to detailed scrutiny, little work has been done on the output response. The elasticity of output with respect to price is the sum of elasticity of yield per acre and the elasticity of acreage. We have already noted that acreage elasticity is positive. If yields are sensitive to the relative prices of output and inputs, one should expect the output elasticity to be not only positive but larger than the acreage elasticity. Some evidence exists, in Pakistan, that farmers adjust the quantities of varyable inputs, especially modern inputs, in response to changes in

the input prices. However, no solid empirical work has been done neve validations palesarous or beal nor participating in this area. Need for a careful analysis of the output response to the relative output-input prices is obvious and urgent and would be of direct policy relevance.

Another area of research which is required on the agenda for further research is the likely effect of prices on the "effective" supply of agricultural products for the non-agricultural sector. Cash crops are marketed almost totally. The price responsiveness of the marketed supply needs to be estimated for different food crops. Studies on this issue are very few, only two studies were available. A possible reason for this is the absence of time series data on marketed surplus. The price elasticity of wheat for market supply was found to lie between 0.3 and 0.7. A significant relationship between marketed surplus and food output was also noted. A rapid increase in the marketed surplus of some food crops in Pakistan, notwithstanding positive elasticity of marketable surplus with respect to prices, is ultimately explainable in terms of technological factors behind the observed increased output. Relative price changes were not of a magnitude that could explain the market expectations and were not terefred. We may est make supply of foodgrains. output response function. The well-mated require show a roseline And

Although supply response, especially the acreage response functions, is a reasonably well researched area, no studies relating aggregate farm output to terms of trade are available for Pakistan.

Some researchers have noted in passing that they had tried to establish a statistical relationship between terms of trade and value added in agriculture. The results were dismal and contrary to a priori

Table A-5.1

Estimated Parameters for Price Responsiveness of Acreage at the National Level.

Crop	Time Period	Constant	Price	Lagged Area	Rainfall	Trend	R <sup>2</sup>
Rice	1949-1968	+963.7 <sup>a</sup> (1.39)	+15.03 <sup>b</sup> (2.11)	+0.310 (0.78)	+0.107 (0.32)	+55.59 <sup>c</sup> (2.21)	0.94
Wheat	1949-1968	+1636.0 (0.03)	+90.83 (1.03)	+0.310 (0.65)	+4.391 a (1.37) -	+120.7 (0.63)	0.89
Barley	1951-1968	+514.8 <sup>d</sup> (4.40)	+ 1.434 (0.38)	-0.308 (1.02)	+0.312 <sup>d</sup> (3.02)	-6.218 (2.17)	0.54
Cotton(American)	1950-1962	+1448.0 <sup>d</sup> (3.46)	+16.46 <sup>d</sup> (0.99)	+0.139 (1.89)	+1.057 <sup>b</sup> (2.60)	+122.96 <sup>c</sup>	0.82
Cotton (Desi)	1950-1962	+690.7 <sup>C</sup> (2.94)	+2.839 <sup>C</sup> (2.34)	-0.490 <sup>b</sup> (1.85)	+0.945 (9.24)	-10.53 (0.78)	0.41
Rape and Mustard	1951-1967	+363.2 (0.94)	+20.50 <sup>a</sup> (1.65)	+0.208 (0.55)	+0.366 (0.51)	-19.42 <sup>a</sup> (1.30)	-
Sesamum	1951-1967	+59.78 <sup>b</sup> (2.01)	-0.160 (0.46)	+0.021 (0.04)	+0.040 <sup>a</sup> (1.32)	+1.255 <sup>a</sup> (1.44)	0.31
Tobacco	1951-1967	+36.98 <sup>a</sup> (1.71)	-0.106 (0.72)	+0.100 (0.30)	+0.031 (0.50)	+7.005 <sup>d</sup> (3.87)	.074

Notes: Figures in parentheses are t-values.

a. 30 percent significance level
b. 10 percent significance level

c. 5 percent significance leveld. 1 percent significance level

Source: Cummings / 19 /.

-621-

Table A-5.2 Short and Long Run Elasticity Estimates based on Data for Divisions

Division	20.30	Rice	Wheat	Bariey	(American)	n Desi	Mustard	Sesamum	Tobacco
W (1821	Mex	Jan Marine	nate state	15.160					
Bahawalpur	SR LR		+9.08 +0.10	+0.48 <sup>a</sup> +0.51	+0.40 <sup>a</sup> +0.29	+0.60 <sup>c</sup> +0.44	+1.09 <sup>c</sup> +0.61	+0.30 <sup>c</sup> +0.42	-
Dera Ismail Khan	SR LR	<u>:</u>	+0.01 +0.01	+0.55 <sup>d</sup> +0.45	130 10 - 100	50 - 07	-1.94 <sup>C</sup> -1.15	1809	- 100
Hyderabad	SR LR		+0.02 +0.02	-0.50 <sup>a</sup> -0.28	0 0		-0.33 -0.11	0.74 <sup>d</sup> -0.51	2010
Khairpur	SR LP.	-0.11 -0.13	+0.57 <sup>b</sup> +0.63	-0.34 <sup>b</sup> -0.30	+1.84 <sup>d</sup> +3.54	+0.45 +0.51	+0.74 <sup>a</sup> +0.77	-0.20 -0.07	-
Lahore	SR LR		-0.09 -0.09	+0.20 <sup>a</sup> +0.34	+0.42 <sup>b</sup> +0.67	-0.05 -0.03	+0.63 +0.73	-0.99 <sup>d</sup> -1.05	-0.03 -0.03
Multan	SR LR	+0.08	-0.03 -0.04	+0.09 +0.06	+0.52 <sup>c</sup> +0.46	-0.57 -3.35	+0.81 <sup>d</sup> +0.47	-0.33 <sup>a</sup> -0.58	-0.37° -0.46
Peshawar	SR LR		0	+0.21 +0.22	214.8	+0.15 +0.21	-0.45 -0.38	92.518	+0.83 <sup>8</sup> +0.94
Quetta	SR LR	+0.11 +0.16	-0.26 <sup>a</sup> -0.27	+0.46 <sup>b</sup> +0.47	638.0 +30	(a) (a)	110 - +0 301	· 0301	0,419
Rawalpindi	SR	+0.76 <sup>d</sup> +38.0	-0.26 <sup>C</sup> -0.42	+0.19 +0.17	+0.26 <sup>a</sup> +0.72	+0.57 <sup>c</sup> +0.32	+0.11 +0.20	_ J	+0.31 <sup>6</sup> +0.34
Sargodha	SR LR	-0.34 <sup>a</sup>	-0.39 <sup>a</sup> -0.40	+0.14 +0.22	+0.23 <sup>c</sup> +0.16	+0.33 <sup>a</sup> +0.20	+0.94 <sup>c</sup> +0.94	+1.01 <sup>d</sup> +0.68	+1.10 +0.65

Notes:

Lack of a figure indicates no supply analysis was performed.
 Significance level of price parameter from which elasticity was derived.
 a. 30 percent
 b. 10 percent
 c. 5 percent
 d. 1 percent

Source: Cummings / 19 /.

Based on Data for Selected Districts

"E	districs		Short-run Elasticity	Long-run Elasticity	District	Short-run Elasticity	Long-run Elasticit
	1		2	3	4	5	6
Rice	7519194			-ukkur	-0.527.005	-0	
Nice	Dadu	10.4%	+0.28 <sup>a</sup>	+0.72	Larkana	-0.06	-0.09
	D.G. Khan	10 110	+0.02	+0.03	Sahiwal	+0.08	-0.24
	Gujranwala		-0.36	-0.32	Muzaffargarh	_ +0.56 <sup>a</sup>	+1.33
	Gujrat		+0.91 <sup>d</sup>	-10.1 - Lames	Sanghar	+0.47	+1.37
	Hazara	10.650	+0.47 <sup>b</sup>	+0.26	Sheikhupura	-0.24	-0.22
	Hyderabad	10 36c	+0.42 <sup>a</sup>	+0.23	Sialkot	+0.82 <sup>C</sup>	+1.00
	Jacobabad	-916479	-0.06	-0.15	Sibi +0 58	+0.13	+0.25
	Khairpur		+0.80	+0.86	Sukkur	+0.05	+0.06
	Kurram		+0.20 <sup>b</sup>	+0.33	Tharparkar	+0.81 <sup>a</sup>	+0.65
	Lahore		+0.18 <sup>c</sup>	÷0.69	Thatta	+0.49 <sup>d</sup>	+4.90
Meat							
	Attock	140.17	-0.33 <sup>d</sup>	-0.67	Loralai	-0.16	-0.19
	Bannu Kaula	-0.00	-0.05	-0.06	Lyallpur	-1.22 <sup>c</sup>	1.03
	Chagai		+0.94 <sup>b</sup>	+1.52	Mianwali	+0.49 <sup>a</sup>	+0.35
	Dadu	190,42	+0.52 <sup>C</sup>	+0.52	Sahiwal	-0.18 <sup>d</sup>	-0.15
	D.G.Khan	4444	-0.05	-0.04	Multan	+0.01	+0.02
	D.I.Khan		-0.19	-0.12	Muzaffargarh	0	0 22 0
	Gujranwala		-0.13	-0.17	Naviabshah	+0.46ª	-0.43
	Gujrat	0.90	-0.13 <sup>a</sup>	-0.20	N.Waziristan	+0.01	+0.01
	Hazara		-0.15 <sup>d</sup>	-0.09	Peshawar	-0.06	-0.24

(Continued)

1	2	3	Ą	5	6	
Wheat-contd.		ALC: N				
Hyderabad	-0.90	-1.03	Quetta	-0.09	-9.11	
Jacobabad	+2.60 <sup>d</sup>	+3.38	Rawalpindi	-0.05	-0.06	
Jhang	-0.51 <sup>d</sup>	-0.46	Sanghar	+0.04	+0.05	
Jhelum	+0.05	+0.09	Sargodha	-0.17 <sup>b</sup>	-0.20	
Khairpur	+0.62 <sup>a</sup>	+1.17	Sheikhupura	-0.07	-0.06	
Kohat	-0.19	-0.19	Sialkot	-0.08	-0.07	
Kurram	-0.02	-0.02	Sibi	+9.13 <sup>a</sup>	+0.25	
Larkana	+0.17	+0.15	Zhob	+0.05	+0.04	
Lahore	-0.07	-0.08				
Barley				That the		
Attock	-0.06	-0.08	Mardan	-0.40 <sup>b</sup>	-1.74	
Bannu	+0.90 <sup>a</sup>	+0.85	Mianwali	+0.26	+0.27	
D. I. Khan	+0.47 <sup>a</sup>	+0.28	Sahiwal	+0.02	+0.02	
Gujranwala	+0.32 <sup>c</sup>	+1.10	Muzaffargarh	+0.01	+0.01	
Gujrat	+0.55 <sup>d</sup>	+2.75	Peshawar	+0.09	+0.21	
Hazara	+0.16 <sup>b</sup>	+0.17	Quetta	-0.04	-0.05	
Khairpur	+0.04	+0.11	Sheikhupura	+0.43 <sup>d</sup>	+0.43	
Kohat	+0.81°	+0.63	Sialkot	+0.08	+0.06	
Lahore	+0.40 <sup>a</sup>	+0.54	Sukkur	-0.37	-0.36	
Loralai	+0.48	+0.50	Thatta	-0.31 <sup>a</sup>	-0.36	
Cotton (American)						
Gujrat	+0.26ª	+0.72	Mul tan	+0.43 <sup>a</sup>	+0.35	
					(Continuedp-	129

1	2	3	4	5	6
Cotton(American) - c	ontd.				
Hyderabad	-0.18	-1.13	Nawabshah	+0.94 <sup>c</sup>	+0.82
Jhang	+0.18	+0.18	Sanghar	+0.90 <sup>a</sup>	+0.07
Khairpur	+1.44 <sup>C</sup>	+4.50	Sargodha	+0.40 <sup>b</sup>	+0.34
	+0.74 <sup>c</sup>	+1.72	Sheikhupura	+0.34 <sup>a</sup>	18.0+
Sahiwal	+0.65 <sup>c</sup>	+0.81	Tharparkar	+0.02	+0.02
Cotton (Desi)			E 8 2 5 19		
Gujranwala	+0.64 d	+0.94	Lahore	-0.41	-0.95
Khairpur	+2.160	+2.14	Nawabshah	-9.22	-0.28
Sahiwal	-0.66	-3.67 -	Sialkot	+0.56°	+0.61
Rape and Mustard				1 3 3 3	
Attock	-0.27	-0.64	Lyallpur	+0.440	+0.27
Dadu	+0.28	+0.31	Mianwali	+0.0.	+9.80
D.I.Khan	-0.50°	-0.36	Nawabshah	+0.71 <sub>b</sub>	+0.74
Hyderabad	-0.57	-2.48	Sanghar	-n.84 <sup>b</sup> .	-4.94
Jacobabad	-0.52	-2.17	Sibi	+2.74b	+5.71 +1.29
Khairpur	-0.06	-0.06	Sukkur	+0.75 <sup>2</sup> -0.87 <sup>a</sup>	-0.88
Larkana	+1.39	+2.44	Thatta	-0.87	-0.65
obacco		1 1 2 6 6			
Attock	+0.16	+0.21	Mardan	+9.46	+0.46
Gujranwala	-0.19 <sup>a</sup>	-0.17	Sahiwal	-0.19°	-0.12
Gujrat	+0.39 <sup>¢</sup>	+0.36	Peshawar	+1.29°	+2.30
Lahore	-0.26	-0.27	Sheikhpura	+0.27	+0.22
Lyallpur	+0.11 <sup>c</sup>	+0.08	Sialkot	-0.19	-0.15

Notes: Significance level of price from which elasticity was derived:

Source: Cummings / 19 /.

<sup>(</sup>a) 30 percent; (b) 10 percent; (c) 5 percent; (d) 1 percent. Sahiwal was formerly known as Montgomery

#### CHAPTER VI INTERSECTORAL TAX EQUITY

#### I. INTRODUCTION

The principle of equity occupies a central place in the desired attributes of taxes. The application of equity criterion is easier when two entities of roughly equal economic capacity are considered in the analysis. The equal treatment of equals in taxation is based on sound ethical principles. The difficult problem arises when two economic entities with unequal economic capacities are to be subjected to the analysis of fair distribution of taxes. Although on the vertical scale the meaning of equity has evaded a precise definition, there is a general consensus that taxes which redistribute incomes from the rich to poor are generally preferable.

To estimate the redistributive impact of the actual tax system, two sets of estimates are needed. First, the incidence of different taxes is to be measured. Second, the desired level of progression is to be specified. Both aspects of this problem are inherently controversial. A theory of empirical estimates of the distribution of tax burden has been evolved nevertheless. The traditional application of this theory has been in the context of allocation of different taxes to different income groups. The theory has also been applied in the context of measuring state tax burden in measuring international tax burdens and in

measuring sectoral tax burdens with in different countries. In discussions of tax reform in Pakistan the proponents of increased agricultural taxation often base their case on the View that agricultural sector is "undertaxed". Previous empirical work in Pakistan on interesectoral tax burden is scanty, outdated and generally weak as far as the method of analysis is concerned. The study by Kazi /43 / is an exception. She has estimated sectoral tax burdens for the years 1972-73 to 1979-80. Her study also contains a comprehensive review of previous work done in this area. Notable work viewed by her is the studies by Javed Hamid entitled "Suggested Approach to Agricultural Taxation Policy in Pakistan", The Pakistan Development Review, Vol.X. No.4, Winter 1970, Ghaffar Chaudhry "The Problems of Agricultural Taxation and an Alternative Solution", The Pakistan Development Review, Vol. XII, No.2, Summer 1973, Sarfaraz Qureshi, "The Problem of Agricultural Taxation and an Alternative Solution: A Comment", The Pakistan Development Review, Vol.XII. No.4, Winter 1973, and Jeetun A. "Incidence of Taxes in Pakistan" Karachi. Applied Economic Research Centre 1978. This chapter attempts to measure the relative tax burden of the agricultural sector for the years 1972-73 to 1983-84.

#### CONCEPTUAL FRAMEWORK AND ITS LIMITATIONS

The theory of equitable distribution of tax—amongst various economic entities can best be described in a series of steps in which it was evolved. Equity is generally defined in terms of the equality of tax burden. The tax burden on any economic unit is measured by the ratio of taxes to taxable capacity. The taxable capacity of an individual unit is measured by its economic capacity (income, wealth, consumption) minus subsistence requirement. In view of the difficulties in measuring subsistence requirement, early economists defined tax burden as the ratio of taxes to income. Equal tax-income ratios for two individuals implied equitable taxation according to this definition. Ghandhi \( \sum 30 \sum has pointed out certain limitations of this measure and Presented a modified measure of intersectoral equity. Before we list the limitations of the traditional measure of tax burden, it should be pointed out that for two units with roughly equal taxable capacity the measure gives valid results.

The most serious limitation of measuring tax burden by the ratio of taxes to income is that it endorses proportional taxation as the "norm" or "line of equal burden". The selection of norm for the equitable taxation of individuals with unequal taxable income is a

thicky and at best an arbitrary exercise. Conclusions concerning the "fair" tax treatment of unequals vary according to the principle of sacrifice and the income-welfare relationship / 26 /. Both the Principle of equal sacrifice and that of proportional sacrifice can lead to regressive, proportional or progressive taxation depending on the slope of marginal-utility-of-income curve. Theoretical complexities aside, there is a general consensus that some progression in taxation is socially desirable.

This limitation of the traditional measure was corrected by Frank <u>/ 27/</u> and Richard Bird <u>/ 13/</u>. Both suggested a measure of the tax burden which gives greater weight to income than to taxes. The measure is defined as the ratio of taxes, as a percentage of personal income (disposal income in case of Bird's measure), to personal income per capita. This measure squares the per capita income and thus introduces progression.

Henry Aaron / 1/ and Ghandhi /30/ have generalized the measure suggested by Frank and Bird. Both show that income can be raised to the power of any specified number ranging from one (as in the traditional measure) to two (as in Frank's measure). Ghandhi has recommended a moderate level of progression by raising taxable capacity to the power of 1.5.

Another limitation of the traditional measure of the tax burden is that it assumes that equal incomes have equal subsistence requirements and that the subsistence requirement increases in proportion to income. The assumption that equal incomes have equal subsistence

requirements may be roughly valid although a purist may quarrel with this assumption. The assumption that subsistence allowance should increase proportionally with income cannot be justified on a priori grounds. A separate estimate for the subsistence requirement for both agricultural and non-agricultural sectors must be incorporated in the measure for relative tax burden.

The third limitation of the traditional measure of tax burden is that the measure of economic capacity has been measured by income only. In the literature on taxable capacity, income is considered as the most important index of taxpaying capacity. Other factors, like permanent income, wealth or consumption could be a substitute and/or complementary indices of taxable capacity. In the context of sectoral tax burden, distribution of income and wealth become additional sources of taxable capacity. The incorporation of these other measures of taxable capacity raises serious index number problems and also the problem of choice of appropriate weights for the different components of taxpaying capacity.

The limitations of the traditional tax burden measure discussed so far and their correction in the measure used in the present study makes the measure theoretically more relevant. Its empirical application is subject to the data weakness present in any practical situation. Before we present the measure and its empirical values in the two sectors in Pakistan, we briefly list the criticisms of any tax burden measure.

These criticisms relate both to the theory of tax incidence and the theory of interpersonal comparison of utilities.

Firstly, it is asserted that the concept of tax burden is meaningless because this assumes, for implicit comparison, the absence of government. In this context it is helpful to distinguish between absolute and relative tax burdens. Taxes are levied mainly to offset government expenditure. All individuals in a society derive benefits from the government. It is possible that none of the individuals may bear any net burden due to the fiscal operation of the government. Even when there is no absolute burden involved, if two individuals have equal taxable capacity but different tax payments, a valid case for tax inequity on relative burden considerations can be made. It must be remembered that the tax burden and not the fiscal burden is the main objective of study in this chapter.

Secondly, the computation of relative sectoral tax burden inherently involves comparison of interpersonal utilities. The comparison of such utilities is now acknowledged to be without operational content. To the extent that public policy demands making such comparisons, the basic limitation must always be kept in view.

Lastly, the incidence of taxes on different sectors ignores the effects of the tax system on the important variables in the economy.

These are only some of the criticisms of tax burden estimates. In view of such limitations it is a moot question whether the results are meaningful at all. The answer depends on one's view of economic analysis. If one considers economic analysis capable of providing

a feasible answer, then the exercise can be useful. In any case, these limitations qualify the results presented in later sections of the chapter.

#### 2.1. Tax Burden Measure

As mentioned earlier, the traditional measure of the tax burden was defined as the ratio of taxes to income. Its limitations are:

(1) proportionality as the norm of equitable taxation, (2) non-accounting of subsistence requirements, and (3) non-accounting of measures other than income as arguments of taxable capacity. The modified measure of tax burden that takes account of subsistence requirements and progression is as follows:

$$B_1 = \frac{t}{(y-s)^{e_0}}$$

Where  $B_1$  is the first approximation measure of the tax burden, t is per capita tax, y is per capita income, s is per capita subsistence requirement and  $e_0$  is the desirable degree of progression.

The first approximation measure of the tax burden can be modified to take account of factors other than income as arguments of taxpaying capacity. The second approximation measure of the tax burden is as follows:

$$B_2 = \frac{t}{f / \overline{y} - s, \overline{w}, w, i / e^0}$$

where additional factors of wealth per capita  $(\tilde{w})$ , wealth inequality (w) and income inequality (i) are included in the denominator of the measure.

## 2.2 Intersectoral Equity

The inequity in taxation is measured by the inequality in the tax burden on two economic units. Our main objective is to test the hypothesis of the under-taxation of the agricultural sector. To do so, separate estimates of the tax burden for each of the two sectors need to be computed. If the tax burden on agriculture is less than the tax burden on non-agriculture, the undertaxation of agriculture is empirically established. In notations, if A and N stand respectively for agriculture and non-agriculture and C for the taxable capacity, the agricultural sector—is undertaxed whenever

$$\frac{t_A}{C_A e_0} < \frac{t_N}{C_N e_0}$$
 or  $\frac{t_A}{t_N} < \frac{C_A}{t_N} e_0$ 

In words, the agricultural sector is undertaxed if its relative tax payments are less than its relative equity-adjusted taxable capacity.

# 3. SECTORAL TAX EQUITY IN PAKISTAN: FIRST APPROXIMATION

The first approximation measure of the tax burden on any sector requires estimates of (1) per capita taxes after taking full account of tax-shifting possibilities, (2) per capita income, and (3) per capita subsistence requirements. To analyse the relative tax burden on agriculture and non-agricultural sectors in Pakistan first, we develop the required information and then present the results regarding sectoral tax equity.

### 3.1 Shifting of Taxes in Pakistan

The theoretical considerations and the special institutional context in Pakistan that have some bearing on the incidence of different taxes need to be briefly reviewed. In conformity with the general practice in empirical studies on tax burden studies, assumptions on shifting about different taxes are made as a prelude to allocation of taxes to different sectors. Partial equilibrium analysis does not give a general answer with regard to the extent of shifting of any particular tax. It lays down conditions in terms of supply and demand elasticities. As would be noted in the ensuing discussion, the assumption in the general literature that commodity taxes are shifted forward to consumers is not quite valid in Pakistan, at least during the earlier eras of its economic history. The general procedure we adopt is to discuss the shifting of each of the major taxes in Pakistan. Given the controversial nature of assumptions about shifting of some of the taxes, the analysis is carried out for three alternative sets of shifting assumptions, because sign and a great state to big signois and the model

#### 3.1.1 Land Tax

tax collection by federal and provincial governments. The tax is applied at fairly low rates — generally in the range of 1-2 per cent of income. The tax is predominently levied on agricultural land. It is assumed that the tax is borne by the producers as there is no evidence to indicate forward shifting to the consumers of agricultural products. It has been argued that land taxes induce an increase in

Hocated entirely to the non-agricultural sector.

total tax levels.

conferiblitor to

tales to different sectors.

Marketable surplus in subsistence economies by forcing the taxpayer to earn enough money income to pay the tax / 87 /. If this argument is valid empirically, the land tax would cause a decline in produce prices suggesting that the landowners would be burdened by more than the amount of tax collected. The entire proceeds of the land tax are allocated to the agricultural sector.

#### 3.1.2 Income Tax

The income tax in Pakistan is levied only on incomes earned in the non-agricultural sector as agricultural incomes are exempt from this tax. It contributes about 10 per cent to the total tax revenue.

Economic theory lays conditions in terms of income and substitution effects to determine the effects of the tax on total earned income.

These effects are not verifiable from available data. Most income tax payers cannot vary their hours of work as a response to the changes in tax rates. Survey data, in the context of professional people who can vary their hours, gives ambiguous results on the extent to which they change their effort levels and/or prices / 25\_/. It is assumed that income taxes are borne by the income receiver. The proceeds of the tax are allocated entirely to the non-agricultural sector.

# 3.1.3 Company Tax

The company tax constitutes about 3 to 4 per cent of total is tax collections in Pakistan. The incidence of this tax has been subjected to a continuing controversy on both theoretical and empirical levels in the context of both developed and developing countries. In view of this controversy, it is fortunate that the tax is a minor contributor to total tax levels.

The profit maximizing firm has no incentive to shift the tax as the profits are maximized at the same price-output mix with or without the tax. In fact, whether the firms are profit maximizers or they follow the pricing rules dictated by the maximization considerations is open to question. Studies on the pricing rules actually practiced indicate that pricing according to a mark up over average cost is the frequent practice.

The empirical studies of the econometric variety on the shifting of corporation tax were pioneered by Musgrave-Krzyzanic model. This model has been tested for U.S.A., India and other countries / 52/. Evidence of forward shifting to a substantial extent is indicated by these studies. However, these findings have been contested on methodological grounds. Ghandhi analysed three industries with divergent market structures in India and found no evidence of forward shifting / 28 /.

The situation in Pakistan has not been analysed in any detail.

Azfar has related the structure of industries to share of taxes paid and the pricing rules followed in different sectors in Pakistan \_\_\_\_9\_\_\_. The conclusion reached by him is that very little forward shifting is to be expected in Pakistan. We argue later that the special conditions in Pakistan impede the forward shifting of commodity taxes. These conditions give an additional reason for the non-shifting of company taxes in Pakistan. It has, therefore, been assumed that the entire proceeds of the company taxes fall on the non-agricultural sector.

shreeten and the state of the same of the

#### 3.1.4 Export Duties

Taxes on exports in Pakistan are levied mainly on agricultural exports. Cotton and jute were the main two items prior to 1971/72. The incidence of the taxes is a matter of relative values of supply and demand elasticities. In the case of jute, Pakistan was a dominant seller. A case could be made that the tax was shifted to foreign buyers. For other commodities, the tax is likely to be on the domestic economy. We assume that the export duty on all agricultural goods was borne by the agricultural sector while the duty on the manufactured goods was borne by the non-agricultural sector.

#### 3.1.5 Imports Duties

Import duties and sales tax on imports are the second largest source of revenue contributing about 40 per cent to the total tax revenue. It used to be the dominant source in the early 1950s. The theoretical and empirical literature, in Pakistan, on the incidence of the import taxes has maintained that the duties are not shifted forward to the consumers. The theoretical argument for the non-shifting of the import duties was made by the author about two decades ago / 56 //. The main message of the argument is that imports are licensed and the quantity of allowable imports is fixed. The market clearing prices for imports are high enough to guarantee abnormally high profits net of taxes and the costs of imports. Under this state of affairs, if the rates of import duties are increased the duties come out of the high profits. The hypothesis has been tested twice in Pakistan once in the mid and again in the late sixties / 75,4//. The mark-up of domestic prices over costs

including import taxes are generally quite high. The mark-ups also show a direct relation with the stringency in licencing for different kinds of imports. By the mid-sixties, the constraint on total allowable imports had been relaxed relative to the early fifties. This relaxation of the constraint occurred due to improved export performance and larger aid flows beginning in 1959/60. In years when the level of imports was relatively high, forward shifting of import taxes is expected.

Beginning in 1959/60, some imports under Bonus Voucher Scheme were allowed. In case of such imports, forward shifting of import taxes is a valid assumption. Due to the conflicting evidence and considerations, alternative shifting assumptions for import taxes are made in the analysis.

#### 3.1.6 Domestic Indirect Taxes

The taxes on commodities and transactions are imposed by the provincial and federal governments in Pakistan. The taxes imposed by provincial government are mainly on liquour, tobacco, narcotics and motor vehicles. These constitute about 10 per cent of the total tax revenue. The taxes imposed by the central government on account of excise and sales tax on domestic goods is the dominant source of tax revenue. The shifting of taxes is a matter of relative values of supply and demand elasticities. The usual assumption in the literature on the incidence of these taxes is to assume full and forward shifting of these taxes to consumers. An argument exactly parallel to the one made for import duties has been made in the case of excise duties. Radhu regressed per cent change in domestic taxes on percentage change in the domestic prices for 33 commodities in Pakistan. He found no evidence of forward shifting of the taxes in Pakistan / 807. This evidence can

support the assumption that domestic commodity taxes are not shifted to consumers. Azfar has critically examined the methodology of Radhu and has shown that if one separates the commodities that yield about 90 per cent of tax revenue, the data used by Radhu indicate forward shifting / 9.7. Azfar cites some evidence relating to some of the major revenue-raising commodities in later years to indicate forward shifting. Similar evidence on the forward shifting of taxes is found in the studies by Mohammad Irfan, "Shifting and Incidence of Indirect Taxes on Tobacco and Petroleum Products in Pakistan", The Pakistan Development Review, Vol. XIII, No.1, Spring 1974 and Bilquis Naqvi "Shifting of Indirect Taxes: A Further Study", The Pakistan Development Review, Vol. XIV.No.2, Summer 1975. However, these studies were limited in scope and coverage. Based on the contradictory evidence, we make alternative assumptions about the shifting of central indirect taxes in Pakistan. In any case, we assume that provincial indirect taxes are fully shifted to consumers. This is based on the nature of commodities taxed that show high inelasticity of demand.

As must be evident by now, there is a consensus on the nature of shifting for the land tax, income tax, export taxes and provincial indirect taxes. These taxes are, however, only an insignificant proportion of the total tax revenue. Assumptions regarding the shifting of these taxes are common to all three alternative sets of estimates for the intersectoral allocation of taxes. In the first alternative, all federal indirect taxes are assumed to rest on producer/seller or importer. In the second alternative, 25 per cent of the import taxes, sales taxes and surcharges and 50 percent of the federal domestic indirect taxes are shifted to the ultimate consumers. In the third alternative, all federal indirect taxes are shifted to the consumers. The first alternative is more realistic in period of acute shortage and tight licensing of imports of consumer. Intermediate and capital goods. The second alternative is valid for most of the early years of our study. The third alternative is comparable to studies for other

countries and represents the state of affairs under a hypothetical alternative situation that might have existed under a free economy. This alternative may be a realistic reflection of the last few years of our study as restrictions on the trading sector have been largely removed.

#### 3.2 Allocation of Taxes of Sectors in Pakistan

Shifting assumptions are only the first step in the allocation of taxes to their final resting place. In addition, for taxes that are not shifted from the point of their impact, we need to know the sector classification of taxpayers. This is an easy task for land tax, income tax, company tax, export duties and import duties. For taxes that are not shifted but the taxpayers can be found in both sectors, the task is more complicated due to the lack of required data. For the commodity taxes that are shifted to consumers, we need to know the location of the consumers and their relative expenditure on the taxed commodities. In this sub-section, general aspects of the methodology of allocation of taxes to the two sectors is presented. The detailed method of analysis for each commodity is presented in the Annexure attached to the chapter.

#### 3.2.1 Burden of Non-agricultural Direct Taxes

These taxes include estate duty, wealth tax, gift tax and stamps and registration. Published official data provide total collections under each of these taxes. There is no easy way to allocate these taxes amongst the two sectors. We could distribute the tax revenue on the basis of relative sectoral population or relative

sectoral income or relative income of high-income groups in the two sectors. We have adopted the last method for the allocation of estate duty, wealth and gift tax as only high-income groups are subject to these taxes due to high exemption limits of these taxes. The ratio between the high incomes in the two sectors was worked out from the 1978/79 survey of distribution of rural and urban incomes in Pakistan  $\sqrt{687}$ . The share of the agricultural sector in the tax collection was about 20 per cent according to this method.

This methodology cannot be adopted for the allocation of stamps and registration taxes. These taxes are paid by poor and rich alike, though we would expect the rich people to incur the expenditure on these items to a greater extent. Agricultural population is arbitrarily assumed to pay about 15 per cent of the taxes.

# 3.2.2 Burden of Shifted Indirect Taxes on Sectors

The tax allocation on account of import duties, sales tax on imports, central excise duties, sales tax on domestic goods and provincial excise taxes are available according to the detailed commodity breakdown for most of the years. For the years that we did not have the commodity break-down, we used the ratio of sectoral shares in taxes for the most recently available year. The allocation of taxes on each commodity between two sectors is made according to the proportion of expenditure in the two sectors. The data on consumption expenditure patterns in the agricultural and non-agricultural sectors is approximated by the rural and urban consumption expenditure as estimated by a Survey conducted during 1979/80. The pattern revealed

in that year is assumed to hold for all years in our study. A detailed examination of the tax and survey data shows that the commodity coverage in the tax data is more comprehensive. The commodities were Commodities for which the divided into three groups. relative consumption expenditure data were available in the survey were treated quite easily. Secondly, the commodities for which data on similar consumption goods in the survey could be used were singled out and it was assumed that relative consumption expenditure on such commodities was the same between the two sectors as those for similar goods. Lastly, in the case of commodities for which no data were available or for which the incidence is likely to be diffused in the entire economic system, the sectoral shares were determined on the basis of relative expenditure on non-food items. For some commodities, the tax rate is graduated with respect to the price of the commodity in Pakistan. In such cases, the relative expenditure in the two sectors was adjusted as, in general, rich and ruban people spend more on high priced commodities compared with poor and rural people.

# 3.3 Estimation of Relative per capita Income and per capita Subsistence Requirement by Sectors.

The first estimate of taxable capacity requires the estimates of per capita income and per capita subsistence in agriculture and non-agricultural sectors. Both estimates are subject to data weaknesses and their precise magnitudes should be taken with the proverbial pinch of salt.

## 3.3.1 Per capita Incomes in the Sectors

The estimates of per capita sectoral incomes require estimates of sectoral income and sectoral population. Sectoral incomes are taken from the national accounts. Net value added by agricultural crops, livestocks, forestry and fisheries is considered as agricultural income. This procedure, widely followed by economists, is not quite valid because the agriculturists may derive part of their income from nonagriculture. Similarly, non-agriculturists may derive part of their income from agriculture. In the absence of data, we have assumed that the agriculturists' non-agricultural income offsets the nonagriculturists' agricultural income. Sectoral population totals are also estimated figures and may contain some errors. The total yearly population for Pakistan is taken from the C.S.O. estimates as presented in national accounts tables. Rural and urban population is estimated for 1972 and 1981 using census ratios in the respective years. The rate of growth of rural and urban population between 1972 and 1981 is estimated and applied to get yearly urban and rural population. Agricultural population is estimated by applying the ratio of agricultural labour force to the total labour force in rural areas. Nonagricultural population is estimated as a residual. The data on per capita income in agriculture and non-agriculture is presented in Table VI.2.

#### 3.3.2 Per capita Subsistence Requirements.

The estimation of subsistence requirements involves specification of the subsistence bundle of goods and services and its valuation for each sector. The specification of the subsistence bundle can be based on nutritional, cultural or social considerations. The valuation of this bundle requires prices of the commodities in agricultural and nonagricultural sector. Price data in both sectors can be generated but the difficulty in specifying a unique subsistence bundle is obvious. Ghandhi used an indirect procedure in estimating rural and urban subsistence / 30/. He equated this with the income of the 30th percentile in the income distribution for each sector. Since the per capita income in non-agriculture is more than twice the income level in agriculture, the per capita subsistence in non-agriculture comes out to be twice the subsistence level in agriculture. Part of the differential can be justified by higher urban prices and some additional items in the urban subsistence bundles (i.e. rented accommodation). However, awarding the urban sector twice the amount of rural subsistence does introduce a major bias in the estimation of relative taxable capacity. Ghandhi's case for the overtaxation of non-agriculture in India is probably an outcome of the generous allowance for subsistence in urban areas.

The estimates of the subsistence requirements for the agricultural sector are based on a recent study on rural poverty / 37/. The estimated rural poverty is assumed to provide 2550 calories to an adult per day and provides him with additional income to pay for non-food items like clothing, fuel and medicine to enable the farm family to live efficiently for farm work. The subsistence food bundle was

priced at the prices prevailing in rural markets during 1979/80 which came to Rs. 75/- per month. This value which is for an average adult was converted into a per capita value by using the average ratio of adult equivalent to population (0.8). The resulting figure was then converted into per capita income using the ratio of food expenditure to income ratio observed in rural areas of Pakistan of 0.55. The corresponding subsistence income for 1979 comes to Rs. 109/- per capita per month. We assume the subsistence requirements to remain constant through the period 1972/73 to 1983/84 in physical terms. We adjust the subsistence requirements for movements in yearly cost of living based on C.S.O.'s index of price changes for the industrial workers as no separate index was available for rural areas.

The subsistence requirements per capita in non-agriculture are estimated with reference to the subsistence estimates for the agricultural sector. The food intake in urban areas is normally somewhat lower than the rural areas based on the fact that urban work is less exacting. However, the price of food is higher due to marketing cost. We assume urban prices 20 per cent above rural prices. Urban subsistence budgets include some extra items for minimum level of living. After taking these factors into account, we estimate the subsistence requirement per capita in non-agriculture to be equal to Rs. 1637.61 during 1979/80. The requirements for other years are shown in Table VI.2. The index used for deflation is again based on the cost of living index for industrial workers.

the like Eiglidently for (Bill Abit - the Pholisterer (30g productors

The estimates for subsistence are admittedly very rough.

Separate cost of living indices for rural and urban areas would be an improvement. In general, the estimates seem to be consistent with the minimum wage standards set by the government for lowpaid workers in industry.

#### 3.4. The Tax Burden Estimates: The First Approximation.

We have built all the blocks required for estimating the sectoral equity of taxation in Pakistan. Table VI.1 shows the tax per capita in both sectors under three alternative sets of shifting assumptions specified in a previous sub-section. For each sector and for each alternative, tax-income ratio is shown. Per capita income in non-agriculture is more than twice the per capita income in agriculture as shown in the last column in Table VI.1. Due to the unequal sectoral incomes, tax income ratio cannot be used as a measure of tax burden. If the norm of equitable taxation is some progression in tax rates, high incomes are expected to pay more taxes than low incomes. As pointed out in Section 1, the specification of the "desirable" level of progression is difficult and strictly outside the scope of economic analysis. Mevertheless, there is a general consensus that progressive taxation promotes equity better than proportional or regressive taxation. Table VI.1 brings out some interesting results. The agricultural sector (poorer) pays a lower percentage of its income in taxes than the nonagricultural sector for the first two alternative sets of shifting assumptions. Under the forward shifting assumption of federal indirect taxes, agriculture pays a higher tax rate for four out of twelve years. For these four years, the agricultural sector is clearly overtaxed.

Table W.1

Incidence of Federal and Provincial Taxes on Different Sectors

Years	Shifting native I	ASS	Assumptions: A of No Shifting	Alter-	Shifting native II		Partial	Alter- Shifting	Shifting Ass Alternative	III		Shifting	
			N	-	A	1 10	N		4.			×	
2013 201720	Tax Per Canita (Ks.)	7/3	Tax Per Canita (Rs.)	1/3	Tax Per Capita (Rs.)	T/y (%)	Tax Per Capita (Rs.)	T/y (%)	Tax Per Capita (Rs.)	7/3	Tax Per Canita (Rs.)	1/3	ture to Per Capita Income in Mgriculture
1972/73	39.69	6.32	206.42	15.51	57.84	9.25	184.98	13.90	86.53	13.79	151.12	11.36	2.18
1973/74	58,76	7.61	266.52	15.67	70.79	9.17	242.82	14.27	116.46	15.08	199.11	11.70	2.22
1974/75	68.35	7.51	340.03	15.57	96.72	10.63	308.13	14.11	150.44	16,53	247.71	11.34	2.40
92,5,16	33.63	3.29	410.52	16.31	64.27	6.30	376.85	14.97	122.32	11.98	313.10	12.44	2.47
1976/77	15,81	1.38	470.75	16.91	53.07	4.60	431.28	15.49	126.41	11.03	352.34	12.66	2.43
1977/78	23.55	1.79	563.44	20.96	63.04	4.79	522,60	19.44	136.31	10.37	446.81	16.62	2.54
1978/79	23.08	1.70	654.16	18.12	69.46	5.11	606.91	16.81	161.33	11.87	513.31	14.22	2.66
1979/80	27.23	1.76	784.71	18.86	80.45	5.20	731.83	17.59	181.62	11.74	628.81	15.12	2.69
1980/81	36.10	2.10	938.79	19.59	98.50	5.74	878.26	18.33	224.19	13.06	756.33	15.78	2.79
1981/82	27.65	1.42	948.58	17.78	116.55	5.96	1232.49	23.10	218.79	11.20	767.55	14.39	2.73
1982/83	28.30	1.33	1070.81	17.72	105.88	5.03	998.13	16.52	259.14	12.20	857.32	14.19	2.85
1983/84	30,23	1.40	1263.30	18.80	155.68	7.20	1150.07	17.18	365.45	16.90	960.73	14.35	3.10

Note . Tax per capita is in rupees: T/y is Tax as per cent of sectoral income. A denotes agricultural

Some Indicators of Taxable Capacity of Different Sectors

'5	Per Capita Income in Agricul- tural Sector	Subsistence Requirements Per Capita in Agricul- tural Sector	Per Capita Income in Non-Agricul- tural Sector	Subsistence Requirements in Non-Agri- cultural Sector	Taxable Capacity in Agri- cultural Sector	Capacity in Non-	Relative Taxable Capacity of Agri- culture
	(Rs.)	(Rs.)	(Rs.)	(Rs.)	(Rs.)	(Rs.)	(Rs.)
-73	6 10.05	551.19	1330.77	668.68	58.86	662.09	O.089
-74	765.86	679.94	1701.16	836.81	85.92	864.35	0.099
-75	9 10.23	863.68	2184.19	1058.98	46.55	1125.21	0.041
-76	1020.71	968.37	2517.61	1181.94	52.34	1335.67	0.039
-77	1 145.93	1051.40	2783.85	1344.43	94.53	1439.42	0.066
-78	1314.74	1120.09	3339.81	1380.03	194.65	1959.78	O.099
-79	1359.27	1205.22	3610.06	1484.54	154.05	2125.51	0.072
-80	1546.74	1328.05	4160.13	1637.61	218.69	2522 ,52	0.087
-81	1716.93	1510.05	4791.87	1864.75	206.88	2927.12	0.071
-82	1953.16	1701.53	5335.10	2104.04	251.63	3231.06	0.078
:-83	2123.59	1770.66	6041.89	2192.17	352.93	3849.72	0.092
1-84	2163.08	1806.42	6695.20	2239.69	356.66	4445.51	0.080
	Self LOSSIN			200 30H 116	TIDE VITA De		

mortganization the second and third attached attached action

culture for the lone year vanishes for high values of destrable group ression.

section, the second ellernative reflects the erested mestity over

elosely show others. The third alternative assumption out, would be

For the other eight years and in case of the other two alternative sets of shifting assumptions for all twelve years, an appraisal of equity requires a specification of "socially desirable progression' and an explicit account of subsistence requirements in both sectors.

These limitations are removed in the measure of relative tax burden as presented in Table VI.3. Relative tax payments by agriculture under different shifting assumptions are presented in the first three rows. The fourth row presents estimates of relative taxable capacity. The capacity is defined by per capita income minus per capita subsistence. The relative capacity of agriculture fluctuates between 0.039 in 1975/76 and 0.99 in both 1973/74 and 1977/78. The relative capacity is raised to powers ranging from 1.1 to 1.9 and the estimates are presented in the last nine rows in Table VI.3. Intersectoral equity can be measured by a comparison of relative tax payments and equity-adjusted relative taxable capacity. We shall present results for only the moderate degree of progression i.e. taxable capacity raised to power 1.5.

Under the first shifting assumption alternative, the relative tax payments of agriculture are larger than the relative equity-adjusted taxable capacity for all but one year. The inequity in favour of agriculture for the lone year vanishes for high values of desirable progression. That agriculture is over-taxed for most years is clearly established.

Under the second and third alternative sets of assumptions, agriculture is overtaxed for all years. As mentioned in an earlier subsection, the second alternative reflects the empirical reality more closely than others. The third alternative assumption set, would be

Table VI.3

Relative Tax Burden on the Agricultural Sector. First Approximation

al gur	1972/	1973/	1974/	1975/	1976/	1977/ 78	1978/	1979/	1980/	1981/	1982/	1983/
Relative Tax Payments: I	0.192	0.220	0.201	0.082	0.034	0.042	0.035	0.035	0.038	0.029	0.026	0.024
Relative Tax Payments: II	0.313	0.292	0.314	0.171	0.146	0.121	0.114	0.110	0.112	0,095	0.107	0.135
Relative Tax Payments: III (Relative Tax Capacity*for value of equal to: )	0.573	0.585	0.607	0.391	0.359	0.305	0.314	0.280	0.296	0.285	0.302	0.380
1.0	0,089	0,099	0.041	0.039	0.066	0.099	0.072	0.087	0.071	0.078	0.092	0.080
prof.	0.070	0.079	0.030	0.028	0.050	0.079	0.055	0.068	0.054	090.0	0.072	0.062
1.2	0.055	0.062	0.022	0.020	0.040	0.062	0.043	0.053	0.042	0.047	0.057	0.048
1:3	0.043	0.049	0.016	0.015	0.030	0.049	0,033	0.042	0.032	0.036	0.045	0.037
1.4	0,034	0.039	0.011	0.011	0.020	0.039	0.025	0.033	0.025	0.028	0.035	0.029
1.5	0.027	0.031	0.008	0.008	0.017	0.031	0.019	0.026	0.019	0.022	0.030	0.023
1.6	0.021	0.021 0.025	0.006	0.006	0.013	0.025	0.015	0.020	0.015	0.017	0.022	0,018

applicable if quotas in the Pakistani economy were non-binding or slack constraints. This is the shifting alternative most likely to be useful for international comparisons of tax burdens on agricultural sectors.

The extent of inequity against or for agriculture is measured and the results are presented in Table VI.4. The figures of tax per capita on agriculture if the sector is taxed equitably are presented in columns 3, 6 and 9 for three alternatives of tax shifting. The calculation in these columns shows the tax-per-head that would be paid by agriculture to satisfy inter-sectoral equity and the need to keep all-Pakistan tax (and hence tax-per-head) at the level actually attained in the particular year. The difference between this and the tax per head in fact paid as a percentage of agricultural income per head shows the proportion of yearly agricultural income being transferred to non-agriculture as compared with a system that taxed the two sectors equitably.

Under the first shifting alternative, the degree of resource flow out of agriculture for the first four years varies from 3 per cent to 7 per cent. The resource flow out of agriculture declines sharply in 1976/77 and is less than 1 per cent. Under the other two alternatives, the extent of resource flow out of agriculture fluctuates but is quite heavy. This is especially so in the case of forward shifting of indirect taxes. The extent of this outflow represents the excess of agricultural tax payments, direct and indirect, over the payments that would reflect ability to pay.

	Shift	Shifting Assumption: Alternative-I	ou:	Shifting Alter	Assumption native-II		Shifting	ifting Assumption: Alternative-III	
Years	Tax Per Capita on A	Tax Per Capita on A if A Taxed Equitably	Resource Transfer	Tax Per Capita on A	Tax Per Capita on A if A Taxed Equitably	Resource	Tax Per Capita On A	Tax Per Capitaon A if A Taxed Equitably	Resource Transfer
	(RS.)	(Rs.)	(%)	(Rs.)	(Rs.)	(%)	(Rs.)	(%)	(%)
(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)	(6)	(10)
1972/73	39.69	6.54	5.43	57.84	6.56	8.40	86.53	6.59	13.10
1973/74	58.76	10.14	6.35	70.79	10.00	7.94	116.46	10.15	13.88
1974/75	68.35	3.54	7.12	96.72	3.48	10.24	150.44	3.48	16.14
1975/76	33.63	3.44	2.96	64.27	3,43	5.96	122.32	3,43	11.65
1976/77	15.81	7.79	0.70	63.07	8,26	4.78	126.41	8.09	10.32
1977/78	23.65	17.08	0.49	63.04	17.83	3.44	136.31	8,43	9.73
1978/79	23.08	12.96	0.74	97.69	12.95	4.16	161.33	12.91	10.92
1979/80	27.23	19.55	0.50	80.45	20.16	3.90	181.62	20.01	10.45
1980/81	36.10	17.33	1.09	98.50	16.21	4.79	224.19	8.11	12.58
1981/82	. 27,65	20.76	0.35	116.55	28.63	4.50	218.79	20.82	10.14
1582/83	. 28,30	29.69	90.0-	106.88	29.69	3.63	259.14	29.68	10.80
1983/84	30.23	28.74	0.07	155.68	28.70	5.87	365.45	28.62	15.57

er in

SPE

F 25

Tax Per Capita, Equity Tax Per Capita, Resource Transfer from Agriculture-

# 4. SECTORAL TAX EQUITY IN PAKISTAN: SECOND APPROXOMATION.

In this section, the tax burden on the agricultural sector is estimated by considering a broader measure of taxable capacity than just the average per capita income net of subsistence. It was argued in the first section that wealth, wealth distribution and income distribution are relevant as determinant of any sector's ability to pay taxes. We would first review the possible alternative methods of the estimation procedures adopted in the literature.

The general problem can be posed very easily but its solution requires considerable ingenuity. The functional relationship of taxable capacity to the relevant variables can be assumed as linear and specified as

$$C = \alpha \overline{Y} + \beta \overline{W} + \gamma W + \delta i$$

Where C is taxable capacity,  $\bar{Y}$  is taxable income per capita,  $\bar{W}$  is wealth per capita,  $\bar{W}$  is some index of wealth inequality and  $\bar{I}$  is some index of income inequality. The problem being faced is one of finding the coefficients  $\alpha$ ,  $\beta$ ,  $\gamma$  and  $\delta$  that can be applied in the construction of an index of taxable capacity. No scientific and simple solution is available for the estimation of these coefficients or weights. The only requirement is that income should have a larger weight than other variables and that the sign of the coefficients should be positive i.e. the higher the values of Y. W. W and i, the higher the taxable capacity. Three possible methods have been suggested in the literature to taxing the issue at hand.

# (a) Model Tax System

Model tax system and model tax rates can be assumed and the actual yields can be compared to the potential yields. This method has been applied to different states and countries / 64 / However, for two sectors, structurally quite different, this method is of limited use. In any case no such comparable data are available for the sectors in Pakistan.

#### (b) Least Squares Method

If the data on the relevant variables for sectors in other countries of comparable structure are available, tax payments can be regressed on the variables of interest. The estimated coefficients from the equations for two sectors could be applied to the data for Pakistan. The major assumption in this application is that the values of independent variables used in the equations do not differ much from the values in Pakistan. Since such data are available mostly for developed countries, this method is not applicable in the case of Pakistan.

#### (c) Arbitrary Weights

have adopted arbitrary weights for the different variables. Ghandhi / 30 has used this method to get an estimate of the relative measure of sectoral taxable capacity. For a broad range of arbitrary weights for income, wealth, income inequality and wealth inequality, the inequity in favour of agriculture not only persisted but increased relative to his first approximation estimates. This result is awkward for two

reasons  $\frac{1}{2}$ . First, since the income and wealth inequality is relatively more even in the agricultural sector and the taxable capacity is a direct function inequality, the broader measure of the taxable capacity for agricultural sector should be lower than in the first estimation. Ghandhi's results are in direct contradiction with the theoretical expectation. The reason for the flaw in Ghandhi's measure is the peculiar nature of his weighting scheme. The agricultural/non-agricultural ratios of incomeper-head and of Lorenz coefficients are weighted by a broad range of arbitrary weights. Since the ratio of agriculture to non-agriculture Lorenz coefficients (while less than unity) is greater than the ratio of agriculture to non-agricultural incomes-per-head, the effect of such weighting is to increase the apparent taxable capacity of the rural areas relative to the urban areas. Greater rural equality than urban equality, in theory, should reduce relative rural taxable capacity of the agricultural sector. In Ghandhi's measure, the ratio of the rural to urban Lorenz coefficients must fall short, not just of unity, but of the ratio of rural to urban income per head. Since the facts are otherwise, linear weighting introduces a theoretically untenable bias. Second, the summary measures of distribution of income do not tell anything about the distributional pattern at the really high and taxable end of the income scale. Due to these reasons, the method suggested by Ghandhi cannot be adopted in our study. To indicate the extent of bias, we have applied the weighting scheme suggested by

Michael Lipton / 57/ was the first one to criticize the Ghandhi's results for India. We build the argument for Pakistan largely on the method of analysis proposed by Lipton / 57/.

Ghandhi to the sectoral data for Pakistan. The results are presented in Table VI.5. The comparison of the first and second approximation estimates of relative taxable capacity of agriculture show considerable divergence. More important is the finding that the second approximation capacity is always higher than the first approximation.

### (d) Method Adopted in the Study

We start from the income distribution patterns in rural and urban areas for 1979/80 / 68 /. Table VI.6 presents estimates for income distribution. We assume that the rural and urban distributional patterns hold respectively for agricultural and non-agricultural sector. Columns 4 and 7 in Table VI.6 estimate the share of sectoral income per capita in each income group in agriculture and non-agriculture. We can combine Tables VI.2 and VI.6 to estimate taxable capacity in agricultural and non-agricultural sectors for 1972/73 to 1983/84. For each year, the figure for each group is (1) income per capita from Table VI.2, times the sectoral share of income from Table VI.6 minus subsistence requirements per capita from Table VI.2. Tables VI.7 and VI.8 respectively show the maximum total sum that could be taken from each income-group of households from agriculture and non-agriculture and still leave members of each sector in each group enough to subsist on. Weighting the taxable capacity in each income-group by the percentage Of sectoral population in each group gives us the taxable capacity of the average member in each of the two sectors. Sectoral weights are shown in columns 3 and 6 in Table VI.6. Negative values in Tables VI.7 and VI.8 are assumed to be zero. This assumption is made as it is practically impossible to transfer incomes from households with positive taxable

Table VI.5

Estimates of Relative Taxable Capacity of the Agricultural Sector with Arbitrary Weights Given to Income Per Capita and Income Inequality

	Relative Tax-	Relative Taxabl	e Capacity: Second	d Estimates
Years	able Capacity First Estimate	Alternative I Weights $\alpha = 10$ $\delta = 0.5$	Alternative II Weights & = 10 & = 1	Alternative II) Weights $\alpha = 1$ $\delta = 1$
1972-73	0.089	0.122	0.152	0.436
1973-74	0.099	0.131	0.161	0.441
1974-75	0.041	0.076	0.108	0.412
1975-76	0.039	0.074	0.107	0.411
1976-77	0.066	0.100	0.131	0.424
1977-78	0.099	0.131	0.161	0.441
1978-79	0.072	0.106	0.137	0.427
1979-80	0.087	0.120	0.150	0.435
1980-81	0.071	0.105	0.136	0.427
1981-82	0.078	0.112	.0.142	0.430
1982-83	0.092	0.125	0.155	0.437
1983-84	0.080	0.113	0.144	0.431

Article Park In Land of Park to be set a court of the second of the seco

Table VI.6

Rural and Urban Income Distribution for the year 1979-80

mthly Income	RUR.	AL AREAS	70 12 112	URI	BAN AREAS	
ar Household Rupees	Percentage of Sectoral Income	Percentage of Sectoral Population	Incomo Per Head : Sectoral Average Inco	Percentage of Sectoral Income	Percentage of Sectoral Population	Income Per Head Sectoral Average Income
(1)	(2)	(3)	(4)	(5)	(6)	(7)
< 300	1.48	2.59	0.57	0.33	0.67	0.49
01-400	3.84	6.06	0.63	1.08	2.12	0.51
01-500	6.66	9.47	0.70	2.47	4.73	0.52
11-600	9.22	11.95	0.77	3.51	6.48	0.54
i1-800	18.86	22.44	0.84	8.97	14.67	0.61
1-1000	15.46	16.73	0.92	10.66	15.37	0.69
11-1500	20.97	18.44	1.14	20.47	25.08	0.82
ii-2000	8.66	6.21	1.40	12.85	12.87	1.00
1-2500	4.55	2.73	1.67	8.21	6.35	1.29
1-3000	1.94	1.07	1.81	5.17	3.29	1.57
ી-3500	1.53	0.73	2.08	4.18	2.29	1.83
3500	6.81	1.59	4.29	22.11	6.09	3.63
Groups	100	100	1.0	100	100	100

Taxable Capacity of Acricultural Sector Per Capita

					9 1 1 9.				a and		25	(In Rupees	)
Year/ Income Group	<300		301 to 400	401 to 500	501 to 600	601 to 800	801 to 1000	1001 to 1500	1501 to 2000	2001 to 2500	2501 to 3000	3001 to 3500	3500 and above
1972-73	-203.	46	-156.86	-124.16	-81.45	-38.75	19.06	144.27	302.88	467.59	553.00	717.71	2065.92
1973-74	-243.	44	-197.45	-143.84	-90.23	-36.62	24.65	193.14	392.26	599.05	706.23	913.05	2605.60
1974-75	-344.	85 -	-289.74	-226.52	-162.80	-99.09	-26.27	173.98	410.64	656.40	783.84	1029.60	3041.21
1975-76	-386.	57	-325.32	-253.83	-182.42	-110.97	-29.32	195.24	460.62	736.22	879.12	1154.71	3410.41
1976-77	-398.	22 -	-329.46	-249.25	-169.04	-88.82	2.86	254.96	552.90	862.30	1022.73	1332.13	3864.64
1977-78	-370	69 -	-291.80	-190.77	-107.74	-15.71	89.47	378.71	720.55	1075 53	1259.59	1614.57	4520.14
1978-79	-430.	44 -	-348.88	-253.73	-158.58	-63.43	45.31	344.35	697.76	1054.76	1255.06	1622.06	4626.05
1979-80	-446	41 -	-353.60	-245.33	-137.06	-28.89	94.95	435.23	837.39	1255.01	1471.55	1889.17	5307.46
1980-81	-531	40 -	-428.38	-308.20	-188.01	-67.83	69.53	447.25	893.65	1357.22	1597.59	2061.16	5855.58
82 -5	88.23	-471	04	-334.32 -	197.60	-60.88	95.38	525.07	1032.89	1560.2	5 1833.69	9 2361.04	6677.5
83 -5	660.21	-432	2.80	-284.15 -	-135.50	13.16	183.04	650.23	1202.37	1775.70	2073.0	2646.4	7339.5
-84 -3	573.46	-443	3.68	-292.26 -	-140.85	10.57	183.61	659.49	1221.89	1805.93	2 2108.7	5 2692.83	2 7473.15

Note: Each entry is agricultural income per capita for the year times the ratio of rural group income per capita to rural average income per capita <u>minus</u> agricultural sector subsistance requirements per capita for the year.

Laver year

# Taxable Capacity of Non-Agricultural Sector per Capita

			1				1	13			(in rupe	25
Income Year groups	∠ 300	301 to 400	401 to 500	501 to 600	601 to 800	801 to 1000	1001 to 1500	1501 to 2000	2001 to 2500	2501 to 3000	3031 to 3500	3500 and above
1972-73	-16.60	10.01	23.32	49.94	143.09	249.55	422.15	662.09	1048.01	1420.63	1760.63	4162.62
1973-74	-3.24	30.78	47.79	81.82	200.90	336.99	558.14	864.35	1357.69	1834.01	2276.31	5338.40
1974-75	11.27	54,96	76.80	120.48	237.38	448.11	732.06	1125.21	1758.63	2370.20	2936.09	6869.63
1975-76	51.69	102.04	127.22	177.57	353.80	555,21	882.50	1335.67	2065.78	2770.71	3425.29	7956.98
1976-77	19.66	75.33	103.17	158.85	353.72	576.43	938.33	1439.42	2246.74	3026.21	3750.02	8760.59
1977-78	256.48	323.27	356.67	423.47	657.25	924.44	1358.61	1959.78	2928.32	3663.47	4731.82	10743.48
1978-79	284.39	356.59	.392.69	464.89	717.60	1006.40	1475.71	2125.52	3172.44	4163.25	5121.87	11619.78
1979-80	400.85	484.05	.525.66	608.86	900.07	1232.88	1773.70	2522.52	3728.96	4893.79	5975.43	13463.66
1980-81	487.27	579.10	627.02	722.86	1058.29	1441.64	2064.58	2927.12	4316.76	5658.49	6904.37	15529.74
1981-82	510.16	616.86	670.21	776.91	1150.37	1577.18	2270,74	3231.06	4778.24	6272.07	7659.19	17262.37
1982-83	768.36	889.19	949.61	1070.45	1493.38	1976.73	2762.18	3849.72	5601.87	7293.60	8864.49	19739.89
1983-84	1040.96	1174.86	1241,81	1375.72	1844.38	2380.00	3250.37	4455.51	6397,12	8271.77	10012.52	22063.88

Note: Each entry is non-agricultural income per capita for the year times the ratio of urban group income per capita to urban average income per capita, Minus non-agricultural sector subsistance recuirements per capita for the year.

capacity to households with negative taxable capacity via the present tax system. Before weighting we raise each value in Tables VI. 7 and VI. 8 to the power of 1.5 to adust the taxable capacity figures for the equity criterion. Table VI.9 presents the equity-adjusted taxable capacities for the two sectors and their relative values. In column 4, we present the first approximation estimates of the relative taxable capacity of agriculture. The relative sectoral capacity in the second approximation is higher than the first approximation. This is inconsistant with the theoretical expectations and arises because we assumed the negative entries in Tables VI. 7 and VI.8 to be equal to zero. This tends to overstate the relative capacity of agriculture since the negative values which we have assumed to be zero are more frequent and larger in absolute terms in agriculture.

The results on sectoral equity in taxation can be derived by the comparison of relative tax payments under three alternative shifting assumptions (first three rows in Table VI.3) with the relative taxable capacity as shown in column 3 in Table VI.9. The general pattern of results is different from the first approximation estimates of tax burden on agriculture as it is relatively undertaxed under the first alternative set of shifting assumptions for all years except for 1973/74 and 1974/75. The extent of resource inflow into agriculture was about 3 to 4 per cent for most years. Under the second set of shifting assumptions, agriculture is overtaxed for six years and undertaxed for the other six years. In the case of the third alternative of tax shifting, agriculture is overtaxed. The resource transfer out of agriculture varies from 4.98 per cent of agricultural income in 1977/78 to 12 per cent in 1983/84.

Table VI.9

# Estimates of Taxable Capacity per Capita of Different Sectors with Moderate Progression Defined as Taxable Capacity Raised to the Power 1.5

(Second Approximation)

		LANGE BALL STEEL LEGIT	OF THE PARTY OF TH	THE PART AND PART OF THE PARTY
Year	Agricultural Sector	Non-Agricultural Sector	Relative Tax- able Capacity of Agriculture	Relative Taxable Capacity Agri- culture
107350	1 0 x61 AB 6360 24	notisks verantato	$(1) \div (2)$	(First Approximation)
	(1)	(2)	(3)	(4)
1972-73	27.01	138.27	0.195	n . 089
1973-74	39.15	205.34	0.191	n.099
1974-75	45.42	302.89	0.150	0.041
1975-76	53.94	391.34	0.138	n.039
1976-77	67.74	439.77	0.154	0.066
1977-78	94.49	687.22	0.137	n .099
1978-79	92.78	776.01	0.120	0.072
1979-80	118.99	1001.19	0.119	0.087
1980-81	133.56	1250.95	0.107	0.071
1981-82	164.72	1451.50	0.113	0.078
1982-83	211.76	1883.13	0.112	0.092
1983-84	213.86	2342.20	0.091	.080
	1974/75			

realive shifting assumbtions of texes except for a few years whe

of the state of texas to income. A clear case of overtweather of the

the matrice and the most research the same longery for the same same.

#### SUMMARY AND CONCLUSIONS

Based on certain assumptions regarding the shifting of direct and indirect taxes levied by the federal and provincial governments, formal incidence for the agricultural and non-agricultural sectors was estimated for each of the twelve years from 1972/73 to 1983/84. The analysis ignores the burden of local taxation as data on tax collection at this level of government are not available. The analysis also ignores the impact of usher, an Islamic levy based on the gross value of agricultural production. This levy was introduced in 1982. Ignoring its impact does not affect the analysis for years prior to 1982/83. The main findings were listed as we proceeded with the analysis. These findings are briefly summarised below:

- (1) The taxes per capita estimated for the agricultural sector are lower than the taxes per capita for the non-agricultural sector for all three alternative sets of shifting assumptions of taxes.
- The agricultural sector's per capita income is less than half the non-agricultural sector's per capita income. The traditional measure of tax burden as a ratio between taxes and income is, therefore, not a valid measure for the intersectoral comparison of the tax burdens. The poorer sector has a lower tax to income ratio for all three sets of alternative shifting assumptions of taxes except for a few years when in the case of forward shifting of indirect taxes the agricultural sector has a higher ratio of taxes to income. A clear case of overtaxation of the agricultural sector can be made for these years.

- (3) First approximation estimates of relative t agriculture were made on the basis of per capita incorquirements. Agriculture was found to be undertaxed under the non-shifting assumption of central indirect overtaxed for all remaining years of the study. The overtaxed for all the years in the case of the other assumptions regarding the shifting of the taxes.
- with respect to a hypothetical situation of equitable two sectors, is sensitive to the shifting assumptions a percentage of yearly agricultural income, the resolvaried from a figure of 0.07 per cent in 1983/84 to 7 for the shifting assumptions defined in this chapter. In the case of the other two alternatives of tax shift the resource transfer out of agriculture was consider transfer ranges from 3.43 per cent in 1975/76 to 29.6 using Alternative-II. The range for the resource transfer alternative-III is from 9.73 per cent in 1977/78 in 1974/75.
- (5) A second approximation estimate of the relative taxable capacity was made. According to this measure, overtaxation of agriculture decreased relative to the measure of tax burden. In the case of the first and salternative shifting assumptions, the agricultural sec be undertaxed for most years of the study. However, if alternative, which incidentally is the more realistic

regarding the shifting assumptions of different taxes for the period under study, the agricultural sector is found to be overtaxed for all 12 years of the study. The general conclusion that can be drawn from the analysis is that judging the tax system from the equity perspective the agricultural sector is overtaxed and bears a higher share of the tax burdens than its counterpart - the non-agricultural sector.

out to metablis elias lups to militaria a contestogal a graphism visit the coclege, leg some start the contist has been as the contist to

a commence years and test threat three comments of the bout

The service of the state of the state of the service of the servic

# DISTRIBUTION OF TAX BURDEN BETWEEN AGRICULTURAL AND NON-AGRICULTURAL SECTORS.

The purpose of this note is to provide information on the allocation of the commodity taxes shifted forward to the consumers between the agricultural and the non-agricultural sectors of the economy. Since most of the indirect taxes are proportional to the expenditure, the revenue has been distributed between the two sectors on the basis of total expenditure on the taxed commodity. The total value of expenditure on a taxed commodity by the agricultural/non-agricultural sector is obtained by multiplying per capita expenditure on the commodity by the agricultural/non-agricultural population. The main body of the chapter outlines shifting assumptions of various direct and indirect taxes and provides details regarding the estimation of agricultural and non-agricultural population.

The consumer expenditure data were collected from the Household Income and Expenditure Survey, 1979. The geographical coverage of the Survey was all rural and urban areas of Pakistan excluding Tribal Agencies, Special Areas of Peshawar and D.I. Khan Divisions and Malakand Division in the N.W.F.P. According to 1972 Population Census the population of these excluded areas is about 6.7 per cent of total population of Pakistan. Data on consumer expenditure were collected from 1400 sample households of which 608 were urban and 792 were rural. The Survey provides detailed information on the consumption expenditure pattern of rural and urban households. It contains 37 items under food, 6 items under apparel,

textile and footwear, 14 items under fuel and lighting, 6 items under house rent and housing, 5 items under furniture and fixture and 13 items under the category of miscellaneous items. Commodity taxes are levied on a large number of commodities. Data are available on tax collection from Central Excise Duteis and Import Duties classified by 45 and 33 broad categories of commodities respectively. The main purpose for using Household Income and Expenditure Survey data was to find out the per capita expenditure on different commodities. The tax on the commodities for which consumption data were available from the Survey was distributed between the two sectors according to the consumption pattern. For those commodities for which consumption data were not available either the consumption pattern of similar commodities was used or arbitrary assumptions on the basis of informed judgment were made to distribute tax revenue between the agricultural and the non-agricultural sectors. Tax collections from commodity taxation has been assigned to the two sectors in the following manner:

# CENTRAL EXCISE DUTLES

and non-agricultural sectors in the proportion of expenditure on tea. (2) Vegetable Non-essential Oil. The revenue was distributed between the two sectors in the ratio of expenditure on the consumption of edible oil. (3) Vegetable Product The sectoral distribution of revenue from Vegetable Products was made on the basis of expenditure on vegetables. (4) Beverages. Revenue from excise duties on Beverages was distributed in the proportion of expenditure on food and drink (5) Sugar. The revenue was distributed in the ratio of combined expenditure on refined (i.e. milled) and desi-sugar. (6) Tobacco. Revenue from Tobacco vas

divided between the agricultural and non-agricultural sectors on the basis of expenditure on cigarettes and cigars. (7) Cement. Most of the houses in the rural areas of Pakistan are "Katcha" houses. These houses are made of mud, grass, stones, etc. A majority of rural population seldom buys cement. Although the number of "Pukka" houses (houses made of cement, bricks, etc.) have increased in the rural areas due to influx of foreign remittances yet these are very small in number relative to the number of houses in urban areas. It was assumed that 90 per cent of total revenue from the source comes from the non-agricultural sector while the remaining 10 per cent from the agricultural sector. (8) Salt. Ratio of expenditure on salt was used to distribute the revenue between the agricultural and nonagricultural sectors. (9) Natural Gas. Ratio of expenditure on gas consumption was used to divide the revenue between the two sectors. (10) Petrolium Gases. Same ratio as in the case of (9) was used. (11). Furnace Oil. It is mainly used for casting iron. Same ratio as in the case of (7) was used to distribute revenue between the agricultural and non-agricultural sectors. (12) High Speed Diesel Oil. Expenditure on transport and travelling was used for sectoral distribution of revenue from high speed diesel oil. (13) Light speed diesel oil. This is mainly used in the agricultural sector. Sector-wise pattern of its consumption, given in Energy Yearbook of Pakistan, revealed that it is divided in the ratio of 95: 05 between the agricultural and the non-agricultural sectors, respectively. (14) Kerosene Oil. Ratio of expenditure on Kerosene oil was used to divide the revenue between the agricultural and non-agricultural sectors. (15) Motor Spirit (including jet fuel). Ratio of expenditure on travelling and transport was used. (16) Petrolium Grease, Petrolium lubricant oil, Solvant Neptha, Asphalt and Other Petrolium Products. These fall in the broad category of

Petrolium oil and lubricant products. No information is available from the Survey data on the consumption pattern of these items. The items appear as intermediate input in the production of crops, grain milling, sugar refining and most of the manufactured items. It was assumed that revenue from the items is divided in the percentage of 75 and 25 between the nonagricultural and the agricultural sectors respectively. (17) Polishes and Creams. Same ratio as in the case of (7) was used. (18) Paints and Varnishes. These are mainly used to paint houses, furniture and different kinds of wooden and metal products. Most of the houses in the rural areas are made of mud. Rural households rarely use paints and varnishes. It was assumed that the revenue is distributed in the proportion of 85:15 between the non-agricultural and the agricultural sectors respectively. (19) Cosmatics. It is believed that a large proportion of the revenue comes from the nonagricultural sector. Therefore, 80 per cent of total revenue was allocated to the non-agricultural sector and the remaining 20 per cent was distributed in the proportion of expenditure on personal care. (20) Soap and Detergent. Ratio of expenditure on personal care was used. (21) Soda Ash. Ratio of expenditure on laundry and cleaning was used. (22) Plastic Products. It was assumed that the revenue is divided in the ratio of 20: 80 between the agricultural and the non-agricultural sectors respectively. (23) Tyres and Tubes. Revenue from tyres and tubes comes from two main sources viz. non-commercial or private vehicles and commercial or public service vehicles. A small proportion of agricultural population owns private motors and public transport service is also centered in the urban areas due to better availability or road and other infrastructure. It was, therefore, assumed that 85 per cent of total revenue is generated from the non-agricultural sector while the remaining 15 per cent was divided between the agricultural

and the non-agricultural sectors in the proportion of expenditure on travelling and transport. (24) Rubber Products. Same ratio as in the case of (22) was used to distribute revenue from taxes on Rubber Products. (25) Tanned Leather. Ratio of expenditure on footwear was used. (26) Paper and Paper Board. The Survey does not provide information on the sectoral consumption pattern of the item. However, the Survey provides information on expenditure on education that includes expenditure on paper and selected paper products. Expenditure on education was used to distribute revenue between the sectors. (27). Bank Cheques. The revenue was divided in the proportion of 05: 95 between the agricultural and the non-agricultural sectors. (28) Cotton Yarn and Fabrics. The revenue was distributed in the proportion of expenditure on clothing. (29) Man-Made-Fabrics. Same ratio as in the case of (28) was used to distribute the revenue between the agricultural and the non-agricultural sectors. (30) Matches. Ratio of expenditure on matches was used. (31) Man-Made-Yarn. Same ratio as in the case of (28) was used. (32) Electric Batteries. Use of electric batteries is very rare in the rural areas of Pakistan. Tax on electric batteries enters as cost item in the manufacture of goods. The revenue was assumed to be distributed in the proportion of 05: 95 between the agricultural and the non-agricultural sectors. (33) Electric Bulbs and Tube Lights. Use of electric bulbs and tubes is a necessity unlike the electric batteries. A large proportion of agricultural households do not use electric bulbs due to the obvious fact that electricity has not reached to their houses. The revenue was, thus divided between the agricultural and the non-agricultural sectors in the proportion of expenditure on the sectors in the sector in the sectors in the sector i electricity. (34) Electric Fans. Same ratio as in the case of (33) was used. (35) Caustic Soda. Ratio of expenditure on Laundry and cleaning was used. (36). Glass and Glassware. Ratio of expenditure on crockery was used. (37) Service Hotels. Same ratio as in the case of (27) was used.

- (38) Wires and Cables. Ratio of expenditure on electricity was used,
- (39) Knitting Yarn. Same ratio as in the case of (28) was used. (40) Woollen Yarn and Fabrics. Same ratio as in the case of (28) was used. (41) Metal Containers. Ratio of expenditure on durable household effect was used to distribute the revenue. (42) Mild Steel Products. Ratio of expenditure on durable items of crockery and cutlery was used. (43) Wires and Cables. Same ratio as in the case of (33) was used. (44) Jute Manufactures. Same ratio as in the case of (22) was used. (45) Miscellaneous Items. Expenditure on these items was used to distribute revenue from miscellaneous items.

#### IMPORT DUTIES

distributed in the ratio of expenditure on meat and fish. (2) Milk, Butter,
Cheese and Honey. The revenue was distributed in the proportion of
expenditure on milk and milk products. (3) Fruits, Nuts and Vegetables.
The revenue was split between the agricultural and the non-agricultural
sectors in the ratio of expenditure on fruits and vegetables. (4) Coffee,
Tea and Spice. The revenue was distributed in the proportion of expenditure
on coffee, tea and spice. (5) Oil-see ds and Miscellaneous Fruits. The
revenue was split in the proportion of combined expenditure on edible oil
and miscellaneous fruits. (6) Sugar and Confectionary. Expenditure on
the combined category of Gur, Sugarcane products, honey and sugar preparation
was used to split the revenue between the agricultural and the non-agricultural
sectors. (7) Animal and vegetable fats and oils. Revenue was
distributed in the ratio of expenditure on edible oil. (8) Beverages,
spirits and vinegar. The Survey provides no information on the consumption

pattern of the items. It was assumed that the revenue is distributed in the proportion of 15:85 between the agricultural and the non-agricultural sectors. (9) Tobacco. The revenue was distributed between the sectors according to consumption expenditure pattern of cigarettes and cigars. (10) Mineral Fuels, Oils and their Products. The revenue was split according to the expenditure on kerosene oil. (11) Chemicals and Chemical Products. The Survey does not provide information on the expenditure pattern of chemicals and chemical products. It is believed that a large proportion of the revenue comes from the non-agricultural sector. The revenue from the source was distributed in the proportion of 15: 85 between the agricultural and the non-agricultural sectors respectively. (12) Pharmaceutical Products. The revenue was distributed in the proportion of expenditure on medical care. (13) Paints and Varnishes. It was assumed that the revenue is distributed in the proportion of 15: 85 between the agricultural and the non-agricultural sectors respectively. (14) Perfumery Soap and Toilet Preparations. The revenue was distributed in the ratio of expenditure on personal care. (15) Matches and Other Explosives. The revenue was distributed in the proportion of expenditure on matches. (16) Photographic and Cinematographic Products. The revenue was split according to expenditure on goods and services relating to reading and recreation. (17) Rubber and Rubber Products. The Survey provides no information on the consumption pattern of the items. Since non-agricultural sector uses more intensively the products relative to the agricultural sector, it was assumed that the revenue is divided in the ratio of 20: 30 between the agricultural and the non-agricultural sectors respectively. (18) Leather Articles and Footwear. The revenue was distributed in the proportion of expenditure on Footwear. (19) Wood, Pulp. Paper and Stationery. Ratio of

expenditure on education was used to distribute the revenue between the sectors. (20) Silk, Yarn and Fabrics. The revenue was distributed in the proportion of expenditure on cloth and its accessaries. (21). Glass and Earthenware. Ratio of expenditure on crockery was used to distribute the revenue. (22). Precious metals, stones, pearls and immitation jewellery. It was assumed that the revenue is distributed in the proportion of 5 : 95 between the agricultural and the non-agricultural sectors respectively. (23) Electrical Machinery and Equipment. The revenue was distributed in the proportion of expenditure on electricity. (24) Iron and Steel and their products. It was assumed that 70 per cent of the revenue comes from the non-agricultural sector whereas the remaining 30 per cent was distributed in the proportion of expenditure on durable household effects. (25) Textile Articles, Carpets, Matting, Lace and other Furnishing Fabrics, Yarn and Fabrics of Man-Made Yarn, Fabrics of Wool Flax, remic metal and vegetable goods and Cotton Yarn and Fabrics. The revenue from all of these categories was distributed in the proportion of expenditure on cloth and its accessaries. (26). Cutlery tools and other articles of base metal. The revenue was distributed in the ratio of expenditure on durable items of cutlery and crockery. (27) Machinery and Mechanical Appliances. Same ratio as in the case of (24) was used to distribute the revenue. (28) Motor and Other Vehicles. A small proportion of agricultural households owns or uses motors or other vehicles, therefore, the burden of such taxes on rural household is insignificant. It was assumed that 80 per cent of total revenue comes from the non-agricultural sector and the remaining 20 per cent was distributed in the proportion of expenditure on transport and travelling. (29) Ships, boats and aircrafts. Rural people rarely travel on ships, boats or air crafts. It was assumed that 80 per cent of the revenue comes from the nonagricultural sector—the remaining 20 per cent was distributed in the proportion of expenditure on transport and travelling. (30) Clocks, Watches and their Parts. Ratio of expenditure on miscellaneous items was used to distribute the revenue. (31) Railway and Transway Plant and Rolling Stock. It was assumed that 50 per cent of total revenue comes from the non-agricultural sector while the remaining 50 per cent was distributed in the proportion of expenditure on transport and travelling. (32) Toys, Games and Sports Goods. The revenue was distributed in the proportion of expenditure on miscellaneous items. (33) Miscellaneous Items. Revenue from taxes on miscellaneous items was distributed in the ratio of expenditure for these items.

CHAPTER VII

#### SALES TAX ON DOMESTIC AND IMPORTED GOODS

Data on commodity-wise collection of sales tax on imports, sales tax on manufactures and sales tax on exciseable items are available for the years 1972/73 to 1975/76 whereas aggregate figures for sales tax collection are available for remaining years of the study. Revenue from each of the taxes was divided between the agricultural and the non-agricultural sectors on the basis of consumption pattern. The basis for sectoral distribution of revenue were the same as those for import duties and central excise duties. An average percentage share of each of the sectors was calculated and used to distribute the revenue for other years of the study.

### PROVINCIAL EXCISE

The revenue comes mainly from intoxicants, alcholic beverages and non-alcholic beverages. It was distributed between the sectors in the ratio of sectoral expenditure on food and drinks.

#### CHAPTER VII

# NET FISCAL BURDEN ON AGRICULTURAL PROBUCERS

#### 1. INTRODUCTION

The finding that the agricultural sector in Pakistan is overtaxed relative to the non-agricultural sector would be a convenient talking point for the agricultural lobby that is always intent on blocking any measures for the increased taxation of the agricultural sector. The case for higher overall taxation is built on the developmental requirements of resources in the public sector. The high income inequities and the high levels of expenditure by the rich in both rural and urban areas imply the existence of taxable capacity measured in absolute terms on both sectors of the economy. Adherence to the principle of "equitable" taxation is always an important consideration in an evaluation of any tax system. From the vantage point of the development prospects of any economy, it is also important to have an idea of the direction and extent of resource transfer from a sector. The transfer is a measure of the overall incentives being provided to a sector. The purpose of the chapter is to describe the open and concealed taxes and subsidies which affect Pakistan's agriculture and to estimate their magnitude for the period 1972/73 to 1983/84.

### 2. OPEN TAXATION OF THE AGRICULTURAL SECTOR

Two different taxes affect the farmers directly while there are a number of indirect taxes on agricultural commodities that affect prices received by agricultural producers. Table 7.1 presents the picture of open taxation of farmers.

there are decorated between the case of a section of

ar sent to the start into read horse an

1983-84	2000	1987-82	19-00-01	08-6761	1978-79	81-1163	1976-77	61-6167	19/4-75	1973-74	1972-73	14 6 6 7		9				-1	79-	
169		260	2770	172	231	117	136	161	189	187	168	Land Revenue (Million Rupees)	DISECT							
,	,	1	. 9	1	w	-		O	4	U	ŧ.a	Agricultural Incoma Tax (Million Rupses)	LAXES							
691	252	187	235	172	234	110	136	203	193	192	171	Total (Million Rupæes			н					
252	706	051	500			-	p=1	340	535	179	442	Export Duty on Cotton (Million Rupecs)	COM		INCIDENCE					
1		-	-	1	1	1 2	500	241	333	464	1-	Export Duty on Rice (Million Rupaes)	COMMODITY		08					4
46	11	-237	723	151	-152	-220	141	- 61	-228	- 76	-	Profits of Cotton Export Corpora- tion		1	DIRECT					
												(Million Rupses) Profits of Rice			AND					
310	341	596	853	589	546	209	44	187	113		1	Export Corpora- tion (Million Rupess)				Table		34	- 14	7
00	134	116	108	103	47	52	88	65	60	59	1	Cotton Fee (Million Rupees)			XIIOWAKO	VII. 1	(9)		. 1-4	e l
		625										Total (Million Rupees)			TAXES	1		10		
3.89	0.00	0.00	5.71	4.26	5,90	3,08	3,55	5.40	5.24	5.23	4.76	Direct (Rupees)	PER CA		NO					
. 90	00	14.91	06	86	H	.07	68	55	.07	.07	.87	(Rupaes)	CAPITA TA		AGRICULTURAL					
19.79	19.13	21.76	58.77	25.12	17.01	4.15	12.23	25.95	27.31	22.30	20.63	Total (Rupess)	TAXES	( A./   )	TURAL					
8.18	12.30	14-12	11.58	8.47	11.71	5.87	6.88	10.24	9.87	9.91	8.94	Direct Taxes per Hectere (Rupees)			PRODUC					
33.46	49.88	30.74		41.67	22,07	2.04		38.95	41.59	32,30	29 , 81	Commodity Taxes Per Hectare (Rupees)			CERS					
42	62		119	50	33	7.91	23.68	49.19	51.46	42.21	38.75	Total Taxes Per Hectars	-					, de		
0	0	0.35	0	0	0	0	0	0	0	0	0	Ratio of Direct T to GDP from Agric ture (Per cent)	ul							
0.77	1.14	0.77	3.12	1.35	0.82	.0.08	0.76	2.01	2.42	2.23	2.60	Ratio of Commodit Taxes to GDP fro Agriculture (Per cent)								
0.96	1.42	1.12	3.46	1.62		0.31		2.54	2.99			Ratio of Total Ta to GDP from Agriculture (Per cent)	xe	9						

Of the direct taxes, land tax generally known in Pakistan as land revenue is the oldest agricultural tax. Agricultural income tax is also a misnomer as it is not a tax on agricultural incomes as conventionally understood. It is a graduated surcharge on land revenue with a high exemption limit and low rates. The yield from land revenue, or the other hand, is much larger than is the case with the agricultural income tax. The yield from the land revenue goes up from Rs. 168 million in 1972/73 to Rs. 286 million in 1981/82 but declines to Rs. 169 million in 1983/84. The incidence of direct taxes per capita or per hectare is low and fluctuates over a narrow range. Direct taxes as a proportion of gross domestic product originating from the agricultural sector have shown a declining trend over the period of study.

The indirect taxes comprising export duties on cotton and rice and profits of rice and cotton export corporations are by far more important revenue raising devices than the direct taxes on farmers. The revenue from the indirect taxes is, however, an instable source for the public treasury. Excepting the year 1980/81, the ratio of taxes to agricultural output shows a declining trend as it goes down from 3.38 per cent in 1972/73 to 0.96 per cent in 1983/84.

# 3. CONCEALED TAXATION OF AGRICULTURE

The agricultural producers in Pakistan have always been subject to concealed taxes. During the 1950s and 1960s, agricultural exporters suffered an income loss through an overvalued exchange rate. After devaluation in 1972, this source of concealed taxation was replaced by open taxes in the form of export duties and profits from government

monopoly trading corporations. In addition to the open taxes, farmers pay taxes (or receive subsidy) if the prices for their products are kept lower (higher) than those that prevail in the international market. The size of the tax (subsidy) to producers is measured by the difference between the value of output at domestic prices and the value of output at world prices.

wheat, Basmati rice and Irri rice. Calculations for other crops have not been made. Sugarcane is often cited as an example of a commodity receiving huge unbudgeted subsidy. For the three crops for which concealed taxes have been estimated, it is clear that producers have suffered a large cut in their income. Similar calculations for other crops and farm activities have not been made. It is not possible to get an overall idea of the extent of concealed taxation of farmers as a group. Need for detailed research is obvious and urgent.

Table VII.2

Concealed Taxation of the Agricultural Sector for Selected Crops

Years	Wheat	Basmati	Irri			Coefficients cernational	s defined as Ratio Prices
	(Million Rupees)	rice (Mil.Rs.	rice (Mil.Rs.)	Cotton	Basmati rice	Irri-6   rice	Wheat
1973/74	-4700	- 718	-1985 <sup>-1</sup>	0.62	0.46	0.35	0.34
1974/75		-1719	- 718	0.90	0.41	0.46	0.58
1975/76	-2517	- 892	- 235	1.02	0.61	0.77	0.56
1976/77	-1408	- 136	- 138	0.82	0.99	0.97	0.72
1977/78	-1608	- 137	876	1.17	0.93	0.75	0.85
1978/79		-1949	- 435	1,30	0.52	0.75	0.73
1979/80	-3788	-1702	-1094	0.97	0.56	0.74	0.61
1980/81	-3593	- 926	-1164	0.87	0.65	0.70	0.64
1981/82	-4548	-2054	18	1.13	0.66	0.94	0.75
	2-10-1		13				

Source: Kee-Cheok Cheong and Emmanuel H.D'Silva, Prices, Terms of Trade, and the Role of Government in Pakistan's Agriculture, World Bank Staff Working Papers No. 643, the World Bank, USA.

Note: - sign-indicates taxation and (+) sign indicates subsidization of the producer through the pricing of the commodity.

#### 4. OPEN SUBSIDIES TO FARMERS

Public policy in Pakistan has provided incentives to farmers through budgeted subsidies on a number of agricultural inputs. Starting in the early 1950s the government has provided a subsidy on land improvements, wells, soil conservation and land consolidation. Later additional subsidies were given for fertilizer, tubewells, plant protection, pesticides and seeds. Data on some of those subsidies are hard to assemble but for major inputs that account for a dominant proportion of open subsidies data are easily available. Table VII.3 gives an indication of the extent of subsidies received by farmers for the use of fertilizer and pesticides and installation of tubewells.

Table VII.3 Open Subsidies on Agricultural Inputs

****	All Inputs		Plant Pro and Pest	Tubewells	Fertilizer	Years
Tea.	345		116	22	207	1972/73
	351		63	9 10	278	1973/74
	452		112	16	326	1974/75
72.526	1012		381	24	607	1975/76
	914		485	48	381	1976/77
\$1797.5	1160		523	20	617	1977/78
	1983		267	24	1692	1978/79
	2694	1949	218	22	2454	19/9/80
	2477	926	0	20	2457	1980/81
	1818		0	24	1794	1981/82
	1972	ses prosit	0	24	1948	1982/83
	1690		0	0	1690	1983/84

The Photogram and the party and reported formation of

The amount of total subsidies increases from Rs. 345 million in 1972/73 to Rs. 2694 million in 1979/80 and declines to Rs. 1690 million by 1983/84. Fertilizer accounts for the major share of the total subsidy bill. In the early years fertilizer was directly subsidized by about 50 per cent. The rate of subsidy has been gradually reduced over time. The government is committed to eliminating the subsidy on fertilizer by 1985. Despite the declining rate of subsidy on fertilizer, the expanded use of fertilizer had resulted in a growing subsidy bill till 1980/81.

The subsidy on pesticides increased from Rs.116 million in 1972/73 to Rs.523 million in 1977/78 and declined to Rs. 218 million in 1979/80.

The subsidy was withdrawn in 1980/81. The subsidy on the installation of tubewells was modest in amount relative to the subsidy on fertilizer and pesticides. The subsidy on tubewells was also withdrawn in 1983/84.

The main motivation of the government in eliminating open subsidies on pesticides, tubewells and fertilizer has been the concern with the budgetary impact. Farmers are now aware of the advantages of these inputs. There is some evidence that suggests that the rates of return on the use of these inputs remain high after input prices increase as a result of the removal of subsidies.

#### CONCEALED SUBSIDIES

The cost of open subsidies to the treasury can be seen easily. As pointed out in the previous section, public policy to reduce the extent of open subsidies was instituted when it became evident that the subsidies were no longer required for their production impact and were a

burden on the exchequre. The concealed subsidies are difficult to measure as income transfer takes place between different economic agents in an implicit manner. Such subsidies are nevertheless important determinants of incentives. The agricultural producers have received concealed subsidies through the supply of cheap credit, irrigation water and electricity. The rate of subsidy per unit could be defined as the difference between the price at which an input for which subsidy is being measured would sell without public intervention and the price charged by a public agency after the government intervention. Posing the estimation problem this way shows clearly that any measurement of the subsidy would be subject to limitations of data and the method of analysis. In view of the large magnitude of implicit subsidies an approximate measurement is also an important piece of information.

#### 5.1. Supply of Low Interest Loans

Agricultural Development Bank of Pakistan, direct government lending through Taccavi loans, commercial banks and cooperatives. These sources of credit charge lower than market interest rates. Some of the loans by these institutions are also not repaid by agricultural borrowers which implies unintended subsidy in the form of defaulted loans. The average interest rate on the institutional sources of credit in 1972/73 was 8.16 per cent per annum while the internal rate on non-institutional sources was 15.34 per cent. Assuming that the opportunity cost of credit for the institutional sources is the rate of interest

<sup>1/</sup> Sarfraz K. Qureshi, Kalbe Abbas, Ahmed Naeem Siddiqui and Ejaz Ghani Rural Credit and Rural Development: Some Issues". Pakistan Development Review. Vol. XXIII, Summer-Autumn 1984.

charged by the non-institutional sources, the subsidy on credit is about 7.18 per cent per year.

Table VII.4 computes the subsidy on credit by assuming the subsidy rate at 5 per cent per year. We ignore the subsidy via defaulted loans and use a conservative estimate of the rate of subsidy. We also ignore the subsidy given to farmers through interest-free loans from the commercial banks. The subsidy on credit, which must be the lower bound, increases from Rs. 15 million in 1972/73 to Rs. 521 million in 1983/84.

Table VII.4

SUPPLY OF AGRICULTURAL CREDIT BY SOURCE AND ESTIMATES OF SUBSIDY ON AGRICULTURAL CREDIT

-		-		(in	Million Ru	pees)
Years	Agricultural Development Bank of Pakistan	Co-opera- tives	Taccavi	Commercial Banks	Total Annual Credit	Estimated Subsidy
(1)	(2)	(3)	(4)	(5)	(6)	(7)
1972-73	168.80	42.02	10.23	85.70	306.75	15
1973-74	415.20	143.72	67.50	286.40	912.82	45
1974-75	395.50	81.54	12.13	520.90	1010.07	50
1975-76	532.20	91.84	25.67	808.10	1457.81	73
1976-77	637.90	95.45	13.14	970.10	1716.59	86
1977-78	429.80	138.04	9.00	1290.92	1867.76	93
1978-79	416.94	413.78	11.96	1381.11	2223.79	111
1979-80	711.55	708.64	8.20	1587.40	3015.79	195
1980-81	1066.62	1126.25	8.30	1826.77	4027.94	286
1981-82	1557.38	1100.80	10.34	2436.10	5104.62	299
1982-83	2310.44	1320.93	2.69	2680.89	6314.95	379
1983-84	3131.67	1449.89	9.30	4088.70	8679.56	521

Source: Pakistan Economic Survey, 1984-85, for columns 2 to 5.

# 5.2 Low Irrigation Water Rates

There is a huge unbudgeted subsidy being given to farmers as they are being charged low irrigation charges on account of water being supplied through public canals. In a competitive market, the price of a product is the cost of supplying the marginal unit. The alternative to water supply through canals is water pumped from tubewells. There are studies that document the marginal cost of water supplied by tubewells. There are also studies that report the market price of tubewell water. The subsidy on water could be calculated by comparing the price of water charged by government with either the marginal cost of tubewell water or the market price of tubewell water. Calculating the subsidy in this way shows the enormous extent of subsidy on irrigation water. Since we do not wish to err on the high side, we have estimated the subsidy on water as the difference between the operation and maintenance expenditure incurred by the government on public canals and the irrigation receipts.

Table VII.5 presents subsidies on irrigation water. It is found that the small surplus in the years 1972/73 and 1973/74 soon turns into deficit and increases to Rs. 926 million for the year 1983/84.

Table VII.5
Subsidy on Irrigation Water Supplied through Canals

Years	Gross Irrigation Receipts	Expenditure on Operation Maintenance of Canals	(in Million Rupees   Net Irrigation   Receipts
1972-73	308.5	271.0	37.5
1973-74	377.2	301.4	75.8
1974-75	341.1	419.4	(-)78.3
1975-76	370.9	507.0	(-)136.1
1976-77	384.8	534.9	(-)150.1
1977-78	497.3	618.0	(-)120.7
1978-79	496.3	727.2	(-)230.9
1979-80	575.7	936.7	(-)361.0
1980-81	635.9	1152.4	(-)516.5
1981-82	884.1	1486.1	(-)602.0
1982-83	974.1	1669.0	(-)694.9
1983-84	1027.3	1953.5	(-)926.2

### 5.3 Subsidy on Electricity

The farmers in Pakistan get electricity at a concessional rate. Certain concessions are also given to farmers on installation and minimum use charges. Data on these types of concessions are not available. However, an estimate of subsidy on account of a favourable rate can be made.

Table VII.6 presents the data for the computation of subsidy. The cost\_of supplying electricity to farmers is always higher than the average sale price of electricity for agriculture. The subsidy on electricity goes up from Rs. 42 million in 1972/73 to Rs. 275 million in 1983/84.

Table VII.6

Estimated Subsidy on Electricity Supply to the Agricultural Sector

Years	Electricity Consumption in the Agri- cultural Sector (M.Kwh.)	Cost per Kwh/Paisa	Average Sale Price for Agriculture Paisa/Kwh	subsidy Per Kwh (Paisas)	Estimated Subsidy (in Million Rupees)
1972-73	1170	13.59	9.96	3.63	42
1973-74	1131	16.91	10.71	6.20	70
1974-75	1531	23.14	11.99	11.15	171
1975-76	1386	26.62	15.53	11.09	154
1976-77	1400	29.95	15.71	14.24	109
1977-78	1717	27.41	14.37	13.04	224
1978-79	1666	28.23	20.42	7.81	130
1979-80	2056	31.75	27.85	3.91	80
1980-81	2125	33.26	31.16	2.10	45
1981-82	2357	41.45	34.66	6.79	160
1982-83 1983-84	2546 2750	45.52 48.50	36.01 38.50 ghth Issue, Pla	9.51 10.00	242 275

Source: "Power System Statistics" Eighth Issue, Planning Department. Power Wing, WAPDA, Nov. 1983.

# 6. NET FISCAL BURDEN ON FARMER-PRODUCERS.

A consolidated picture of agricultural taxes and subsidies given in Table VII.7 shows up a number of important conclusions which are briefly listed. Firstly, an unsatisfactory performance regarding resource mobilization is sharply brought out as not taxes are positive for 1972/73, 1973/74 and 1974/75 and negative for all the remaining years. It should, however, be pointed out that concealed taxation of farmers has not been included in the computation of the net fiscal burden. On the other hand, it should also be noted that we have greatly under-estimated the concealed subsidies on credit and irrigation water. Secondly, the incidence of taxes forms a small proportion of gross domestic output from agriculture. There is a declining trend in the tax burden. It is clear that the agricultural sector has not contributed much to the resources required for financing development. Third, the yield from taxes on commodities shows a large element of instability while the yield from direct taxes is stable. Finally, open subsidies on agricultural inputs are being eliminated. However, concealed subsidies on inputs are increasing with the passage of time.

468         159         675         1015         2           1         58         - <td< th=""><th></th><th>1972-</th><th>1973-</th><th>1974-</th><th>1975-</th><th>1976-</th><th>1977-</th><th>1978-</th><th>1979-</th><th>1980-</th><th>1981-</th><th>1982-</th><th>Rupees) 1983-</th><th>1</th></td<>		1972-	1973-	1974-	1975-	1976-	1977-	1978-	1979-	1980-	1981-	1982-	Rupees) 1983-	1
Tt Corporation  Tt Corporation	TAXES	741	818	1006	975	468	159	675	1015	2419	912	1268	861	
Tr Corporation 76 -228 -61 141 -220 -152 151  - 76 -228 -61 141 -220 -152 151  - 59 60 65 88 52 47 103  168 187 189 197 136 117 231 172  ome Tax 3 5 4 6 - 1 3  207 278 326 607 381 617 1692 2454 2  207 278 326 607 381 617 1692 2454 2  ion & Pesticides 116 63 112 381 485 523 267 218  42 70 171 154 199 224 130 80  -37 -76 428 253 -400 -881 -1439 -1780 -2315 -4	Rew Cotton	128	179	333	241	00 erd	1-1	1 1	1 1	200	150	1 797	100	
Ome Tax	Profits of Export Corporation												100	
ome Tax     365     187     189     197     136     117     231     172       ome Tax     3     5     4     6     -     1     3       207     278     326     607     381     617     1692     2454       207     278     326     607     381     617     1692     2454       22     10     16     24     48     20     24     22       22     10     16     24     48     20     24     22       24     48     20     24     48     20     24     22       45     50     73     86     93     111     195       42     70     171     154     199     224     130     80       -37     -76     78     136     150     121     231     361       376     428     253     -400     -881     -1439     -1780     -2315	Rice (RECP) Cotton (CECP) Cotton Fee	111	-76	113 -228 60	187	44 141 88	209 220 52	546	589	853 723 108	596 -237 116	347	310 46 83	
ome Tax 3 5 4 6 - 1 3 - 1 3	Land Revenue	163	187	189	197	136	1117	231	172	226	285	249	169	
365 390 753 1375 1349 1598 2455 3330 207 278 326 607 381 617 1692 2454 22 10 16 24 48 20 24 22 10 15 24 73 135 135 135 135 267 218 15 45 50 73 86 93 111 195 42 70 171 154 199 224 130 80 -37 -76 78 136 150 -1439 -1780 -2315	Agricultural Income Tax	m	5	4	9	M. A	1	63	1	01	7	7	1	
207 278 326 607 381 617 1692 2454 22 10 16 24 48 20 24 22 22 10 16 24 18 20 24 22 24 22 25 10 16 53 112 381 485 523 267 218 25 45 50 73 86 93 111 195 42 70 171 154 199 224 130 80 -37 -76 78 136 150 .121 231 361 376 428 253 -400 -881 -1439 -1780 -2315	SUBSIDIES	365	390	753	1375	1349	1598	2455	3330	3324	2879	3288	3412	
ion & Pesticides 116 63 112 381 485 523 267 218 15 45 50 73 86 93 111 195 42 130 80 -37 -76 78 136 150 .121 231 361 376 428 253 -400 -881 -1439 -1780 -2315	Fertilizer	207	278	326	607	381	617	1692	2454	2457	1794	1948	1690	
15 45 50 73 86 93 111 195 42 70 171 154 199 224 130 80 -37 -76 78 136 150 121 231 361 376 428 253 -400 -881 -1439 -1780 -2315	Plant Protection & Pesticides		63	112	381	485	523	24	218	20 M A	24	24	N.A.	
42 70 171 154 199 224 130 80 -37 -76 78 136 150 121 231 361 376 428 253 -400 -881 -1439 -1780 -2315	Credit	15	45	90	73	98	63	111	195	286	299	379	521	
376 428 253 -400 -881 -1439 -1780 -2315	Irrigation	-37	-76	171	136	199	224	130	361	45	160	242	275	
	NET TAXES (A-B)	376	4.28	253	005-	-881	-1439	-1780	-2315	-905	-1967		-2551	

Note: "-" indicates entries as insignificant or zero.

notated from

5.

9

t'

# CHAPTER VIII POLITICAL ECONOMY OF LAND TAXES

#### 1. INTRODUCTION

Taxation of the agricultural sector is a major instrument for mobilization of the surplus to finance development projects within the agricultural sector and/or the rest of the economy. For many years the need for a heavier taxation of agricultural land has formed part of the conventional wisdom about the ways to extract the agricultural surplus and to increase the tempo of agricultural development in poor countries. Land taxes have both equity and efficiency properties that gladden the hearts of economists and vocal politicians belonging to urban areas alike. Taxes on land promote efficiency in the allocation of scarce resources by creating incentives for farmers to increase their effort and reduce their consumption, expanding the amount of agricultural produce available to the non-agricultural sectors of the economy. A tax on land has an important redistributive function because its incidence falls squarely on the landlord and is neither shifted forward to consumers nor backwards to suppliers of agricultural inputs, nor does it introduce distortions in the allocation of productive resources.

The conventional wisdom has, had a difficult time in becoming conventional practice. The share of land taxes in total tax revenues is generally low in developing countries and its share has been declining. In Pakistan, the share of land taxes in total taxes has fallen from 16 per cent in 1960 to 6 per cent in 1970, to 2 per cent in 1975 and

finally to just .5 per cent in 1982. As a share of total agricultural incomes, land taxes have shown a similar downward trend. The decline stems in part from the fact that rates and assessments have not kept pace with the growth of agricultural incomes, but in some cases the taxes on land are either being abolished or rendered ineffective by exemptions and archaic assessment procedures. In India, some states have simply eliminated the land tax. In Pakistan, the land tax has recently been seriously diluted by the exemption of irrigated land holdings below 12.5 acres and unirrigated holdings below 25 acres.

The growing gap between theory and practice has its roots in the balance of political forces that determine tax policy in developing countries. It is axiomatic that interest groups will attempt to shift the burden of taxation to others while reaping, to the maximum extent possible, the benefits of government expenditure. Since the agricultural sector is the dominant political force in many of the poorest developing countries, the erosion of land taxes in particular and taxes on rural incomes in general is almost an inevitable outcome in a world where voters and political leaders act on the basis of personal benefit-cost calculation, as eloquently analysed by Anthony Downs  $/2\sqrt{2}$ . In countries where rural interest groups predominate, the relevant question is then how far can the agriculture sector shift the burden of taxation on to other groups.

The recent experience of Pakistan offers some interesting perspectives on the political economy of agricultural development because the benefits of government development expenditures to the

rural sector have undergone a dramatic change in the past two decades. In the early 1960s yields began to rise as a result of tubewell-supplied irrigation water and increased fertilizer applications, both heavily subsidized by government. In the second half of the decade of the 1960s, these new inputs were joined by new seed varieties, whose multiplication and distribution were also the responsibility of government. One would expect, therefore, that, the farmers' resistance to land taxes or other forms of direct taxes would have weakened.

But the share of taxes in Pakistan's GNP and the rural sector's share of those taxes have shown no tendency to rise. One explanation for this phenomenon, at least in the case of Pakistan, stems from another Downsian principle, namely that it is the <u>perceived</u> benefits and costs which are relevant to voters in deciding how they will cast their ballots. From the point of view of the rural voter, a tax on his land is direct and real while the benefits are diffuse, sometime intangible, and frequently vary.

If the perceived private returns to government expenditures could be raised, it follows that the taxation of land might become not only politically palatable but even actively sought by farmers as a mean of raising their income.

## THE SYSTEM OF LAND TAXES IN PAKISTAN

The present land tax system in Pakistan has its origins in the British tax system introduced to the subcontinent in the 18th and 19th centuries. The British system was based on the classic economic principle that rent or the net output of land (i.e. the difference

between gross output and costs of the variable inputs) is an unearned source of income for landlords and can be taxed without any adverse effect on production. Over the centuries, the British developed an elaborate administrative machinery for administering the land tax system. Cadastral surveys were painstakingly conducted to measure land productivity. Data on crop prices and cultivation costs were compiled for different geographical sub-units. The assessment of land rental values (generally known in the subcontinent as settlement) was first made by the British in the second half of the 19th century and have been periodically revised since then. A general reassessment was carried out in the 1920s and 1930s and special reassessments have subsequently been performed in some districts. Except for a few agriculturally poor districts in the Baluchistan province. Northern areas and some of the former princely states, Virtually all areas in Pakistan have been surveyed at least once for land tax purposes.

Two main systems of land tax assessment have evolved. The Punjab system is applicable to the provinces of Punjab, N.W.F.P., and most of Baluchistan. For the purposes of assessment, each district is treated separately. The district is further sub-divided in clusters of much smaller areas for the purpose of assessment. In principle, the assessment area — or Patwar circle, as it is commonly known — is supposed to be roughly homogenous with respect to the general agricultural conditions, and an equitable assessment rate is fixed for all land lying within the area. The fields within the patwar circle are measured and classified according to soil type. The tax rate for each plot of land is then geared to the quality of the soil and expressed as a multiple of the assessment rate for the area. The basis

of the assessment is net produce accruing to the landowner. The gross value of output on each field is ascertained by multiplying the yield per acre, average acreage under each crop and the average price for each crop in the village market. The net produce (rental value) is obtained by deducting the share going to the tenant and costs incurred in cultivation by the landowner from the gross value of production.

The assessed values remain fixed between the settlements that are generally made in each district at 40 year intervals. In the three districts of Lahore, Faisalabad and Sahiwal, the assessed land values are reduced if the price of the main crop in the district for any year is lower than the price at which the physical output was valued during the year of tax settlement. If the price is higher, the assessed values remain at the level fixed in the settlement.

The rate of tax is fixed at the time of tax settlement. This rate is applied to the rental value of cultivated land and land left uncultivated does not incur any tax liability.

In areas other than the ones covered by the Punjab System, a slightly different method known as the Sind System is applied for fixing the assessed land values. This system is closer to the Ricardian doctrine of land taxation. The tax settlements are more frequent as compared with the Punjab System. For cotton, paddy and wheat the rate of assessment per acre varies with \_\_\_\_\_ crop prices in both the rising and falling phases of price changes. For other crops, the assessments remain fixed for the period between the two tax settlements. In theory, the Sind System is relatively more income elastic than the Punjab System.

An outcome of the efforts by the British to assess land for taxation has been a legacy of an elaborate system of land tax assessment and collection and maintenance of records of land rights. The administrative advantages of this legacy are obvious for strengthening the land tax system in Pakistan.

# 3. ANOMALIES IN THE LAND TAX SYSTEM

The administration of the land tax system in Pakistan contains a number of anomalies, the first and most important of which is the infrequent revision of the assessment rate. There has been no general revision in land assessments since the 1930s although in some districts changes in the assessment rates are made annually. In Lahore, Faisalabad, and Sahiwal districts in the province of Punjab, assessment values are reduced if the price of the main crop in the district during the tax year is lower than the price of this crop in the year of the original tax assessment. This procedure is not symmetrical, however, and higher crop prices do not affect the assessment rate. In Sind, the tax administrators have attempted to revise tax assessments more frequently by taking into consideration the fluctuations over time in the prices of the principal crops grown in their province. Such efforts have proved difficult because landowners there constitute an even more powerful political force than elsewhere in Pakistan.

A second anomaly of the system is that the rate and the tax base are fixed at the same time. A variable tax rate would be one way to compensate for annual fluctuation in crop prices. Nevertheless, the British fixed a statutory tax rate in the original legislation that established the land tax system and the tax rate cannot be changed

without amending this legislation. Prior to 1871, the rate of taxation was fixed at two-thirds of the assessed rental value, and the Land Revenue Act of 1871 reduced this rate to 50 per cent. In the 1920s the tax rate was revised along with the assessed evaluations, and the rate was reduced from 50 per cent to 25 per cent.

An additional anomaly in the procedures for establishing assessment rates is the legal ceiling on increases in the assessment rate for individual farms in the Patwar circle. The Land Revenue Act of 1928 stipulated that no assessment could be revised upwards by more than two-thirds for any single parcel of land or by more than one quarter for the assessment area as a whole. This limitation would not prove to be a substantial barrier if there were more frequent revisions in land assessments. But the tendency to revise land assessments at forty year intervals and the secular increase in agricultural prices virtually assure that land taxes will not rise as fast as increases in agricultural income in Pakistan.

Another anomaly is the narrow definition of the net produce of land used in establishing the assessment rate. No attempt is made to include the income generated by livestock or other on farm activities. Revenue from these activities escapes all taxation because agricultural income is excluded from payment of income taxes.

# 4. EQUITY ASPECTS OF LAND TAXES

It is generally believed in Pakistan that land tax is a regressive and an inequitable levy. It is argued that since tax settlements were done at different times in different areas, it is likely that wide variations in effective rates prevail. The horizontal and vertical

inequities in the land tax are ascribed to the outdated assessed land values. The opposition to increases in land taxes has argued that the design and administration of a progressive land tax is a difficult task and the present system is inequitable therefore the burden of the land tax should not be increased by ad hoc increases in rates and/or assessed land values. In fact, the argument has often been made that since flagrant inequities are tolerated only when the tax is light, there is a need for lowering the land tax rates in Pakistan.

No empirical study on the incidence of land taxes has tested the hypothesis of this tax being a regressive one in Pakistan. The land tax system in Pakistan is badly in need of reform if it is to play any significant role in the generation of government revenues. Given the configuration of political forces in Pakistan, one important consideration in any reform of the land tax system will be its incidence. Given the anomalies described above, it is impossible to say on a priori basis whether the land tax system is progressive, regressive or neutral. A direct test of the progressivity of Pakistan's land tax system can be made by examining the share of taxes in the individual income for landowners of different income classes. However, in the absence of appropriate household income data for landowning families, it is necessary to use more aggregative data. To determine tax progressivity, the share of land taxes in each district's income has been regressed against the per capita agricultural income of 37 districts of Pakistan.

The data needed for the analysis are: (1) tax collections by districts, (2) agricultural income by districts, and (3) agricultural labour force by districts. Before we present the result, a brief account of the data sources and their quality is given.

Regressions of percentage of income paid in taxes on Per Capita Income.

quation No.	Sample		1	stimated E	quation	R <sup>2</sup>	e
1.	37 Districts	T/Yg	. 325	01218 +	.00002 Y <sub>g</sub>	.36	2.039
2,	37 Districts	T/Yn		-*01365 +	.00003 Y <sub>n</sub> (5.17161) <sup>n</sup>	.44	2.22
3.	19 Below Mean Income Districts				.00001 Y <sub>g</sub> (1.912)	.18	1.48
4.	19 Below Mean Income Districts	T/Y <sub>n</sub>		00453 +	.00001 Y <sub>n</sub> (2.495)	.27	1.05
5.	18 Above Mean Income Districts	T/Yg		01509 +	.00002 Y <sub>g</sub>	.17	1.63
6.	18 Above Mean Income Districts	T/Yn	m	0177 +	.00003 Y <sub>n</sub>	.24	1,81
7.	12 Districts under Sind System	T/Yg	22	00283 +	.00002 Y <sub>g</sub>	.31	1.61
8.	12 Districts under Sind System	T/Yn	9	00211 +	.00002 Y <sub>n</sub>	,30	0.88
9.	25 Districts under Punjab System	T/Yg	12	00020 +	.00001 Y <sub>g</sub>	.24	1.72
10.	25 Districts under Punjab System	T/Y <sub>n</sub>	=	.00024 ÷	.00001 Y <sub>n</sub> (2.637)	.23	2.94

Notes: 1) T,  $Y_g$  and  $Y_n$  are respectively per capita tax, per capita gross income and per capita net income.

The second secon

<sup>2)</sup> Figures in parentheses are t-values of the coefficients.

elasticities of tax-in two related of the local value are presented in the table.

Equations 1 and 2 relate to the 37 districts of Pakistan. The coefficients are highly significant and positive. The values of the elasticity are higher than 2. The land tax is highly progressive as between districts. The proportion of variance explained is also quite high.

We tested against the possibility that the progressivity found in the land tax may be just a statistical artifact. A scatter diagram between the percentage of income in taxes and district per capita income did not show any distinct clusters between high and low income districts. Nevertheless, we divided the districts in two categories those above or below the mean district per capita income.

A separate equation was estimated for each category of districts. The results are presented in equations 3, 4, 5 and 6. The coefficients are significant and positive for each type of district. The elasticities are greater than one.

Separate equations for districts belonging to the Sind and the Punjab Systems of land taxation were estimated. The results are presented in equations 7. 8, 9 and 10. The coefficients are positive and significant for both systems. The Punjab System is relatively more progressive than the Sind System. This is in contradiction with theoretical expectation. This finding may be explained by the nature of the administration of land taxes as well as to different political power structures in the two regions. The Sind System may be administered in a way that large land-owners pay relatively lower taxes than under the Punjab System. This probably explains the reversion to the Sind System in the

Sind province after the break-up of one unit in West Pakistan in 1968/69.

In the new provincial Assembly in Sind, the Sindhi landowners could and of it lagislate for the reversion to the Sind System.

#### ES. EFFICIENCY ASPECTS OF LAND TAXES

Economists often stress the beneficial effects of increased I and taxes on agricultural output but the politicians generally argue that additional taxation adversely affects agricultural productivity. Thus section examines the effects of an increase in the burden of land tax, given the level of government expenditure benefiting the agricultural sector, on agricultural production in Pakistan. The crosssection data of different districts in Punjab for four years are analysed. We have regressed land tax per cultivated acre  $(X_1)$  and three alternative proxy variables ( $X_2$  or  $X_3$  or  $X_4$ ) for government expenditure on the value of agricultural output per cultivated acre. The data sources for values of agricultural output, taxes and cultivated area were described in the previous section. No data are available for government expenditure on agriculture on a district basis. We have used three proxy variables. The government has provided subsidized credit to farmers to buy fertilizer. install tubewells and purchase tractors and other items of agricultural machinery. Institutional credit in each district as a proportion of to tal institutional credit advanced to farmers in Punjab  $(X_2)$  and Institutional credit per acre in each district  $(X_3)$  are proxy variables for government expenditure benefitting the agricultural sector. The relative acreage under high yielding crop varieties is positively related to government expenditure on subsidized water, credit, fertilizer and, probably, agricultural machinery. Acreage under high yielding varieties of wheat as a per cent of total wheat acreage in each

district  $(X_4)$  is a proxy variable for government expenditure in year 1970/71 and 1972/73.

Table VIII-2 presents the regression results. The land variable is positively and significantly related to the agricul output per acre for all four year: in all equations. Governmen expenditures on agricultural development are positively and sig cantly correlated with the agricultural output per acre for eig of ten equations. For the two equations the variables are rela positively but the government expenditure variable is statistical insignificant. It is thus safe to conclude that upward revision the level of land tax would help to increase agricultural produths beneficial effect would be reinforced if the revenue from it land taxes is spent on agricultural development via increased go expenditure benefitting the agricultural sector.

## 6. POLITICS OF LAND TAXES

The archaic and inflexible land tax system in Pakistan contraced to the political dominance of the landowning class who, all as a matter of principle, have resisted any reforms in the land is system. More than 40 per cent of the voters in Pakistan own agritural land but their representation in elective bodies has always more than proportional to their numbers because of the influence they exercise over tenants and landless agricultural labour. The political power of landlords was apparent in the earliest days of

The political influence of landlords is felt not only at the tive stage but also in the administration of existing land tax the collection of land taxes is the responsibility of the district admining and specifically of the District Commissioner, or the Collector sometimes known during the British period. The District oner, however, is required by law to consult with members of the munity. This practice has provided considerable level rage to 1 community in the area of land tax administration. A fior to adduction by General Ayub Khan of the system of Basic D mocracies local government took the form of a Panchayat which invariably

an interesting analysis of the political importance of s, see Shahid Alam, "Economics of the Landed Interest the landed y of Pakistan", Pakistan Economic and Social Review, — A Case a view that land reforms have been hindered by these pring 1974. Herring, R. and Chaudhry, M.G., "The 1972 Land Reform their Economic Implications: A Preliminary Analysis" ame forces, in Pakistan Interest the landed — A Case pring 1974.

NO.	2 3 3	cept	ν.	X2 X3	X,	Degrees of	55	
	1958/59	+	9.086 + 1.	(3)	77	reedon 19	.72	
2.		17,348 + 9	2011	+ .563		19	. 67	
r°	1959/60	87,440 + 19	+	1,783	Heart of the second sec	10	.61	
		89,018 + 10	10,747 (4,312)	(1,433)	Panal Panal	16	.59	
ń	1970/71	85,427 + 29	29,098 + 6,714 (3,120)	20)		6	.67	
		84,684 + 32	32,123 (4,655)	+ 2.541		6	.61	
1.00 0	E SS	64,144 + 19	19.812. (4.250)		+ 1.931	19	. 65	
ού.	1972/73	149,700 + 55,	55,929 + 8,525 (5,819) (3,314)	25		19	.74	
o'i	n	126,265 + 61	(0.403)	+ 3,937 (2,452)		19	.74	
.0.		102,466 + 39.	39,352 (5,203)		+ 2,783	61	. 94	
Notes:	(c) (d) (d)	The figures in parenthe  X <sub>1</sub> - Land tax per acre  X <sub>2</sub> - Institutional cred  X <sub>3</sub> - Institutional cred  y	parenthesis arrer acre (in rug na! credit in a anced to farmen	1 0 1	e coefficients.	total institutional	na l	

was made up of the representatives of the major land-owning clients in each village. With the advent of the Basic Democracies system, the balance of political nower shifted in form but not in substance. The chairmanship of the union council was generally rotated among the largest landlords while the membership was made up of smaller landowners.

Land owners not only controlled the administration of the land tax system through the political process but they themselves are frequently the administrators. At the Towest Tevel, the local official responsible for the collection of the land tax is the Namberdar who is, in almost every case the landlord owning the largest amount of land in the village. At higher levels the tax officers at the district, divisional, provincial and federal levels are civil servants, the great majority of whom have ties to the landowning classes. Because of the historic inequalities in the education system, the sons of the landlords have had easier access to higher education than sons of other rural classes and were able to enter the civil service more readily than any other group. Similarly, the military, which at times has been a dominant political force in Pakistan, draws its officer corps mainly from the landed classes. Even though Pakistan was endowed at the time of its creation with a civil service in which rank was based on merit and not social background, the fact that so many government officers had direct family ties to the landed classes has made effective reform and implementation of land tax laws difficult to achieve.

The tendency on the part of a local tax officers to minimize tax liabilities on land can be seen from a recent case study of the operation of land tax system in Multan district  $\sqrt{107}$ .

The settlement officer in 1966 underestimated the prices and yield per acre. The measurement of matured area in the circle was accurate. The costs of cultivation were under-estimated. The underestimation of prices, costs and physical output cannot be explained because reliable data on these aspects are available from other government agencies. The yie'd data published by the government originates with revenue officials who are also responsible for the settlement operations. The extent of underestimation is massive. The gross value of crop output in 1966 settlement for the assessment circle was 12 million rupees. An alternative estimate based on more accurate price and yield data raises the value of gross output to Rs. 33 million. Net income is similarly underestimated. The theoretical upper limit for land tax per matured acre was Rs. 14.16 in 1966 settlement. An alternative estimate based on realistic figures of yields and prices would have yielded the tax of Rs. 31.09 per acre. The rate per matured acre actually fixed in 1966 settlement was only Rs. 3.64. This is much lower than the theoretical limit as calculated by the settlement officer himself. Interestingly enough, the actual assessment in 1966 is the same as in 1921.

Two main factors explain the under-assessment of the land tax. The narrow and legalistic factor could have prescribed the limits beyond which the assessment on the circle and village level cannot be increased between two consecutive settlements. This explanation is not relevant to the 1966 settlement in Multan as the settlement officer did not propose any increase in the assessment over 1921 level. This explanation does not have much weight as a

### CHAPTER IX

# CONCLUSION AND POLICY IMPLICATIONS

The foregoing analysis permits us to pinpoint several interesting and useful policy implications. These are briefly summarized below:

- 1. A tendency towards a reduction in income inequities during the 1960s generally heralded in Pakistan as the Green Revolution period and an increase in income inequities during the 1970s is clearly evident. On the other hand, the incidence of rural poverty shows an increase in the 1960s and a fall in the 1970s. Increased wage employment, non-farm employment and livestock are associated with a reduction in income inequalities. Remittances from overseas migrant workers have also reduced the incidence of rural poverty but have also paradoxically resulted in increased income inequalities.
- 2. The barter terms of the agricultural sector have shown an upward trend over the entire period from 1951/52 to 1983/84. However, this trend conceals in it periods of considerable decline, considerable increase and large fluctuations in the net barter terms of trade.

  There was a declining trend in the 1950s, a sharp upward trend in the 1960s, large fluctuations around a rising trend in the 1970s and a declining trend since 1977/78.
- 3. The extent of improvement in the purchasing power of agriculture as evidenced by changes in income terms of trade and single

factoral terms of trade since 1953/56 is large. Farmers' ability to pay taxes and their capacity to pay for modern inputs has improved considerably. This fact should be noted in debates on reducing farm subsidies or increasing the tax burden on agriculture. In an absolute sense, the taxable capacity and the purchasing power of the agricultural sector has certainly shown a considerable improvement. However, the limitations of any terms of trade index may be noted as it can be argued that the purchasing power of agriculture may worsen in relative terms if other sectors show a more rapid growth in their output or productivity.

- 4. Government policies on prices for crops and inputs have been based on a number of factors. A mere glance over the terms of reference of the Agricultural Prices Commission would show the varied nature of considerations guiding the policy matter in this area of public policy. The need to provide incentives for adoption of improved technology, the need to ensure efficient use of resources, the impact of prices on different agents in the economy and the need to take into account the changes in the terms of trade between crops, between output prices and input prices and between gross income and input prices are some of the basic considerations to be kept in view by the Commission at the time when it evolves its recommendations for appropriate prices for different commodities.
- by the fixed price if the farmer is to be provided an incentive to produce the crop. Current data on the costs of production of different crops are too patchy to be of much usefulness for policy purposes.

It would be helpful to be clear about the cost concept to be adopted as a basis for price determination. "It would also be useful to decide the level of aggregation at which costs are to be estimated. The issue at stake is whether the appropriate level for data estimation is the province, country or a region of homogenous crop mix.

- 6. In the case of the parity approach, we have shown that the gross-income input prices index shows a deterioration in the early to mid 1970s but has sustained a rapid rise since 1978/79. Government policies on prices of inputs and crops have, among other things, been based on the notion of a fair return to farmers. Changes in input prices are often followed by adjustment in output prices in the same direction. An increase in the farmers' income noted by us earlier, is also evident from the trends in the gross income-input prices index parity. In view of the shortage of pulses and oilseeds in the country, a case can be made against the present policy of maintaining a high gross income parity in favour of wheat vis-a-vis gram and oilseeds. In general an appropriate relationship needs to be maintained between costs and prices of and the income from different crops if specific targets of production by crops are to be fulfilled.
- 7. The impact of prices on the composition of output and on the total output of the agricultural sector is pervasive. The empirical evidence that the acreages of different crops vary systematically in response to intercrop price movements is abundant and shows that cropping patterns are a function of prices. The price elasticities for food crops are lower than those for cash crops, but even in their cases, a positive price responsiveness is indicated.

While acreage response functions have been subjected to detailed scrutiny, little work has been done on the output response. The elasticity of output with respect to price is the sum of elasticity of yield per acre and the elasticity of acreage. We have already noted that acreage elasticity is positive. If yields are sensitive to the relative prices of output and inputs, one should expect the output elasticity to be not only positive but larger than the acreage elasticity. Some evidence exists in Pakistan that farmers adjust the quantities of variable inputs, especially modern inputs, in response to changes in the input prices. However, no solid empirical work has been done in this area. Need for a careful analysis of the output response to the relative output-input prices is obvious and urgent and would be of direct policy relevance.

Another area of research which requires to be on the agenda for further research is the likely effect of prices on the "effective" supply of agricultural products for the non-agricultural sector. Cash crops are marketed almost totally. The price responsiveness of marketed supply needs to be estimated for different food crops. Studies on this issue are very few. A possible reason is the absence of time series data on marketed surplus. Only two studies were available. The price elasticity of wheat for market supply was found to lie between 0.3 and 0.7. A significant relationship between marketed surplus and food output was also noted. A rapid increase in the marketed surplus of some food crops in Pakistan, notwithstanding positive elasticity of marketable surplus with respect to prices, is ultimately explainable in terms of technological factors behind the observed increased output. Relative price changes were not of a magnitude that

could explain the market supply of foodgrains.

Although supply response, especially acreage response is a reasonably well researched area, no studies relating aggregate farm output to terms of trade are available for Pakistan. Some researchers have noted in passing that they had tried to establish a statistical relationship between terms of trade and value added in agriculture. The results were dismal and contrary to the a priori expectations and were not reported. We have estimated an aggregate output response function. The estimated results show a positive and significant (at 15 per cent significance level) relation between \agricultural output and lagged terms of trade. The short-run price elasticity of 0.18 was estimated. The shifter variables, like an index of irrigation and the time trend, were also found to be highly significant. The policy implication that emerges from the analysis calls for a favourable price environment for agriculture and a technology policy aimed at boosting output. In fact rising factor productivity in agriculture can lead to increasing profitability even when terms of trade do not rise very much.

8. The impact of agricultural price policies, on both the output and input sides, on income distribution has interesting and useful policy implications. That high farm prices benefit large producers is obvious. We have also found some support for the notion that high farm prices also benefit small farmers and landless labour. The farmers' capacity to hire labour is a function of farm prices. Benefits from subsidized inputs tend to accrue in large measure to large producers.

Reducing input subsidies would not greatly harm the small farmer and would release public resources that could finance investments for the benefits of small farmers. In this sense remunerative farm prices and low or no subsidies on farm inputs should increase the efficiency of resource use and the welfare position of the small farmer and the rural poor.

 Incentives to producers 'ould be given through attractive produce prices, technological innovations

and through investment in complimentary sectors. Detailed specific research on the relative benefits and costs of providing incentives through these policies are required. The theoretical and empirical aspects of determination of support prices for different crops that provide just the right amount of incentives and maintain appropriate price relatives for different crops should be given a high priority in the research agenda. Incentives, measured by the rural-urban terms of trade, are the outcome of a host of interacting sectoral and macro polities. It is important to ensure that the commodity-specific price policies and the macro-economic policies are a consistent policy set in their impact on farm incentives.

10. A strong case can be made for raising irrigation water rates and reducing subsidies on fertilizer, credit and other modern inputs. These inputs have been around for a long time and farmers are familiar with the their impact on output.

Raising input prices, with an associated increase in output prices, would have little effect on production, either in total or in its

composition. It would contribute to resource generation and would not in any major way worsen the income distribution. What is required is not low prices of inputs for small farmers but an equal and assured access to the inputs and services for small farmers but an equal and assured access. If access to the inputs and services is not assured for the small farmers, the benefits of any technological innovations would accrue mainly to the large farm sector.

- II. The nature of government financial subsidies to agriculture changed during the decade of the 1970s. The agricultural sector was taxed much more than it was subsidized in the early part of the 1970s. However, the net fiscal burden became negative as subsidies outpaced the tax revenue. Despite the stated government policy of eliminating open subsidies on fertilizer and pesticides, this situation has remained serious and will continue unless unbudgeted subsidies on water, credit and electricity are reduced drastically.
- 12. We have presented some evidence on the economic subsidies or taxation implicit in commodity transactions due to the divergence between domestic and international prices for a few selected crops. In the absence of similar calculations for all crops, the situation for the entire sector cannot be established. We have noted that distorting prices of commodities has definite implications for the efficient allocation of resources. The computation of economic subsidies assume that producers and consumers do not change their behaviour as a result of changed prices. In this sense, the computed magnitudes of economic subsidies or taxation may be biased.

13. The difficulties currently being faced with respect to resource generation and the undesirable impact of relying on commodity taxation as a source of government revenue has led us to appraise the potential of land taxes in Pakistan. The detailed analysis of this tax highlights

interesting implications for public policy. The evidence that the tax is progressive in its incidence and has desirable properties with respect to allocation of resources should be kept in view. A machinery to collect the tax from millions of farmers is in place and knews the rural scene very intimately. The main stumbling block, in the enhanced taxation from this source, has been the power of the landed class, who have always succeeded in blocking—any tax reform proposals that would hurt them. We have shown a way to lessen the effectiveness of the landed lobby. It is proposed that the tax revenue generated from land taxation be spent on projects benefitting the agricultural sector only. Once this practice takes hold, opposition to this tax should vanish. We also recommend that the present structure of the tax be kept as simple as possible. To begin with the land assessments and rates of land tax could be increased on an ad hoc basis to a level that generates significant revenue.

# BIBLIOGRAPHY

- Aaron, Henry. "Some Criticisms of Tax Burden Sharing". <u>National</u> Tax Journal, 18, No.3 (September 1965).
- Acharya, S.S. "Parity Issues in Pricing of Foodgrain Output".
   Indian Journal of Agricultural Economics. Vol.XXXVI, No.4.
   October-December, 1981.
- 3. Ahluwalia. Montek. "Inequality, Poverty and Development".

  Journal of Development Economics. Vol.3, pp.307-342.

  1976.
- 4. Alamgir, M. "The Domestic Price of Imported Commodities in Pakistan:
  A Further Study". Pakistan Development Review. Vol.VIII,
  No.1. Spring 1968.
- Alauddin, Talat. "Mass Poverty in Pakistan: A Further Study".
   The Pakistan Development Review. No.14, pp.431-50.
   (Winter 1975).
- 6. Alavi, Hamza. "The Rural Elite and Rural Development in Pakistan".
  Rural Development in Bangladesh and Pakistan, ed.
  Pohert Q. Stevens et. al. The University of Hawaii.
  Honolulu. 1976.
- Ayub, M.A. "Income Inequality in a Growth-Theoretic Context: The Case of Pakistan". Ph.D. thesis submitted to Yale University. May 1977.
- 8. Azam, A.K. "The Future of Green Revolution in West Pakistan: A Choice of Strategy". International Journal of Agrarian Affairs. March 1973.
- Azfar, Javed. "The Income Distribution in Pakistan Before and After Taxes: 1966-67". Unpublished dissertation, Harvard University, 1971.
- Azhar, B.A. "Land Revenue Assessment: A Case Study". <u>Pakistan</u> <u>Development Review</u>. Autumn 1982.
- Bergan, A. "Personal Income Distribution and Personal Savings in Pakistan". Pakistan Development Review. 1967.
- Bhatia, M.S. "Intra-Sectoral Parity Between Cost, Price and Income in Agriculture". Indian Journal of Agricultural Economics. Vol. XXXVI, No.4. October-December 1981.
- Bird, Richard. "A Note on 'Tax Sacrifice' Comparisons". National Tax Journal. Vol.XVII, No.3. (September 1964).

- Brown, Gilbert T. "Agricultural Pricing Policies in Developing Countries". In Theodore W. Shultz (ed.) Distortions of Agricultural Incentives. Bloomington: Indiana University Press. 1978.
- Chaudhary, M.A. and Malik Muhammad Ashraf. "An Economic Analysis
  of Level and Structure of Irrigation Water Charges".
  Pakistan Institute of Development Economics. March 1981.
- Chaudhry, M.G. "The Green Revolution and Income Inequality: Some Empirical Evidence from Rural Pakistan 1960-75". Ph.D. Thesis submitted to the University of Wisconsin. 1980.
- Cheema, Aftab Ahmad. "Poverty in Pakistan: Some New Dimensions". Unpublished Research Paper, PIDE, Islamabad 1985.
- 18. Cheong, Kee Cheok and Emmanuel H. D'Silva. "Prices, Terms of Trade and the Role of Government in Pakistan's Agriculture". World Bank Staff Working Paper No.643.
- 19. Cummings, Thomas. "Cultivator Market Responsiveness in Pakistan Cereal and Cash Crops". The Pakistan Development Review. Vol.XIV, No.3. Autumn 1975.
- Downs, Anthony. "Economic Theory of Democracy". London: Harper and Row Ltd. 1957.
- Ercelawn, Aly. "Income Inequality in Rural Pakistan: A Study of Sample Villages". Pakistan Journal of Applied Economics. Vol.III, No.1. 1984.
- Faiz, Mohammad. "Rural Income Distribution in Pakistan: A Synthesis". Islamabad. Pakistan Institute of Development Economics. (Unpublished).
- 23. Falcon, Walter P. "Farmer Response to Price in A Subsistence Economy: The Case of West Pakistan". The American Economic Review. Papers and Proceedings. Vol.54. 1964.
- 24. Falcon, W.P. "The Green Revolution: Second-Generations of Problems". American Journal of Agricultural Economics. Vol.52. December 1970.
- Fields, D.B. and W.T. Stanbury. "Incentives Disincentives and the Income Tax: Further Empirical Evidence". Public Finance, XXV, No.3. 1970.
- Fogan, Elmer D. "Recent and Contemporary Theories of Progressive Taxation". Journal of Political Economy. Vol.46, No.4. (August 1938).
- Frank, Henry J. "Measuring State Tax Burden". National Tax Journal. Vol.XII, No.2. (June 1959).
- Ghandhi, Ved P. "Company Tax Incidence-Industry Studies", Indian Economic Review. III, (New Series), No.1. 1968.

re of

Revenue from Indirect Taxes in Pakistan". The Paki Development Review. IV, No.3. 1964.

on, Michael. "Why Poor People Stay Poor". Harvard Un Press, Cambridge, Massachusetts, 1977, Ch.12. "Tax Towards the Rural Sector". pp.270-286.

a, Ashok. "Terms of Trade and Class Relations: An Es Ltd. Political Economy". London: Frank Cass and Compan 1977.

Patterns in West Pakistan". Pakistan Development Vol. V, No. 1. Spring 1965.

- hid, G.B.S. "A Note on Measurement of Poverty and Income Inequalities in Pakistan: Some Observations on Methodology" Pakistan Development Review. Vol.XVII, Autumn 1978.
- i, Syed Nawab Haider and Peter A. Cornelisse. "The Antomy of the Wheat Market in Pakistan. Islamabad: Pakistan institute of Development Economics. October 1984.
- m, S.M. "Mass Poverty in Pakistan: Some Preliminary Findings".

  Pakistan Development Review. Winter 1973.
- n, S.M. "Rural Poverty and Landlessness in Pakistan" in Poverty and Landlessness in Rural Asia; ILO, Geneva 1977.
- omber, Mabel. "An Index of the Taxpaying Ability of State and Local Governments". (New Yrok, Columbia University Press 1935).
- stan, Central Statistical Office, Survey on Population and Labour Force, 1967-68.
- stan, Government of. Agricultural Marketing Advisor. Markets and Prices Annual Bulletin 1971-72. Vol.26, No.12 A. Karachi, n.d.
- stan, Government of. Agricultural and Livestock Market Advisor. Markets and Prices Annual Bulletin 1979-sing Vol.32, No.12-A. Karachi 1980.
- Income and Expenditure Survey. 1979.
- Islamabad: Economic Adviser's Wing, Finance Divisi 1983.
- tan, Government of. Pakistan Economic Survey 1983-84
  Islamabad: Economic Adviser's Wing, Finance Division 1984

- 72. Pakistan, Government of Statistical Division. Monthly Statistic Bulletin. Various Issues (1971-72, 1979). Karachi: Central Statistical Office.
- 73. Pakistan Institute of Development Economics. Population, Labour Force, and Migration Survey. Islamabad. 1979.
- 74. Pakistan, Ministry of Agriculture. "Utilization of Agricultural Commodities".
- 75. Pal. Mati Lal. "The Determinants of the Domestic Prices of Impor Pakistan Development Review. Vol.IV, No.4. Winter 1964
- 76. Pearse, Andrew. "Technology and Production: Reflections on Globa Study". International Perspective in Rural Sociology. Ed. H. Newby (John Wiley and Sons Chichester, New York.
- Punjab Planning and Development Statistical Unit. Fertilizer and Mexican Wheat Survey Report. Lahore. 1970.
- 78. Qureshi, Sarfraz Khan. "Price Responsiveness of Marketed Surplus of Wheat in Pakistan". The Pakistan Development Review. Vol.XII, No.2. Summer 1974.
- 79. Qureshi, S.K. and A.N. Siddiqui. "Terms of Trade and the Agricul Price Policy". Islamabad, Pakistan Institute of Develop Economics, 1984. (Studies in Agriculture Pricing and Agricultural Taxation, Working Paper No.3).
- 80. Radhu, Ghulam Mohammad. "The Relation of Indirect Tax Changes to Price Changes". The Pakistan Development Review. No.1,
- Sadiq, N. "Income Distribution in Pakistan". Unpublished paper (Mimeographed) Statistics Division. 1972.
- 82. Scandizzo, Pasquale L. and Colin Bruce. "Methodologies for Measur Agricultural Price Intervention Effects". World Bank Sta Working Paper No.387. April 1980.
- 83. Sen, Amartya. "Poverty. "Poverty: An Ordinal Approach to Measurer Econometrica. 44 (March 1976), pp.219-231.
- 84. Shu, David Lin. "Supply Responses of Primary Producers". Penerb University, Malaya, Kuala Lumpur 1975.

- 85. Suleman, P.M.U. "Employment, Income Distribution and Social Justice: Analysis of Distribution of Personal Income in West Pakistan". Paper presented to the 16th Annual All Pakistan Economic Conference, Islamabad University Islamabad. February 18-20, 1973.
- 86. Tyagi, D.S. "Review of Agriculture: Farm Prices and Class Bias in India". Economic and Political Weekly.
  September 1979.
- 87. Wald, Haskell. "Taxation of Agricultural Land in Underdeveloped Economics". (Cambridge, Harvard University Press, 1952)
- 88. Wasay, Abdul. "An Urban Poverty Line Estimate". The Pakistan Development Review. 16 (Spring 1977). pp.49-51.
- 89. West Pakistan, Government of. "Data on Taxes, Duties and Other Receipts". 1970.
- 90. West Pakistan, Government of. Planning Board. "Agricultural Income in West Pakistan, 1967-68.
- 91. Wizarat, Shahida. "Technical Change in Pakistan's Agriculture: 1953-54 to 1977-78". Pakistan Institute of Development Economics, Islamabad. Research Report No. 120.

This work is licensed under a Creative Commons
Attribution – NonCommercial - NoDerivs 3.0 Licence.

To view a copy of the licence please see: http://creativecommons.org/licenses/by-nc-nd/3.0/